PATENT ASSIGNMENT COVER SHEET

Electronic Version v1.1 Stylesheet Version v1.2 EPAS ID: PAT7179454

SUBMISSION TYPE:	NEW ASSIGNMENT
NATURE OF CONVEYANCE:	ASSIGNMENT

CONVEYING PARTY DATA

Name	Execution Date
POLLEN GEAR HOLDINGS LLC	01/04/2019
POLLEN GEAR LLC	01/04/2019

RECEIVING PARTY DATA

Name:	GREENLANE HOLDINGS LLC
Street Address:	1095 BROKEN SOUND PKWY, SUITE 300
City:	BOCA RATON
State/Country:	FLORIDA
Postal Code:	33487

PROPERTY NUMBERS Total: 26

Property Type	Number
Application Number:	62715955
Application Number:	62711172
Application Number:	62586773
Application Number:	62490345
Application Number:	62439357
Application Number:	62422256
Application Number:	62419209
Application Number:	62405436
Application Number:	29670346
Application Number:	29670338
Application Number:	29658099
Application Number:	29658083
Application Number:	29658079
Application Number:	29626205
Application Number:	29626177
Application Number:	29601826
Application Number:	29601823
Application Number:	29589025
Application Number:	29589022

PATENT REEL: 059126 FRAME: 0610

507132613

Property Type	Number
Application Number:	29589016
Application Number:	29584476
Application Number:	29583722
Application Number:	29580322
Application Number:	29517993
PCT Number:	US2018006138
PCT Number:	US2017055458

CORRESPONDENCE DATA

Fax Number:

Correspondence will be sent to the e-mail address first; if that is unsuccessful, it will be sent using a fax number, if provided; if that is unsuccessful, it will be sent via US Mail.

Phone: 5612886578

Email: drichmond@greenlane.com

Correspondent Name: DEREK RICHMOND

Address Line 1: 1095 BROKEN SOUND PKWY

Address Line 2: SUITE 300

Address Line 4: BOCA RATON, FLORIDA 33487

NAME OF SUBMITTER:	DEREK RICHMOND/
SIGNATURE:	/Derek Richmond/
DATE SIGNED:	02/16/2022

Total Attachments: 136

source=Tab 01 I	Pollen Gear -	Contribution	Agreement#page1.tif
source=Tab 01 I	Pollen Gear -	Contribution	Agreement#page2.tif
source=Tab 01 I	Pollen Gear -	Contribution	Agreement#page3.tif
source=Tab 01 I	Pollen Gear -	Contribution	Agreement#page4.tif
source=Tab 01 I	Pollen Gear -	Contribution	Agreement#page5.tif
source=Tab 01 I	Pollen Gear -	Contribution	Agreement#page6.tif
source=Tab 01 I	Pollen Gear -	Contribution	Agreement#page7.tif
source=Tab 01 I	Pollen Gear -	Contribution	Agreement#page8.tif
source=Tab 01 I	Pollen Gear -	Contribution	Agreement#page9.tif
source=Tab 01 I	Pollen Gear -	Contribution	Agreement#page10.tif
source=Tab 01 I	Pollen Gear -	Contribution	Agreement#page11.tif
source=Tab 01 I	Pollen Gear -	Contribution	Agreement#page12.tif
source=Tab 01 I	Pollen Gear -	Contribution	Agreement#page13.tif
source=Tab 01 I	Pollen Gear -	Contribution	Agreement#page14.tif
source=Tab 01 I	Pollen Gear -	Contribution	Agreement#page15.tif
source=Tab 01 I	Pollen Gear -	Contribution	Agreement#page16.tif
source=Tab 01 I	Pollen Gear -	Contribution	Agreement#page17.tif
source=Tab 01 I	Pollen Gear -	Contribution	Agreement#page18.tif
source=Tab 01 I	Pollen Gear -	Contribution	Agreement#page19.tif
source=Tab 01 I	Pollen Gear -	Contribution	Agreement#page20.tif
source=Tab 01 I	Pollen Gear -	Contribution	Agreement#page21.tif

source=Tab 01 Pollen Gear - Contribution Agreement#page24.tif source=Tab 01 Pollen Gear - Contribution Agreement#page25.tif source=Tab 01 Pollen Gear - Contribution Agreement#page26.tif source=Tab 01 Pollen Gear - Contribution Agreement#page26.tif source=Tab 01 Pollen Gear - Contribution Agreement#page27.tif source=Tab 01 Pollen Gear - Contribution Agreement#page29.tif source=Tab 01 Pollen Gear - Contribution Agreement#page29.tif source=Tab 01 Pollen Gear - Contribution Agreement#page30.tif source=Tab 01 Pollen Gear - Contribution Agreement#page31.tif source=Tab 01 Pollen Gear - Contribution Agreement#page31.tif source=Tab 01 Pollen Gear - Contribution Agreement#page31.tif source=Tab 01 Pollen Gear - Contribution Agreement#page33.tif source=Tab 01 Pollen Gear - Contribution Agreement#page33.tif source=Tab 01 Pollen Gear - Contribution Agreement#page36.tif source=Tab 01 Pollen Gear - Contribution Agreement#page36.tif source=Tab 01 Pollen Gear - Contribution Agreement#page36.tif source=Tab 01 Pollen Gear - Contribution Agreement#page37.tif source=Tab 01 Pollen Gear - Contribution Agreement#page39.tif source=Tab 01 Pollen Gear - Contribution Agreement#page40.tif source=Tab 01 Pollen Gear - Contribution Agreement#page41.tif source=Tab 01 Pollen Gear - Contribution Agreement#page41.tif source=Tab 01 Pollen Gear - Contribution Agreement#page44.tif source=Tab 01 Pollen Gear - Contribution Agreement#page44.tif source=Tab 01 Pollen Gear - Contribution Agreement#page48.tif source=Tab 01 Pollen Gear - Contribution Agreement#page48.tif source=Tab 01 Pollen Gear - Contribution Agreement#page45.tif source=Tab 01 Pollen Gear - Contribution Agreement#page45.tif source=Tab 01 Pollen Gear - Contribution Agreement#page50.tif	source=Tab 01 Pollen Gear - Contribution Agreement#page22.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page24.tif source=Tab 01 Pollen Gear - Contribution Agreement#page25.tif source=Tab 01 Pollen Gear - Contribution Agreement#page25.tif source=Tab 01 Pollen Gear - Contribution Agreement#page25.tif source=Tab 01 Pollen Gear - Contribution Agreement#page29.tif source=Tab 01 Pollen Gear - Contribution Agreement#page29.tif source=Tab 01 Pollen Gear - Contribution Agreement#page30.tif source=Tab 01 Pollen Gear - Contribution Agreement#page31.tif source=Tab 01 Pollen Gear - Contribution Agreement#page32.tif source=Tab 01 Pollen Gear - Contribution Agreement#page32.tif source=Tab 01 Pollen Gear - Contribution Agreement#page33.tif source=Tab 01 Pollen Gear - Contribution Agreement#page33.tif source=Tab 01 Pollen Gear - Contribution Agreement#page35.tif source=Tab 01 Pollen Gear - Contribution Agreement#page35.tif source=Tab 01 Pollen Gear - Contribution Agreement#page37.tif source=Tab 01 Pollen Gear - Contribution Agreement#page33.tif source=Tab 01 Pollen Gear - Contribution Agreement#page39.tif source=Tab 01 Pollen Gear - Contribution Agreement#page39.tif source=Tab 01 Pollen Gear - Contribution Agreement#page40.tif source=Tab 01 Pollen Gear - Contribution Agreement#page41.tif source=Tab 01 Pollen Gear - Contribution Agreement#page43.tif source=Tab 01 Pollen Gear - Contribution Agreement#page44.tif source=Tab 01 Pollen Gear - Contribution Agreement#page44.tif source=Tab 01 Pollen Gear - Contribution Agreement#page44.tif source=Tab 01 Pollen Gear - Contribution Agreement#page45.tif source=Tab 01 Pollen Gear - Contribution Agreement#page47.tif source=Tab 01 Pollen Gear - Contribution Agreement#page45.tif source=Tab 01 Pollen Gear - Contribution Agreement#page45.tif source=Tab 01 Pollen Gear - Contribution Agreement#page50.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif	1
source=Tab 01 Pollen Gear - Contribution Agreement#page25.tif source=Tab 01 Pollen Gear - Contribution Agreement#page26.tif source=Tab 01 Pollen Gear - Contribution Agreement#page27.tif source=Tab 01 Pollen Gear - Contribution Agreement#page28.tif source=Tab 01 Pollen Gear - Contribution Agreement#page29.tif source=Tab 01 Pollen Gear - Contribution Agreement#page30.tif source=Tab 01 Pollen Gear - Contribution Agreement#page31.tif source=Tab 01 Pollen Gear - Contribution Agreement#page31.tif source=Tab 01 Pollen Gear - Contribution Agreement#page33.tif source=Tab 01 Pollen Gear - Contribution Agreement#page37.tif source=Tab 01 Pollen Gear - Contribution Agreement#page33.tif source=Tab 01 Pollen Gear - Contribution Agreement#page39.tif source=Tab 01 Pollen Gear - Contribution Agreement#page40.tif source=Tab 01 Pollen Gear - Contribution Agreement#page40.tif source=Tab 01 Pollen Gear - Contribution Agreement#page44.tif source=Tab 01 Pollen Gear - Contribution Agreement#page44.tif source=Tab 01 Pollen Gear - Contribution Agreement#page44.tif source=Tab 01 Pollen Gear - Contribution Agreement#page46.tif source=Tab 01 Pollen Gear - Contribution Agreement#page46.tif source=Tab 01 Pollen Gear - Contribution Agreement#page45.tif source=Tab 01 Pollen Gear - Contribution Agreement#page45.tif source=Tab 01 Pollen Gear - Contribution Agreement#page50.tif source=Tab 01 Pollen Gear - Contribution Agreement#page55.tif source=Tab 01 Pollen Gear - Contribution Agreement#page55.tif source=Tab 01 Pollen Gear - Contribution Agreement#page55.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif	source=Tab 01 Pollen Gear - Contribution Agreement#page24.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page28.tif source=Tab 01 Pollen Gear - Contribution Agreement#page28.tif source=Tab 01 Pollen Gear - Contribution Agreement#page29.tif source=Tab 01 Pollen Gear - Contribution Agreement#page30.tif source=Tab 01 Pollen Gear - Contribution Agreement#page30.tif source=Tab 01 Pollen Gear - Contribution Agreement#page33.tif source=Tab 01 Pollen Gear - Contribution Agreement#page33.tif source=Tab 01 Pollen Gear - Contribution Agreement#page33.tif source=Tab 01 Pollen Gear - Contribution Agreement#page35.tif source=Tab 01 Pollen Gear - Contribution Agreement#page35.tif source=Tab 01 Pollen Gear - Contribution Agreement#page37.tif source=Tab 01 Pollen Gear - Contribution Agreement#page37.tif source=Tab 01 Pollen Gear - Contribution Agreement#page37.tif source=Tab 01 Pollen Gear - Contribution Agreement#page33.tif source=Tab 01 Pollen Gear - Contribution Agreement#page33.tif source=Tab 01 Pollen Gear - Contribution Agreement#page40.tif source=Tab 01 Pollen Gear - Contribution Agreement#page44.tif source=Tab 01 Pollen Gear - Contribution Agreement#page44.tif source=Tab 01 Pollen Gear - Contribution Agreement#page44.tif source=Tab 01 Pollen Gear - Contribution Agreement#page45.tif source=Tab 01 Pollen Gear - Contribution Agreement#page50.tif source=Tab 01 Pollen Gear - Contribution Agreement#page55.tif source=Tab 01 Pollen Gear - Contribution Agreement#page66.tif source=Tab 01 Pollen Gear - Contribution Agreement#page66.tif	source=Tab 01 Pollen Gear - Contribution Agreement#page25.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page28.tif source=Tab 01 Pollen Gear - Contribution Agreement#page28.tif source=Tab 01 Pollen Gear - Contribution Agreement#page29.tif source=Tab 01 Pollen Gear - Contribution Agreement#page30.tif source=Tab 01 Pollen Gear - Contribution Agreement#page30.tif source=Tab 01 Pollen Gear - Contribution Agreement#page33.tif source=Tab 01 Pollen Gear - Contribution Agreement#page33.tif source=Tab 01 Pollen Gear - Contribution Agreement#page33.tif source=Tab 01 Pollen Gear - Contribution Agreement#page35.tif source=Tab 01 Pollen Gear - Contribution Agreement#page35.tif source=Tab 01 Pollen Gear - Contribution Agreement#page37.tif source=Tab 01 Pollen Gear - Contribution Agreement#page37.tif source=Tab 01 Pollen Gear - Contribution Agreement#page37.tif source=Tab 01 Pollen Gear - Contribution Agreement#page33.tif source=Tab 01 Pollen Gear - Contribution Agreement#page33.tif source=Tab 01 Pollen Gear - Contribution Agreement#page40.tif source=Tab 01 Pollen Gear - Contribution Agreement#page44.tif source=Tab 01 Pollen Gear - Contribution Agreement#page44.tif source=Tab 01 Pollen Gear - Contribution Agreement#page44.tif source=Tab 01 Pollen Gear - Contribution Agreement#page45.tif source=Tab 01 Pollen Gear - Contribution Agreement#page50.tif source=Tab 01 Pollen Gear - Contribution Agreement#page55.tif source=Tab 01 Pollen Gear - Contribution Agreement#page66.tif source=Tab 01 Pollen Gear - Contribution Agreement#page66.tif	source=Tab 01 Pollen Gear - Contribution Agreement#page26.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page28.tif source=Tab 01 Pollen Gear - Contribution Agreement#page30.tif source=Tab 01 Pollen Gear - Contribution Agreement#page30.tif source=Tab 01 Pollen Gear - Contribution Agreement#page31.tif source=Tab 01 Pollen Gear - Contribution Agreement#page33.tif source=Tab 01 Pollen Gear - Contribution Agreement#page35.tif source=Tab 01 Pollen Gear - Contribution Agreement#page35.tif source=Tab 01 Pollen Gear - Contribution Agreement#page37.tif source=Tab 01 Pollen Gear - Contribution Agreement#page38.tif source=Tab 01 Pollen Gear - Contribution Agreement#page39.tif source=Tab 01 Pollen Gear - Contribution Agreement#page40.tif source=Tab 01 Pollen Gear - Contribution Agreement#page41.tif source=Tab 01 Pollen Gear - Contribution Agreement#page42.tif source=Tab 01 Pollen Gear - Contribution Agreement#page43.tif source=Tab 01 Pollen Gear - Contribution Agreement#page44.tif source=Tab 01 Pollen Gear - Contribution Agreement#page45.tif source=Tab 01 Pollen Gear - Contribution Agreement#page45.tif source=Tab 01 Pollen Gear - Contribution Agreement#page48.tif source=Tab 01 Pollen Gear - Contribution Agreement#page48.tif source=Tab 01 Pollen Gear - Contribution Agreement#page50.tif source=Tab 01 Pollen Gear - Contribution Agreement#page51.tif source=Tab 01 Pollen Gear - Contribution Agreement#page53.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif	
source=Tab 01 Pollen Gear - Contribution Agreement#page29.tif source=Tab 01 Pollen Gear - Contribution Agreement#page30.tif source=Tab 01 Pollen Gear - Contribution Agreement#page31.tif source=Tab 01 Pollen Gear - Contribution Agreement#page32.tif source=Tab 01 Pollen Gear - Contribution Agreement#page32.tif source=Tab 01 Pollen Gear - Contribution Agreement#page34.tif source=Tab 01 Pollen Gear - Contribution Agreement#page35.tif source=Tab 01 Pollen Gear - Contribution Agreement#page36.tif source=Tab 01 Pollen Gear - Contribution Agreement#page36.tif source=Tab 01 Pollen Gear - Contribution Agreement#page37.tif source=Tab 01 Pollen Gear - Contribution Agreement#page39.tif source=Tab 01 Pollen Gear - Contribution Agreement#page39.tif source=Tab 01 Pollen Gear - Contribution Agreement#page40.tif source=Tab 01 Pollen Gear - Contribution Agreement#page41.tif source=Tab 01 Pollen Gear - Contribution Agreement#page42.tif source=Tab 01 Pollen Gear - Contribution Agreement#page43.tif source=Tab 01 Pollen Gear - Contribution Agreement#page44.tif source=Tab 01 Pollen Gear - Contribution Agreement#page45.tif source=Tab 01 Pollen Gear - Contribution Agreement#page47.tif source=Tab 01 Pollen Gear - Contribution Agreement#page48.tif source=Tab 01 Pollen Gear - Contribution Agreement#page49.tif source=Tab 01 Pollen Gear - Contribution Agreement#page49.tif source=Tab 01 Pollen Gear - Contribution Agreement#page50.tif source=Tab 01 Pollen Gear - Contribution Agreement#page50.tif source=Tab 01 Pollen Gear - Contribution Agreement#page53.tif source=Tab 01 Pollen Gear - Contribution Agreement#page54.tif source=Tab 01 Pollen Gear - Contribution Agreement#page55.tif source=Tab 01 Pollen Gear - Contribution Agreement#page64.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif	
source=Tab 01 Pollen Gear - Contribution Agreement#page30.tif source=Tab 01 Pollen Gear - Contribution Agreement#page31.tif source=Tab 01 Pollen Gear - Contribution Agreement#page32.tif source=Tab 01 Pollen Gear - Contribution Agreement#page33.tif source=Tab 01 Pollen Gear - Contribution Agreement#page34.tif source=Tab 01 Pollen Gear - Contribution Agreement#page35.tif source=Tab 01 Pollen Gear - Contribution Agreement#page36.tif source=Tab 01 Pollen Gear - Contribution Agreement#page37.tif source=Tab 01 Pollen Gear - Contribution Agreement#page38.tif source=Tab 01 Pollen Gear - Contribution Agreement#page39.tif source=Tab 01 Pollen Gear - Contribution Agreement#page39.tif source=Tab 01 Pollen Gear - Contribution Agreement#page40.tif source=Tab 01 Pollen Gear - Contribution Agreement#page41.tif source=Tab 01 Pollen Gear - Contribution Agreement#page42.tif source=Tab 01 Pollen Gear - Contribution Agreement#page44.tif source=Tab 01 Pollen Gear - Contribution Agreement#page44.tif source=Tab 01 Pollen Gear - Contribution Agreement#page45.tif source=Tab 01 Pollen Gear - Contribution Agreement#page45.tif source=Tab 01 Pollen Gear - Contribution Agreement#page48.tif source=Tab 01 Pollen Gear - Contribution Agreement#page49.tif source=Tab 01 Pollen Gear - Contribution Agreement#page50.tif source=Tab 01 Pollen Gear - Contribution Agreement#page50.tif source=Tab 01 Pollen Gear - Contribution Agreement#page51.tif source=Tab 01 Pollen Gear - Contribution Agreement#page53.tif source=Tab 01 Pollen Gear - Contribution Agreement#page54.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif	
source=Tab 01 Pollen Gear - Contribution Agreement#page31.tif source=Tab 01 Pollen Gear - Contribution Agreement#page32.tif source=Tab 01 Pollen Gear - Contribution Agreement#page33.tif source=Tab 01 Pollen Gear - Contribution Agreement#page34.tif source=Tab 01 Pollen Gear - Contribution Agreement#page35.tif source=Tab 01 Pollen Gear - Contribution Agreement#page36.tif source=Tab 01 Pollen Gear - Contribution Agreement#page37.tif source=Tab 01 Pollen Gear - Contribution Agreement#page38.tif source=Tab 01 Pollen Gear - Contribution Agreement#page39.tif source=Tab 01 Pollen Gear - Contribution Agreement#page39.tif source=Tab 01 Pollen Gear - Contribution Agreement#page40.tif source=Tab 01 Pollen Gear - Contribution Agreement#page41.tif source=Tab 01 Pollen Gear - Contribution Agreement#page42.tif source=Tab 01 Pollen Gear - Contribution Agreement#page43.tif source=Tab 01 Pollen Gear - Contribution Agreement#page44.tif source=Tab 01 Pollen Gear - Contribution Agreement#page45.tif source=Tab 01 Pollen Gear - Contribution Agreement#page48.tif source=Tab 01 Pollen Gear - Contribution Agreement#page48.tif source=Tab 01 Pollen Gear - Contribution Agreement#page49.tif source=Tab 01 Pollen Gear - Contribution Agreement#page49.tif source=Tab 01 Pollen Gear - Contribution Agreement#page50.tif source=Tab 01 Pollen Gear - Contribution Agreement#page50.tif source=Tab 01 Pollen Gear - Contribution Agreement#page51.tif source=Tab 01 Pollen Gear - Contribution Agreement#page53.tif source=Tab 01 Pollen Gear - Contribution Agreement#page54.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page64.tif	1
source=Tab 01 Pollen Gear - Contribution Agreement#page32.tif source=Tab 01 Pollen Gear - Contribution Agreement#page33.tif source=Tab 01 Pollen Gear - Contribution Agreement#page34.tif source=Tab 01 Pollen Gear - Contribution Agreement#page35.tif source=Tab 01 Pollen Gear - Contribution Agreement#page36.tif source=Tab 01 Pollen Gear - Contribution Agreement#page37.tif source=Tab 01 Pollen Gear - Contribution Agreement#page37.tif source=Tab 01 Pollen Gear - Contribution Agreement#page39.tif source=Tab 01 Pollen Gear - Contribution Agreement#page39.tif source=Tab 01 Pollen Gear - Contribution Agreement#page39.tif source=Tab 01 Pollen Gear - Contribution Agreement#page40.tif source=Tab 01 Pollen Gear - Contribution Agreement#page41.tif source=Tab 01 Pollen Gear - Contribution Agreement#page42.tif source=Tab 01 Pollen Gear - Contribution Agreement#page43.tif source=Tab 01 Pollen Gear - Contribution Agreement#page44.tif source=Tab 01 Pollen Gear - Contribution Agreement#page45.tif source=Tab 01 Pollen Gear - Contribution Agreement#page47.tif source=Tab 01 Pollen Gear - Contribution Agreement#page48.tif source=Tab 01 Pollen Gear - Contribution Agreement#page49.tif source=Tab 01 Pollen Gear - Contribution Agreement#page49.tif source=Tab 01 Pollen Gear - Contribution Agreement#page50.tif source=Tab 01 Pollen Gear - Contribution Agreement#page50.tif source=Tab 01 Pollen Gear - Contribution Agreement#page52.tif source=Tab 01 Pollen Gear - Contribution Agreement#page53.tif source=Tab 01 Pollen Gear - Contribution Agreement#page54.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif source=Tab 01 Pollen Gear - Contribution Agreement#page64.tif source=Tab 01 Pollen Gear - Contribution Agreement#page64.tif source=Tab 01 Pollen Gear - Contribution Agreement#page64.tif	1
source=Tab 01 Pollen Gear - Contribution Agreement#page33.tif source=Tab 01 Pollen Gear - Contribution Agreement#page34.tif source=Tab 01 Pollen Gear - Contribution Agreement#page35.tif source=Tab 01 Pollen Gear - Contribution Agreement#page36.tif source=Tab 01 Pollen Gear - Contribution Agreement#page37.tif source=Tab 01 Pollen Gear - Contribution Agreement#page38.tif source=Tab 01 Pollen Gear - Contribution Agreement#page39.tif source=Tab 01 Pollen Gear - Contribution Agreement#page39.tif source=Tab 01 Pollen Gear - Contribution Agreement#page40.tif source=Tab 01 Pollen Gear - Contribution Agreement#page41.tif source=Tab 01 Pollen Gear - Contribution Agreement#page41.tif source=Tab 01 Pollen Gear - Contribution Agreement#page43.tif source=Tab 01 Pollen Gear - Contribution Agreement#page44.tif source=Tab 01 Pollen Gear - Contribution Agreement#page45.tif source=Tab 01 Pollen Gear - Contribution Agreement#page46.tif source=Tab 01 Pollen Gear - Contribution Agreement#page47.tif source=Tab 01 Pollen Gear - Contribution Agreement#page48.tif source=Tab 01 Pollen Gear - Contribution Agreement#page49.tif source=Tab 01 Pollen Gear - Contribution Agreement#page49.tif source=Tab 01 Pollen Gear - Contribution Agreement#page50.tif source=Tab 01 Pollen Gear - Contribution Agreement#page50.tif source=Tab 01 Pollen Gear - Contribution Agreement#page51.tif source=Tab 01 Pollen Gear - Contribution Agreement#page53.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page64.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif	
source=Tab 01 Pollen Gear - Contribution Agreement#page35.tif source=Tab 01 Pollen Gear - Contribution Agreement#page37.tif source=Tab 01 Pollen Gear - Contribution Agreement#page37.tif source=Tab 01 Pollen Gear - Contribution Agreement#page38.tif source=Tab 01 Pollen Gear - Contribution Agreement#page39.tif source=Tab 01 Pollen Gear - Contribution Agreement#page40.tif source=Tab 01 Pollen Gear - Contribution Agreement#page41.tif source=Tab 01 Pollen Gear - Contribution Agreement#page42.tif source=Tab 01 Pollen Gear - Contribution Agreement#page43.tif source=Tab 01 Pollen Gear - Contribution Agreement#page43.tif source=Tab 01 Pollen Gear - Contribution Agreement#page44.tif source=Tab 01 Pollen Gear - Contribution Agreement#page45.tif source=Tab 01 Pollen Gear - Contribution Agreement#page46.tif source=Tab 01 Pollen Gear - Contribution Agreement#page44.tif source=Tab 01 Pollen Gear - Contribution Agreement#page49.tif source=Tab 01 Pollen Gear - Contribution Agreement#page49.tif source=Tab 01 Pollen Gear - Contribution Agreement#page49.tif source=Tab 01 Pollen Gear - Contribution Agreement#page50.tif source=Tab 01 Pollen Gear - Contribution Agreement#page50.tif source=Tab 01 Pollen Gear - Contribution Agreement#page51.tif source=Tab 01 Pollen Gear - Contribution Agreement#page52.tif source=Tab 01 Pollen Gear - Contribution Agreement#page53.tif source=Tab 01 Pollen Gear - Contribution Agreement#page53.tif source=Tab 01 Pollen Gear - Contribution Agreement#page55.tif source=Tab 01 Pollen Gear - Contribution Agreement#page50.tif source=Tab 01 Pollen Gear - Contribution Agreement#page53.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif	
source=Tab 01 Pollen Gear - Contribution Agreement#page35.tif source=Tab 01 Pollen Gear - Contribution Agreement#page37.tif source=Tab 01 Pollen Gear - Contribution Agreement#page37.tif source=Tab 01 Pollen Gear - Contribution Agreement#page38.tif source=Tab 01 Pollen Gear - Contribution Agreement#page39.tif source=Tab 01 Pollen Gear - Contribution Agreement#page40.tif source=Tab 01 Pollen Gear - Contribution Agreement#page41.tif source=Tab 01 Pollen Gear - Contribution Agreement#page42.tif source=Tab 01 Pollen Gear - Contribution Agreement#page43.tif source=Tab 01 Pollen Gear - Contribution Agreement#page43.tif source=Tab 01 Pollen Gear - Contribution Agreement#page44.tif source=Tab 01 Pollen Gear - Contribution Agreement#page45.tif source=Tab 01 Pollen Gear - Contribution Agreement#page46.tif source=Tab 01 Pollen Gear - Contribution Agreement#page44.tif source=Tab 01 Pollen Gear - Contribution Agreement#page49.tif source=Tab 01 Pollen Gear - Contribution Agreement#page49.tif source=Tab 01 Pollen Gear - Contribution Agreement#page49.tif source=Tab 01 Pollen Gear - Contribution Agreement#page50.tif source=Tab 01 Pollen Gear - Contribution Agreement#page50.tif source=Tab 01 Pollen Gear - Contribution Agreement#page51.tif source=Tab 01 Pollen Gear - Contribution Agreement#page52.tif source=Tab 01 Pollen Gear - Contribution Agreement#page53.tif source=Tab 01 Pollen Gear - Contribution Agreement#page53.tif source=Tab 01 Pollen Gear - Contribution Agreement#page55.tif source=Tab 01 Pollen Gear - Contribution Agreement#page50.tif source=Tab 01 Pollen Gear - Contribution Agreement#page53.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif	source=Tab 01 Pollen Gear - Contribution Agreement#page34.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page37.tif source=Tab 01 Pollen Gear - Contribution Agreement#page38.tif source=Tab 01 Pollen Gear - Contribution Agreement#page39.tif source=Tab 01 Pollen Gear - Contribution Agreement#page40.tif source=Tab 01 Pollen Gear - Contribution Agreement#page41.tif source=Tab 01 Pollen Gear - Contribution Agreement#page42.tif source=Tab 01 Pollen Gear - Contribution Agreement#page43.tif source=Tab 01 Pollen Gear - Contribution Agreement#page43.tif source=Tab 01 Pollen Gear - Contribution Agreement#page45.tif source=Tab 01 Pollen Gear - Contribution Agreement#page46.tif source=Tab 01 Pollen Gear - Contribution Agreement#page47.tif source=Tab 01 Pollen Gear - Contribution Agreement#page49.tif source=Tab 01 Pollen Gear - Contribution Agreement#page49.tif source=Tab 01 Pollen Gear - Contribution Agreement#page50.tif source=Tab 01 Pollen Gear - Contribution Agreement#page50.tif source=Tab 01 Pollen Gear - Contribution Agreement#page52.tif source=Tab 01 Pollen Gear - Contribution Agreement#page53.tif source=Tab 01 Pollen Gear - Contribution Agreement#page53.tif source=Tab 01 Pollen Gear - Contribution Agreement#page54.tif source=Tab 01 Pollen Gear - Contribution Agreement#page55.tif source=Tab 01 Pollen Gear - Contribution Agreement#page55.tif source=Tab 01 Pollen Gear - Contribution Agreement#page56.tif source=Tab 01 Pollen Gear - Contribution Agreement#page59.tif source=Tab 01 Pollen Gear - Contribution Agreement#page58.tif source=Tab 01 Pollen Gear - Contribution Agreement#page61.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page66.tif source=Tab 01 Pollen Gear - Contribution Agreement#page61.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif	source=Tab 01 Pollen Gear - Contribution Agreement#page35.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page38.tif source=Tab 01 Pollen Gear - Contribution Agreement#page39.tif source=Tab 01 Pollen Gear - Contribution Agreement#page40.tif source=Tab 01 Pollen Gear - Contribution Agreement#page41.tif source=Tab 01 Pollen Gear - Contribution Agreement#page42.tif source=Tab 01 Pollen Gear - Contribution Agreement#page43.tif source=Tab 01 Pollen Gear - Contribution Agreement#page44.tif source=Tab 01 Pollen Gear - Contribution Agreement#page45.tif source=Tab 01 Pollen Gear - Contribution Agreement#page45.tif source=Tab 01 Pollen Gear - Contribution Agreement#page47.tif source=Tab 01 Pollen Gear - Contribution Agreement#page47.tif source=Tab 01 Pollen Gear - Contribution Agreement#page49.tif source=Tab 01 Pollen Gear - Contribution Agreement#page50.tif source=Tab 01 Pollen Gear - Contribution Agreement#page50.tif source=Tab 01 Pollen Gear - Contribution Agreement#page51.tif source=Tab 01 Pollen Gear - Contribution Agreement#page52.tif source=Tab 01 Pollen Gear - Contribution Agreement#page53.tif source=Tab 01 Pollen Gear - Contribution Agreement#page54.tif source=Tab 01 Pollen Gear - Contribution Agreement#page55.tif source=Tab 01 Pollen Gear - Contribution Agreement#page56.tif source=Tab 01 Pollen Gear - Contribution Agreement#page57.tif source=Tab 01 Pollen Gear - Contribution Agreement#page59.tif source=Tab 01 Pollen Gear - Contribution Agreement#page61.tif source=Tab 01 Pollen Gear - Contribution Agreement#page61.tif source=Tab 01 Pollen Gear - Contribution Agreement#page61.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page64.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif source=Tab 01 Pollen Gear - Contribution Agreement#page66.tif source=Tab 01 Pollen Gear - Contribution Agreement#page67.tif	source=Tab 01 Pollen Gear - Contribution Agreement#page36.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page38.tif source=Tab 01 Pollen Gear - Contribution Agreement#page39.tif source=Tab 01 Pollen Gear - Contribution Agreement#page40.tif source=Tab 01 Pollen Gear - Contribution Agreement#page41.tif source=Tab 01 Pollen Gear - Contribution Agreement#page42.tif source=Tab 01 Pollen Gear - Contribution Agreement#page43.tif source=Tab 01 Pollen Gear - Contribution Agreement#page44.tif source=Tab 01 Pollen Gear - Contribution Agreement#page45.tif source=Tab 01 Pollen Gear - Contribution Agreement#page45.tif source=Tab 01 Pollen Gear - Contribution Agreement#page47.tif source=Tab 01 Pollen Gear - Contribution Agreement#page47.tif source=Tab 01 Pollen Gear - Contribution Agreement#page49.tif source=Tab 01 Pollen Gear - Contribution Agreement#page50.tif source=Tab 01 Pollen Gear - Contribution Agreement#page50.tif source=Tab 01 Pollen Gear - Contribution Agreement#page51.tif source=Tab 01 Pollen Gear - Contribution Agreement#page52.tif source=Tab 01 Pollen Gear - Contribution Agreement#page53.tif source=Tab 01 Pollen Gear - Contribution Agreement#page54.tif source=Tab 01 Pollen Gear - Contribution Agreement#page55.tif source=Tab 01 Pollen Gear - Contribution Agreement#page56.tif source=Tab 01 Pollen Gear - Contribution Agreement#page57.tif source=Tab 01 Pollen Gear - Contribution Agreement#page59.tif source=Tab 01 Pollen Gear - Contribution Agreement#page61.tif source=Tab 01 Pollen Gear - Contribution Agreement#page61.tif source=Tab 01 Pollen Gear - Contribution Agreement#page61.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page64.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif source=Tab 01 Pollen Gear - Contribution Agreement#page66.tif source=Tab 01 Pollen Gear - Contribution Agreement#page67.tif	source=Tab 01 Pollen Gear - Contribution Agreement#page37.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page39.tif source=Tab 01 Pollen Gear - Contribution Agreement#page40.tif source=Tab 01 Pollen Gear - Contribution Agreement#page41.tif source=Tab 01 Pollen Gear - Contribution Agreement#page42.tif source=Tab 01 Pollen Gear - Contribution Agreement#page43.tif source=Tab 01 Pollen Gear - Contribution Agreement#page44.tif source=Tab 01 Pollen Gear - Contribution Agreement#page45.tif source=Tab 01 Pollen Gear - Contribution Agreement#page45.tif source=Tab 01 Pollen Gear - Contribution Agreement#page47.tif source=Tab 01 Pollen Gear - Contribution Agreement#page48.tif source=Tab 01 Pollen Gear - Contribution Agreement#page49.tif source=Tab 01 Pollen Gear - Contribution Agreement#page50.tif source=Tab 01 Pollen Gear - Contribution Agreement#page51.tif source=Tab 01 Pollen Gear - Contribution Agreement#page52.tif source=Tab 01 Pollen Gear - Contribution Agreement#page53.tif source=Tab 01 Pollen Gear - Contribution Agreement#page53.tif source=Tab 01 Pollen Gear - Contribution Agreement#page54.tif source=Tab 01 Pollen Gear - Contribution Agreement#page55.tif source=Tab 01 Pollen Gear - Contribution Agreement#page55.tif source=Tab 01 Pollen Gear - Contribution Agreement#page56.tif source=Tab 01 Pollen Gear - Contribution Agreement#page61.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page64.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif source=Tab 01 Pollen Gear - Contribution Agreement#page66.tif source=Tab 01 Pollen Gear - Contribution Agreement#page67.tif source=Tab 01 Pollen Gear - Contribution Agreement#page67.tif	
source=Tab 01 Pollen Gear - Contribution Agreement#page40.tif source=Tab 01 Pollen Gear - Contribution Agreement#page41.tif source=Tab 01 Pollen Gear - Contribution Agreement#page42.tif source=Tab 01 Pollen Gear - Contribution Agreement#page43.tif source=Tab 01 Pollen Gear - Contribution Agreement#page44.tif source=Tab 01 Pollen Gear - Contribution Agreement#page45.tif source=Tab 01 Pollen Gear - Contribution Agreement#page45.tif source=Tab 01 Pollen Gear - Contribution Agreement#page46.tif source=Tab 01 Pollen Gear - Contribution Agreement#page48.tif source=Tab 01 Pollen Gear - Contribution Agreement#page49.tif source=Tab 01 Pollen Gear - Contribution Agreement#page50.tif source=Tab 01 Pollen Gear - Contribution Agreement#page51.tif source=Tab 01 Pollen Gear - Contribution Agreement#page52.tif source=Tab 01 Pollen Gear - Contribution Agreement#page53.tif source=Tab 01 Pollen Gear - Contribution Agreement#page54.tif source=Tab 01 Pollen Gear - Contribution Agreement#page55.tif source=Tab 01 Pollen Gear - Contribution Agreement#page55.tif source=Tab 01 Pollen Gear - Contribution Agreement#page56.tif source=Tab 01 Pollen Gear - Contribution Agreement#page57.tif source=Tab 01 Pollen Gear - Contribution Agreement#page58.tif source=Tab 01 Pollen Gear - Contribution Agreement#page60.tif source=Tab 01 Pollen Gear - Contribution Agreement#page61.tif source=Tab 01 Pollen Gear - Contribution Agreement#page61.tif source=Tab 01 Pollen Gear - Contribution Agreement#page62.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page64.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif source=Tab 01 Pollen Gear - Contribution Agreement#page66.tif source=Tab 01 Pollen Gear - Contribution Agreement#page66.tif	1
source=Tab 01 Pollen Gear - Contribution Agreement#page42.tif source=Tab 01 Pollen Gear - Contribution Agreement#page43.tif source=Tab 01 Pollen Gear - Contribution Agreement#page44.tif source=Tab 01 Pollen Gear - Contribution Agreement#page45.tif source=Tab 01 Pollen Gear - Contribution Agreement#page46.tif source=Tab 01 Pollen Gear - Contribution Agreement#page47.tif source=Tab 01 Pollen Gear - Contribution Agreement#page48.tif source=Tab 01 Pollen Gear - Contribution Agreement#page49.tif source=Tab 01 Pollen Gear - Contribution Agreement#page50.tif source=Tab 01 Pollen Gear - Contribution Agreement#page51.tif source=Tab 01 Pollen Gear - Contribution Agreement#page52.tif source=Tab 01 Pollen Gear - Contribution Agreement#page53.tif source=Tab 01 Pollen Gear - Contribution Agreement#page54.tif source=Tab 01 Pollen Gear - Contribution Agreement#page55.tif source=Tab 01 Pollen Gear - Contribution Agreement#page56.tif source=Tab 01 Pollen Gear - Contribution Agreement#page57.tif source=Tab 01 Pollen Gear - Contribution Agreement#page59.tif source=Tab 01 Pollen Gear - Contribution Agreement#page60.tif source=Tab 01 Pollen Gear - Contribution Agreement#page61.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page64.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif source=Tab 01 Pollen Gear - Contribution Agreement#page66.tif source=Tab 01 Pollen Gear - Contribution Agreement#page67.tif source=Tab 01 Pollen Gear - Contribution Agreement#page68.tif source=Tab 01 Pollen Gear - Contribution Agreement#page68.tif	
source=Tab 01 Pollen Gear - Contribution Agreement#page43.tif source=Tab 01 Pollen Gear - Contribution Agreement#page44.tif source=Tab 01 Pollen Gear - Contribution Agreement#page45.tif source=Tab 01 Pollen Gear - Contribution Agreement#page46.tif source=Tab 01 Pollen Gear - Contribution Agreement#page47.tif source=Tab 01 Pollen Gear - Contribution Agreement#page48.tif source=Tab 01 Pollen Gear - Contribution Agreement#page49.tif source=Tab 01 Pollen Gear - Contribution Agreement#page50.tif source=Tab 01 Pollen Gear - Contribution Agreement#page51.tif source=Tab 01 Pollen Gear - Contribution Agreement#page53.tif source=Tab 01 Pollen Gear - Contribution Agreement#page53.tif source=Tab 01 Pollen Gear - Contribution Agreement#page54.tif source=Tab 01 Pollen Gear - Contribution Agreement#page55.tif source=Tab 01 Pollen Gear - Contribution Agreement#page56.tif source=Tab 01 Pollen Gear - Contribution Agreement#page57.tif source=Tab 01 Pollen Gear - Contribution Agreement#page59.tif source=Tab 01 Pollen Gear - Contribution Agreement#page60.tif source=Tab 01 Pollen Gear - Contribution Agreement#page60.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif source=Tab 01 Pollen Gear - Contribution Agreement#page64.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif source=Tab 01 Pollen Gear - Contribution Agreement#page66.tif source=Tab 01 Pollen Gear - Contribution Agreement#page66.tif source=Tab 01 Pollen Gear - Contribution Agreement#page66.tif source=Tab 01 Pollen Gear - Contribution Agreement#page67.tif source=Tab 01 Pollen Gear - Contribution Agreement#page68.tif	source=Tab 01 Pollen Gear - Contribution Agreement#page41.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page44.tif source=Tab 01 Pollen Gear - Contribution Agreement#page45.tif source=Tab 01 Pollen Gear - Contribution Agreement#page46.tif source=Tab 01 Pollen Gear - Contribution Agreement#page47.tif source=Tab 01 Pollen Gear - Contribution Agreement#page48.tif source=Tab 01 Pollen Gear - Contribution Agreement#page49.tif source=Tab 01 Pollen Gear - Contribution Agreement#page50.tif source=Tab 01 Pollen Gear - Contribution Agreement#page51.tif source=Tab 01 Pollen Gear - Contribution Agreement#page52.tif source=Tab 01 Pollen Gear - Contribution Agreement#page53.tif source=Tab 01 Pollen Gear - Contribution Agreement#page54.tif source=Tab 01 Pollen Gear - Contribution Agreement#page55.tif source=Tab 01 Pollen Gear - Contribution Agreement#page56.tif source=Tab 01 Pollen Gear - Contribution Agreement#page57.tif source=Tab 01 Pollen Gear - Contribution Agreement#page59.tif source=Tab 01 Pollen Gear - Contribution Agreement#page59.tif source=Tab 01 Pollen Gear - Contribution Agreement#page60.tif source=Tab 01 Pollen Gear - Contribution Agreement#page61.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page64.tif source=Tab 01 Pollen Gear - Contribution Agreement#page64.tif source=Tab 01 Pollen Gear - Contribution Agreement#page64.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif source=Tab 01 Pollen Gear - Contribution Agreement#page66.tif source=Tab 01 Pollen Gear - Contribution Agreement#page66.tif source=Tab 01 Pollen Gear - Contribution Agreement#page68.tif	source=Tab 01 Pollen Gear - Contribution Agreement#page42.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page45.tif source=Tab 01 Pollen Gear - Contribution Agreement#page46.tif source=Tab 01 Pollen Gear - Contribution Agreement#page47.tif source=Tab 01 Pollen Gear - Contribution Agreement#page48.tif source=Tab 01 Pollen Gear - Contribution Agreement#page49.tif source=Tab 01 Pollen Gear - Contribution Agreement#page50.tif source=Tab 01 Pollen Gear - Contribution Agreement#page51.tif source=Tab 01 Pollen Gear - Contribution Agreement#page52.tif source=Tab 01 Pollen Gear - Contribution Agreement#page53.tif source=Tab 01 Pollen Gear - Contribution Agreement#page54.tif source=Tab 01 Pollen Gear - Contribution Agreement#page55.tif source=Tab 01 Pollen Gear - Contribution Agreement#page56.tif source=Tab 01 Pollen Gear - Contribution Agreement#page57.tif source=Tab 01 Pollen Gear - Contribution Agreement#page58.tif source=Tab 01 Pollen Gear - Contribution Agreement#page59.tif source=Tab 01 Pollen Gear - Contribution Agreement#page60.tif source=Tab 01 Pollen Gear - Contribution Agreement#page61.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page64.tif source=Tab 01 Pollen Gear - Contribution Agreement#page64.tif source=Tab 01 Pollen Gear - Contribution Agreement#page64.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif source=Tab 01 Pollen Gear - Contribution Agreement#page66.tif source=Tab 01 Pollen Gear - Contribution Agreement#page66.tif source=Tab 01 Pollen Gear - Contribution Agreement#page66.tif source=Tab 01 Pollen Gear - Contribution Agreement#page68.tif	source=Tab 01 Pollen Gear - Contribution Agreement#page43.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page46.tif source=Tab 01 Pollen Gear - Contribution Agreement#page47.tif source=Tab 01 Pollen Gear - Contribution Agreement#page48.tif source=Tab 01 Pollen Gear - Contribution Agreement#page49.tif source=Tab 01 Pollen Gear - Contribution Agreement#page50.tif source=Tab 01 Pollen Gear - Contribution Agreement#page51.tif source=Tab 01 Pollen Gear - Contribution Agreement#page52.tif source=Tab 01 Pollen Gear - Contribution Agreement#page53.tif source=Tab 01 Pollen Gear - Contribution Agreement#page53.tif source=Tab 01 Pollen Gear - Contribution Agreement#page55.tif source=Tab 01 Pollen Gear - Contribution Agreement#page56.tif source=Tab 01 Pollen Gear - Contribution Agreement#page57.tif source=Tab 01 Pollen Gear - Contribution Agreement#page59.tif source=Tab 01 Pollen Gear - Contribution Agreement#page59.tif source=Tab 01 Pollen Gear - Contribution Agreement#page60.tif source=Tab 01 Pollen Gear - Contribution Agreement#page61.tif source=Tab 01 Pollen Gear - Contribution Agreement#page62.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif source=Tab 01 Pollen Gear - Contribution Agreement#page66.tif source=Tab 01 Pollen Gear - Contribution Agreement#page68.tif	source=Tab 01 Pollen Gear - Contribution Agreement#page44.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page47.tif source=Tab 01 Pollen Gear - Contribution Agreement#page48.tif source=Tab 01 Pollen Gear - Contribution Agreement#page49.tif source=Tab 01 Pollen Gear - Contribution Agreement#page50.tif source=Tab 01 Pollen Gear - Contribution Agreement#page51.tif source=Tab 01 Pollen Gear - Contribution Agreement#page52.tif source=Tab 01 Pollen Gear - Contribution Agreement#page53.tif source=Tab 01 Pollen Gear - Contribution Agreement#page54.tif source=Tab 01 Pollen Gear - Contribution Agreement#page55.tif source=Tab 01 Pollen Gear - Contribution Agreement#page56.tif source=Tab 01 Pollen Gear - Contribution Agreement#page57.tif source=Tab 01 Pollen Gear - Contribution Agreement#page59.tif source=Tab 01 Pollen Gear - Contribution Agreement#page59.tif source=Tab 01 Pollen Gear - Contribution Agreement#page60.tif source=Tab 01 Pollen Gear - Contribution Agreement#page61.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page64.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif source=Tab 01 Pollen Gear - Contribution Agreement#page66.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif	source=Tab 01 Pollen Gear - Contribution Agreement#page45.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page48.tif source=Tab 01 Pollen Gear - Contribution Agreement#page49.tif source=Tab 01 Pollen Gear - Contribution Agreement#page50.tif source=Tab 01 Pollen Gear - Contribution Agreement#page51.tif source=Tab 01 Pollen Gear - Contribution Agreement#page52.tif source=Tab 01 Pollen Gear - Contribution Agreement#page53.tif source=Tab 01 Pollen Gear - Contribution Agreement#page54.tif source=Tab 01 Pollen Gear - Contribution Agreement#page55.tif source=Tab 01 Pollen Gear - Contribution Agreement#page56.tif source=Tab 01 Pollen Gear - Contribution Agreement#page57.tif source=Tab 01 Pollen Gear - Contribution Agreement#page58.tif source=Tab 01 Pollen Gear - Contribution Agreement#page69.tif source=Tab 01 Pollen Gear - Contribution Agreement#page60.tif source=Tab 01 Pollen Gear - Contribution Agreement#page61.tif source=Tab 01 Pollen Gear - Contribution Agreement#page62.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif source=Tab 01 Pollen Gear - Contribution Agreement#page66.tif source=Tab 01 Pollen Gear - Contribution Agreement#page68.tif	source=Tab 01 Pollen Gear - Contribution Agreement#page46.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page49.tif source=Tab 01 Pollen Gear - Contribution Agreement#page50.tif source=Tab 01 Pollen Gear - Contribution Agreement#page51.tif source=Tab 01 Pollen Gear - Contribution Agreement#page52.tif source=Tab 01 Pollen Gear - Contribution Agreement#page53.tif source=Tab 01 Pollen Gear - Contribution Agreement#page54.tif source=Tab 01 Pollen Gear - Contribution Agreement#page55.tif source=Tab 01 Pollen Gear - Contribution Agreement#page56.tif source=Tab 01 Pollen Gear - Contribution Agreement#page57.tif source=Tab 01 Pollen Gear - Contribution Agreement#page59.tif source=Tab 01 Pollen Gear - Contribution Agreement#page60.tif source=Tab 01 Pollen Gear - Contribution Agreement#page61.tif source=Tab 01 Pollen Gear - Contribution Agreement#page62.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page64.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif source=Tab 01 Pollen Gear - Contribution Agreement#page66.tif	source=Tab 01 Pollen Gear - Contribution Agreement#page47.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page50.tif source=Tab 01 Pollen Gear - Contribution Agreement#page51.tif source=Tab 01 Pollen Gear - Contribution Agreement#page52.tif source=Tab 01 Pollen Gear - Contribution Agreement#page53.tif source=Tab 01 Pollen Gear - Contribution Agreement#page54.tif source=Tab 01 Pollen Gear - Contribution Agreement#page55.tif source=Tab 01 Pollen Gear - Contribution Agreement#page56.tif source=Tab 01 Pollen Gear - Contribution Agreement#page57.tif source=Tab 01 Pollen Gear - Contribution Agreement#page58.tif source=Tab 01 Pollen Gear - Contribution Agreement#page59.tif source=Tab 01 Pollen Gear - Contribution Agreement#page60.tif source=Tab 01 Pollen Gear - Contribution Agreement#page61.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page64.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif source=Tab 01 Pollen Gear - Contribution Agreement#page66.tif source=Tab 01 Pollen Gear - Contribution Agreement#page67.tif source=Tab 01 Pollen Gear - Contribution Agreement#page68.tif	source=Tab 01 Pollen Gear - Contribution Agreement#page48.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page51.tif source=Tab 01 Pollen Gear - Contribution Agreement#page52.tif source=Tab 01 Pollen Gear - Contribution Agreement#page53.tif source=Tab 01 Pollen Gear - Contribution Agreement#page54.tif source=Tab 01 Pollen Gear - Contribution Agreement#page55.tif source=Tab 01 Pollen Gear - Contribution Agreement#page56.tif source=Tab 01 Pollen Gear - Contribution Agreement#page57.tif source=Tab 01 Pollen Gear - Contribution Agreement#page58.tif source=Tab 01 Pollen Gear - Contribution Agreement#page59.tif source=Tab 01 Pollen Gear - Contribution Agreement#page60.tif source=Tab 01 Pollen Gear - Contribution Agreement#page61.tif source=Tab 01 Pollen Gear - Contribution Agreement#page62.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page64.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif source=Tab 01 Pollen Gear - Contribution Agreement#page66.tif source=Tab 01 Pollen Gear - Contribution Agreement#page67.tif source=Tab 01 Pollen Gear - Contribution Agreement#page67.tif source=Tab 01 Pollen Gear - Contribution Agreement#page68.tif	source=Tab 01 Pollen Gear - Contribution Agreement#page49.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page52.tif source=Tab 01 Pollen Gear - Contribution Agreement#page53.tif source=Tab 01 Pollen Gear - Contribution Agreement#page54.tif source=Tab 01 Pollen Gear - Contribution Agreement#page55.tif source=Tab 01 Pollen Gear - Contribution Agreement#page56.tif source=Tab 01 Pollen Gear - Contribution Agreement#page57.tif source=Tab 01 Pollen Gear - Contribution Agreement#page58.tif source=Tab 01 Pollen Gear - Contribution Agreement#page59.tif source=Tab 01 Pollen Gear - Contribution Agreement#page60.tif source=Tab 01 Pollen Gear - Contribution Agreement#page61.tif source=Tab 01 Pollen Gear - Contribution Agreement#page62.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page64.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif source=Tab 01 Pollen Gear - Contribution Agreement#page66.tif source=Tab 01 Pollen Gear - Contribution Agreement#page67.tif source=Tab 01 Pollen Gear - Contribution Agreement#page67.tif source=Tab 01 Pollen Gear - Contribution Agreement#page67.tif source=Tab 01 Pollen Gear - Contribution Agreement#page68.tif	source=Tab 01 Pollen Gear - Contribution Agreement#page50.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page53.tif source=Tab 01 Pollen Gear - Contribution Agreement#page54.tif source=Tab 01 Pollen Gear - Contribution Agreement#page55.tif source=Tab 01 Pollen Gear - Contribution Agreement#page56.tif source=Tab 01 Pollen Gear - Contribution Agreement#page57.tif source=Tab 01 Pollen Gear - Contribution Agreement#page58.tif source=Tab 01 Pollen Gear - Contribution Agreement#page59.tif source=Tab 01 Pollen Gear - Contribution Agreement#page60.tif source=Tab 01 Pollen Gear - Contribution Agreement#page61.tif source=Tab 01 Pollen Gear - Contribution Agreement#page62.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page64.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif source=Tab 01 Pollen Gear - Contribution Agreement#page66.tif source=Tab 01 Pollen Gear - Contribution Agreement#page67.tif source=Tab 01 Pollen Gear - Contribution Agreement#page68.tif	source=Tab 01 Pollen Gear - Contribution Agreement#page51.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page54.tif source=Tab 01 Pollen Gear - Contribution Agreement#page55.tif source=Tab 01 Pollen Gear - Contribution Agreement#page56.tif source=Tab 01 Pollen Gear - Contribution Agreement#page57.tif source=Tab 01 Pollen Gear - Contribution Agreement#page58.tif source=Tab 01 Pollen Gear - Contribution Agreement#page59.tif source=Tab 01 Pollen Gear - Contribution Agreement#page60.tif source=Tab 01 Pollen Gear - Contribution Agreement#page61.tif source=Tab 01 Pollen Gear - Contribution Agreement#page62.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page64.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif source=Tab 01 Pollen Gear - Contribution Agreement#page66.tif source=Tab 01 Pollen Gear - Contribution Agreement#page67.tif source=Tab 01 Pollen Gear - Contribution Agreement#page67.tif source=Tab 01 Pollen Gear - Contribution Agreement#page68.tif	source=Tab 01 Pollen Gear - Contribution Agreement#page52.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page55.tif source=Tab 01 Pollen Gear - Contribution Agreement#page56.tif source=Tab 01 Pollen Gear - Contribution Agreement#page57.tif source=Tab 01 Pollen Gear - Contribution Agreement#page58.tif source=Tab 01 Pollen Gear - Contribution Agreement#page59.tif source=Tab 01 Pollen Gear - Contribution Agreement#page60.tif source=Tab 01 Pollen Gear - Contribution Agreement#page61.tif source=Tab 01 Pollen Gear - Contribution Agreement#page62.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page64.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif source=Tab 01 Pollen Gear - Contribution Agreement#page66.tif source=Tab 01 Pollen Gear - Contribution Agreement#page67.tif source=Tab 01 Pollen Gear - Contribution Agreement#page67.tif source=Tab 01 Pollen Gear - Contribution Agreement#page68.tif	source=Tab 01 Pollen Gear - Contribution Agreement#page53.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page56.tif source=Tab 01 Pollen Gear - Contribution Agreement#page57.tif source=Tab 01 Pollen Gear - Contribution Agreement#page58.tif source=Tab 01 Pollen Gear - Contribution Agreement#page59.tif source=Tab 01 Pollen Gear - Contribution Agreement#page60.tif source=Tab 01 Pollen Gear - Contribution Agreement#page61.tif source=Tab 01 Pollen Gear - Contribution Agreement#page62.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page64.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif source=Tab 01 Pollen Gear - Contribution Agreement#page66.tif source=Tab 01 Pollen Gear - Contribution Agreement#page67.tif source=Tab 01 Pollen Gear - Contribution Agreement#page68.tif	
source=Tab 01 Pollen Gear - Contribution Agreement#page57.tif source=Tab 01 Pollen Gear - Contribution Agreement#page58.tif source=Tab 01 Pollen Gear - Contribution Agreement#page59.tif source=Tab 01 Pollen Gear - Contribution Agreement#page60.tif source=Tab 01 Pollen Gear - Contribution Agreement#page61.tif source=Tab 01 Pollen Gear - Contribution Agreement#page62.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page64.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif source=Tab 01 Pollen Gear - Contribution Agreement#page66.tif source=Tab 01 Pollen Gear - Contribution Agreement#page67.tif source=Tab 01 Pollen Gear - Contribution Agreement#page67.tif source=Tab 01 Pollen Gear - Contribution Agreement#page68.tif	source=Tab 01 Pollen Gear - Contribution Agreement#page55.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page58.tif source=Tab 01 Pollen Gear - Contribution Agreement#page59.tif source=Tab 01 Pollen Gear - Contribution Agreement#page60.tif source=Tab 01 Pollen Gear - Contribution Agreement#page61.tif source=Tab 01 Pollen Gear - Contribution Agreement#page62.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page64.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif source=Tab 01 Pollen Gear - Contribution Agreement#page66.tif source=Tab 01 Pollen Gear - Contribution Agreement#page67.tif source=Tab 01 Pollen Gear - Contribution Agreement#page68.tif	1
source=Tab 01 Pollen Gear - Contribution Agreement#page59.tif source=Tab 01 Pollen Gear - Contribution Agreement#page60.tif source=Tab 01 Pollen Gear - Contribution Agreement#page61.tif source=Tab 01 Pollen Gear - Contribution Agreement#page62.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page64.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif source=Tab 01 Pollen Gear - Contribution Agreement#page66.tif source=Tab 01 Pollen Gear - Contribution Agreement#page67.tif source=Tab 01 Pollen Gear - Contribution Agreement#page68.tif	
source=Tab 01 Pollen Gear - Contribution Agreement#page60.tif source=Tab 01 Pollen Gear - Contribution Agreement#page61.tif source=Tab 01 Pollen Gear - Contribution Agreement#page62.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page64.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif source=Tab 01 Pollen Gear - Contribution Agreement#page66.tif source=Tab 01 Pollen Gear - Contribution Agreement#page67.tif source=Tab 01 Pollen Gear - Contribution Agreement#page68.tif	
source=Tab 01 Pollen Gear - Contribution Agreement#page61.tif source=Tab 01 Pollen Gear - Contribution Agreement#page62.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page64.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif source=Tab 01 Pollen Gear - Contribution Agreement#page66.tif source=Tab 01 Pollen Gear - Contribution Agreement#page67.tif source=Tab 01 Pollen Gear - Contribution Agreement#page68.tif	
source=Tab 01 Pollen Gear - Contribution Agreement#page62.tif source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page64.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif source=Tab 01 Pollen Gear - Contribution Agreement#page66.tif source=Tab 01 Pollen Gear - Contribution Agreement#page67.tif source=Tab 01 Pollen Gear - Contribution Agreement#page68.tif	
source=Tab 01 Pollen Gear - Contribution Agreement#page63.tif source=Tab 01 Pollen Gear - Contribution Agreement#page64.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif source=Tab 01 Pollen Gear - Contribution Agreement#page66.tif source=Tab 01 Pollen Gear - Contribution Agreement#page67.tif source=Tab 01 Pollen Gear - Contribution Agreement#page68.tif	1
source=Tab 01 Pollen Gear - Contribution Agreement#page64.tif source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif source=Tab 01 Pollen Gear - Contribution Agreement#page66.tif source=Tab 01 Pollen Gear - Contribution Agreement#page67.tif source=Tab 01 Pollen Gear - Contribution Agreement#page68.tif	1
source=Tab 01 Pollen Gear - Contribution Agreement#page65.tif source=Tab 01 Pollen Gear - Contribution Agreement#page66.tif source=Tab 01 Pollen Gear - Contribution Agreement#page67.tif source=Tab 01 Pollen Gear - Contribution Agreement#page68.tif	1
source=Tab 01 Pollen Gear - Contribution Agreement#page66.tif source=Tab 01 Pollen Gear - Contribution Agreement#page67.tif source=Tab 01 Pollen Gear - Contribution Agreement#page68.tif	
source=Tab 01 Pollen Gear - Contribution Agreement#page67.tif source=Tab 01 Pollen Gear - Contribution Agreement#page68.tif	
source=Tab 01 Pollen Gear - Contribution Agreement#page68.tif	1
l	
source= rab 01 Pollen Gear - Contribution Agreement#page69.tif	1
l .	source= Lab UT Pollen Gear - Contribution Agreement#page69.tif

source=Tab 01 Pollen Gear - Contribution Agreement#page70.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page71.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page72.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page73.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page74.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page75.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page76.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page77.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page78.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page79.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page80.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page81.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page82.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page83.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page84.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page85.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page86.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page87.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page88.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page89.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page90.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page91.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page92.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page93.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page94.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page95.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page96.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page97.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page98.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page99.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page100.tif
source=Tab 01 Pollen Gear - Contribution Agreement#page101.tif
source=Tab 02 Disclosure Schedules#page1.tif
source=Tab 02 Disclosure Schedules#page2.tif
source=Tab 02 Disclosure Schedules#page3.tif
source=Tab 02 Disclosure Schedules#page4.tif
source=Tab 02 Disclosure Schedules#page5.tif
source=Tab 02 Disclosure Schedules#page6.tif
source=Tab 02 Disclosure Schedules#page7.tif
source=Tab 02 Disclosure Schedules#page8.tif
source=Tab 02 Disclosure Schedules#page9.tif
source=Tab 02 Disclosure Schedules#page10.tif
source=Tab 02 Disclosure Schedules#page11.tif
source=Tab 02 Disclosure Schedules#page12.tif
source=Tab 02 Disclosure Schedules#page13.tif
source=Tab 02 Disclosure Schedules#page14.tif
source=Tab 02 Disclosure Schedules#page15.tif
source=Tab 02 Disclosure Schedules#page16.tif
DATELIT

source=Tab 02 Disclosure Schedules#page17.tif	
source=Tab 02 Disclosure Schedules#page18.tif	
source=Tab 02 Disclosure Schedules#page19.tif	
source=Tab 02 Disclosure Schedules#page20.tif	
source=Tab 02 Disclosure Schedules#page21.tif	
source=Tab 02 Disclosure Schedules#page22.tif	
source=Tab 02 Disclosure Schedules#page23.tif	
source=Tab 02 Disclosure Schedules#page24.tif	
source=Tab 02 Disclosure Schedules#page25.tif	
source=Tab 02 Disclosure Schedules#page26.tif	
source=Tab 02 Disclosure Schedules#page27.tif	
source=Tab 02 Disclosure Schedules#page28.tif	
source=Tab 02 Disclosure Schedules#page29.tif	
source=Tab 02 Disclosure Schedules#page30.tif	
source=Tab 02 Disclosure Schedules#page31.tif	
source=Tab 02 Disclosure Schedules#page32.tif	
source=Tab 02 Disclosure Schedules#page33.tif	
source=Tab 02 Disclosure Schedules#page34.tif	
source=Tab 02 Disclosure Schedules#page35.tif	
I and the second	

CONTRIBUTION AGREEMENT
DATED AS OF JANUARY 4, 2019
BY AND AMONG
GREENLANE HOLDINGS, LLC,
POLLEN GEAR LLC

AND

POLLEN GEAR HOLDINGS LLC

TABLE OF CONTENTS

		Page
ARTICLE I	CONTRIBUTION OF THE CONTRIBUTED INTERESTS	1
1.01.	Contribution of the Contributed Interests.	1
1.01.	Contribution Consideration	
1.02.	Withholding	
1.03.	Tax Treatment	
1.0-1.	TUA TTUUINGII	
ARTICLE I	I <u>CLOSING</u>	2
2.01.	Closing Date	2
	Deliveries by the Seller.	
2.03.	<u>Deliveries by the Purchaser</u> .	4
ARTICLE I	II REPRESENTATIONS AND WARRANTIES OF THE SELLER	4
3.01.	Organization; Power; Capacity	4
3.02.	Authorization and Validity of Agreement	
3.03.	Title to the Contributed Interests.	
3.04.	No Conflict; Required Filings and Consents	5
3.05.	Litigation.	
3.06.	Acquisition for Own Account.	5
3.07.	Seller Diligence	5
3.08.	Accredited Investor	
3.09.	No Bad Actor Disqualification.	
3.10.	Limited Liquidity; Economic Risk.	
3.11.	Anti-Terrorism and Money Laundering Activities	
3.12.	Broker's and Finder's Fees.	
3.13.	No Reliance on IPO	
3.14.	Seller Capitalization	7
	V REPRESENTATIONS AND WARRANTIES REGARDING THE	
COM	PANY	7
4.01.	Organization; Power.	7
4.02.	<u>Capitalization</u> .	7
4.03.	No Conflict or Violation.	
4.04.	Consents and Approvals.	
4.05.	Financial Statements; No Undisclosed Liabilities.	9
4.06.	Tax Matters.	
4.07.	Absence of Certain Changes	
4.08.	Owned Real Property	
4.09.	<u>Leased Real Property</u>	
4.10.	Assets.	
4.11.	Accounts Receivable	13

i

4.12.	<u>Intellectual Property</u> .	13
4.13.	Employee Benefit Plans	16
4.14.	Personnel; Labor Relations.	18
4.15.	Environmental Compliance.	19
4.16.	<u>Licenses and Permits</u>	20
4.17.	<u>Insurance</u> .	
4.18.	Payment Card Standards	21
4.19.	Contracts and Commitments.	
4.20.	<u>Customers and Suppliers</u> .	
4.21.	Compliance with Law.	23
4.22.	<u>Litigation</u>	
4.23.	<u>Title to and Sufficiency of Assets and Related Matters</u>	
4.24.	Broker's and Finder's Fees.	
4.25.	Affiliate Transactions.	24
4.26.	<u>Inventory</u> .	
4.27.	Product Matters.	
4.28.	Bank Accounts	
4.29.	Plans and Designs.	
4.30.	<u>Privacy</u>	
4.31.	<u>Full Disclosure</u>	25
	REPRESENTATIONS AND WARRANTIES OF THE PURCHASER	
5.01.	Organization; Power	26
5.01. 5.02.		
	Title to the Greenlane Interests.	26
5.02.		26 26
5.02. 5.03.	Title to the Greenlane Interests. Authorization and Validity of Agreement.	26 26
5.02. 5.03. 5.04.	Title to the Greenlane Interests Authorization and Validity of Agreement No Conflict or Violation	26 26 26
5.02. 5.03. 5.04. 5.05.	Title to the Greenlane Interests. Authorization and Validity of Agreement. No Conflict or Violation. SEC Documents; Financial Statements.	26 26 26
5.02. 5.03. 5.04. 5.05. 5.06.	Title to the Greenlane Interests. Authorization and Validity of Agreement. No Conflict or Violation. SEC Documents; Financial Statements. Broker's and Finder's Fees.	26 26 26 27
5.02. 5.03. 5.04. 5.05. 5.06. 5.07. 5.08.	Title to the Greenlane Interests Authorization and Validity of Agreement No Conflict or Violation SEC Documents; Financial Statements Broker's and Finder's Fees Consents and Approvals	26 26 26 27 27
5.02. 5.03. 5.04. 5.05. 5.06. 5.07. 5.08.	Title to the Greenlane Interests. Authorization and Validity of Agreement. No Conflict or Violation. SEC Documents; Financial Statements. Broker's and Finder's Fees. Consents and Approvals. Independent Investigation.	26 26 27 27 27
5.02. 5.03. 5.04. 5.05. 5.06. 5.07. 5.08.	Title to the Greenlane Interests. Authorization and Validity of Agreement. No Conflict or Violation. SEC Documents; Financial Statements. Broker's and Finder's Fees. Consents and Approvals. Independent Investigation. I INDEMNIFICATION; SURVIVAL.	26 26 27 27 27
5.02. 5.03. 5.04. 5.05. 5.06. 5.07. 5.08. ARTICLE V	Title to the Greenlane Interests. Authorization and Validity of Agreement. No Conflict or Violation. SEC Documents; Financial Statements. Broker's and Finder's Fees. Consents and Approvals. Independent Investigation. INDEMNIFICATION; SURVIVAL. Indemnification by the Seller.	26 26 27 27 27 28 28
5.02. 5.03. 5.04. 5.05. 5.06. 5.07. 5.08. ARTICLE V 6.01. 6.02.	Title to the Greenlane Interests. Authorization and Validity of Agreement. No Conflict or Violation. SEC Documents; Financial Statements. Broker's and Finder's Fees. Consents and Approvals. Independent Investigation. INDEMNIFICATION; SURVIVAL. Indemnification by the Seller. Indemnification by the Purchaser.	26 26 27 27 27 28 28
5.02. 5.03. 5.04. 5.05. 5.06. 5.07. 5.08. ARTICLE V 6.01. 6.02. 6.03.	Title to the Greenlane Interests. Authorization and Validity of Agreement. No Conflict or Violation. SEC Documents; Financial Statements. Broker's and Finder's Fees. Consents and Approvals. Independent Investigation. INDEMNIFICATION; SURVIVAL. Indemnification by the Seller. Indemnification by the Purchaser. Indemnification Notice; Litigation Notice. Defense of Third Party Claims. Survival.	26 26 27 27 27 28 28 28 28
5.02. 5.03. 5.04. 5.05. 5.06. 5.07. 5.08. ARTICLE V 6.01. 6.02. 6.03. 6.04.	Title to the Greenlane Interests. Authorization and Validity of Agreement. No Conflict or Violation. SEC Documents; Financial Statements. Broker's and Finder's Fees. Consents and Approvals. Independent Investigation. INDEMNIFICATION; SURVIVAL. Indemnification by the Seller. Indemnification by the Purchaser. Indemnification Notice; Litigation Notice. Defense of Third Party Claims.	26 26 27 27 27 28 28 28 28
5.02. 5.03. 5.04. 5.05. 5.06. 5.07. 5.08. ARTICLE V 6.01. 6.02. 6.03. 6.04. 6.05.	Title to the Greenlane Interests. Authorization and Validity of Agreement. No Conflict or Violation. SEC Documents; Financial Statements. Broker's and Finder's Fees. Consents and Approvals. Independent Investigation. INDEMNIFICATION; SURVIVAL. Indemnification by the Seller. Indemnification by the Purchaser. Indemnification Notice; Litigation Notice. Defense of Third Party Claims. Survival.	26 26 27 27 28 28 28 30
5.02. 5.03. 5.04. 5.05. 5.06. 5.07. 5.08. ARTICLE V 6.01. 6.02. 6.03. 6.04. 6.05. 6.06.	Title to the Greenlane Interests. Authorization and Validity of Agreement. No Conflict or Violation. SEC Documents; Financial Statements. Broker's and Finder's Fees. Consents and Approvals. Independent Investigation. INDEMNIFICATION; SURVIVAL. Indemnification by the Seller. Indemnification by the Purchaser. Indemnification Notice; Litigation Notice. Defense of Third Party Claims. Survival. Additional Indemnification Provisions. Special Rule for Fraud. Sole Remedy.	26 26 27 27 28 28 30 31 31
5.02. 5.03. 5.04. 5.05. 5.06. 5.07. 5.08. ARTICLE V 6.01. 6.02. 6.03. 6.04. 6.05. 6.06. 6.07.	Title to the Greenlane Interests. Authorization and Validity of Agreement No Conflict or Violation. SEC Documents; Financial Statements. Broker's and Finder's Fees. Consents and Approvals. Independent Investigation. INDEMNIFICATION; SURVIVAL. Indemnification by the Seller. Indemnification by the Purchaser. Indemnification Notice; Litigation Notice. Defense of Third Party Claims. Survival. Additional Indemnification Provisions. Special Rule for Fraud. Sole Remedy. Determination of Loss Amount.	26262727282830313132
5.02. 5.03. 5.04. 5.05. 5.06. 5.07. 5.08. ARTICLE V 6.01. 6.02. 6.03. 6.04. 6.05. 6.06. 6.07. 6.08.	Title to the Greenlane Interests. Authorization and Validity of Agreement. No Conflict or Violation. SEC Documents; Financial Statements. Broker's and Finder's Fees. Consents and Approvals. Independent Investigation. INDEMNIFICATION; SURVIVAL. Indemnification by the Seller. Indemnification by the Purchaser. Indemnification Notice; Litigation Notice. Defense of Third Party Claims. Survival. Additional Indemnification Provisions. Special Rule for Fraud. Sole Remedy.	26262727282830313132

7.01.	Conduct of the Business.	32
7.02.	Confidential Information	33
7.03.	Transfer Taxes.	34
7.04.	Preparation of Tax Returns; Payment of Taxes.	34
7.05.	Cooperation on Tax Matters	35
7.06.	Tax Contests	35
7.07.	Release.	36
7.08.	Employees; Employee Benefits	36
7.09.	Non-Competition; Non-Solicitation.	37
7.10.	Remedies	38
ADDICEES		20
ARTICLE V	TII CONDITIONS TO CLOSING	39
8.01.	Conditions to the Obligations of the Purchaser.	30
	Conditions to the Obligations of the Seller and the Company.	
0.02.	Conditions to the Obligations of the Scher and the Company.	
ARTICLE I	X <u>TERMINATION</u>	40
9.01.	<u>Termination</u>	
9.02.	Effect of Termination.	41
ARTICLE X	MISCELLANEOUS	41
	Public Announcements.	
10.02	Costs and Expenses.	42
	Further Assurances.	
10.04	Addresses for Notices, Etc.	42
10.05	Headings	43
	Construction	
10.07	Severability	44
	Entire Agreement and Amendment.	
	No Waiver; Cumulative Remedies.	
	Parties in Interest.	
	Successors and Assigns; Assignment.	
	Governing Law; Jurisdiction and Venue.	
	Waiver of Jury Trial	
	Counterparts.	
10.15	Privileged and Confidential Information	45

EXHIBITS

Definitions Exhibit A Seller Capitalization
Kilduff Employment Agreement
Brown Employment Agreement Exhibit B Exhibit C-1

Exhibit C-2

CONTRIBUTION AGREEMENT

THIS CONTRIBUTION AGREEMENT (this "Agreement"), dated as of the 4th day of January, 2019, is made and entered into by and among Greenlane Holdings, LLC, a Delaware limited liability company (the "Purchaser"), Pollen Gear Holdings LLC, a California limited liability company (the "Seller") and Pollen Gear LLC, a Delaware limited liability company (the "Company"). Capitalized terms used, but not defined herein shall have the meanings ascribed to them in Exhibit A attached hereto.

WITNESSETH:

WHEREAS, the Seller is the owner of all of the issued and outstanding limited liability company membership interests (the "*Contributed Interests*") of the Company;

WHEREAS, the Seller desires to contribute the Contributed Interests to the Purchaser, and the Purchaser desires to accept the Contributed Interests from the Seller and, in exchange therefor, to issue to the Seller the Greenlane Interests (defined below) pursuant to the terms and conditions set forth herein;

NOW, THEREFORE, in consideration of the foregoing and the mutual covenants and promises contained in this Agreement and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto, intending to be legally bound, hereby agree as follows:

AGREEMENT

ARTICLE I CONTRIBUTION OF THE CONTRIBUTED INTERESTS

- 1.01. Contribution of the Contributed Interests. On the terms and subject to the conditions set forth in this Agreement and on the basis of the representations, warranties, covenants and agreements herein contained, the Purchaser hereby receives, acquires and accepts from the Seller, and the Seller hereby contributes, transfers, assigns, conveys and delivers to the Purchaser, all of the Seller's right, title and interest in and to the Contributed Interests to be contributed by the Seller pursuant to this Agreement, in accordance with the terms of the Governing Documents of the Company, free and clear of all Encumbrances (other than any restrictions under the Securities Act or applicable state securities Laws).
- **1.02.** Contribution Consideration. As consideration for the contribution and transfer of the Contributed Interests, the Purchaser shall issue to the Seller (the "Contribution Consideration") membership interests representing in the aggregate four percent (4.0%) of the issued and outstanding membership interests of the Purchaser (the "Greenlane Interests"), in accordance with the terms of the Governing Documents of the Purchaser and free and clear of all Encumbrances (other than any restrictions under the Securities Act or applicable state securities Laws or as set forth in the Purchaser's Governing Documents).
- **1.03.** Withholding. Notwithstanding any provision contained herein to the contrary, the Purchaser shall be entitled to deduct and withhold from any amounts payable to the

Seller pursuant to this Agreement to the extent required under any provision of Tax Law. If the Purchaser so deducts or withholds amounts, such amounts shall be treated for all purposes of this Agreement as having been paid to the Seller in respect of which such deduction and withholding was made.

1.04. Tax Treatment. It is the intent of the Purchaser and the Seller that, for U.S. federal income tax purposes, the contribution by the Seller of the Contributed Interests in exchange for the Greenlane Interests be governed by Section 721 of the Code. The parties shall not take any action or position inconsistent with this Section 1.04, except as otherwise required by Law.

ARTICLE II CLOSING

- **2.01.** Closing Date. The closing of the transactions contemplated by this Agreement (the "Closing") shall take place on the date that is the third (3rd) Business Day following the satisfaction or waiver of the conditions specified in Article VIII (other than conditions with respect to actions the respective parties shall take at the Closing itself, but otherwise subject to the satisfaction thereof at the Closing), or such other date as the parties may mutually agree to in writing (the day on which the Closing takes place, the "Closing Date"), effective as of 11:59 p.m. on the Closing Date. In lieu of an in-person Closing, the Closing may instead be accomplished by electronic transmission to the respective offices of legal counsel for the parties of the requisite documents, duly executed where required, delivered upon actual confirmed receipt. The parties hereto acknowledge and agree that all proceedings to be taken and all documents to be executed and delivered by all parties at the Closing shall be deemed to have been taken and executed simultaneously on the Closing Date, and no proceedings shall be deemed taken nor any documents executed or delivered until all have been taken, executed and delivered.
- **2.02.** <u>Deliveries by the Seller</u>. At the Closing, the Seller shall deliver (or cause to be delivered) to the Purchaser originals or copies, if specified, of the following:
- (a) assignments or other instruments of transfer with respect to the Contributed Interests as requested by the Purchaser, duly endorsed in blank, and, to the extent the Contributed Interests are certificated, certificates representing the Contributed Interests;
- (b) a joinder to the amended and restated operating agreement of the Purchaser dated as of the date hereof (the "Greenlane Operating Agreement"), in form and substance satisfactory to the Purchaser, in its reasonable discretion, duly executed by the Seller;
- (c) counterparts of all other agreements, documents and instruments required to be delivered by the Seller pursuant to this Agreement or any of the Related Agreements, duly executed by the Seller;
- (d) copies of each consent, waiver, authorization and approval required in connection with the consummation of the transactions contemplated hereby, including those contemplated pursuant to <u>Section 4.04</u> of this Agreement, in each case, in form and substance satisfactory to the Purchaser;

2

- (e) a Certificate of Good Standing of the Company issued by the Secretary of State of the State of Delaware and of each other state or jurisdiction in which the Company is qualified to do business, dated within five (5) Business Days of the Closing Date;
- (f) a Certificate of Good Standing of the Seller issued by the Secretary of State of the State of Delaware and of each other state or jurisdiction in which the Company is qualified to do business, dated within five (5) Business Days of the Closing Date;
- (g) a copy of all Governing Documents of the Company, including: (i) the Certificate of Formation or similar document of the Company, together with all amendments thereto, certified as true, complete and correct by the Secretary of State of the State of Delaware; and (ii) the Amended and Restated Operating Agreement of the Company, together with all amendments thereto and/or restatements thereof certified as true, complete and correct and in full force and effect by the Manager of the Company;
- (h) a copy of all Governing Documents of the Seller, including: (i) the Articles of Organization or similar document of the Seller, together with all amendments thereto, certified as true, complete and correct by the Secretary of State of the State of California; and (ii) the Operating Agreement of the Seller, together with all amendments thereto and/or restatements thereof certified as true, complete and correct and in full force and effect by the Manager of the Company;
- (i) IRS Forms W-9 from the Seller and an affidavit dated as of the Closing Date in form and substance reasonably satisfactory to the Purchaser, sworn under penalty of perjury and in form and substance required under the Treasury Regulations issued pursuant to Section 1445 of the Code stating that the Seller is not a "foreign person" as defined in Section 1445 of the Code;
- (j) evidence in form and substance reasonably satisfactory to the Purchaser of the consent by the Landlord to the assignment of the Standard Industrial/Commercial Multi-Tenant Lease for the 601 Cypress Avenue, Hermosa Beach, CA property;
- (k) evidence in form and substance reasonably satisfactory to the Purchaser of the conversion of all outstanding Convertible Notes of the Company into units of the Seller, and the discharge of all outstanding Debt in connection with such Convertible Notes, in each case, prior to the Closing;
- (l) evidence of revised bank access documentation providing Aaron LoCascio and Adam Schoenfeld with access and control upon Closing over the accounts referenced in Schedule 4.28 from and after Closing;
- (m) employment agreements duly executed by each of Edward Kilduff and Jason Brown, in substantially the form attached hereto as <u>Exhibit C-1</u> and <u>C-2</u>, respectively (collectively, the "*Employment Agreements*");
- (n) an assignment of all Intellectual Property owned by any of the Employees that is used in or necessary to the Business as operated or proposed to be operated by the Company; and

- (o) all other documentation reasonably requested by the Purchaser.
- **2.03.** <u>Deliveries by the Purchaser</u>. At the Closing, the Purchaser shall deliver (or cause to be delivered) originals or copies, if specified, of the following agreements, documents and other items:
 - (a) the Employment Agreements duly executed by the Purchaser;
- (b) counterparts of all agreements, documents and instruments required to be delivered by the Purchaser pursuant to this Agreement or any of the Related Agreements, duly executed by the Purchaser;
- (c) certificates to the Seller representing the Greenlane Interests, to the extent certificated representing, in the aggregate, four percent (4.0%) of the outstanding Equity Securities of the Purchaser;
- (d) copies of each consent, waiver, authorization and approval required in connection with the consummation of the transactions contemplated hereby, including those contemplated pursuant to <u>Section 5.07</u> of this Agreement, in each case, in form and substance satisfactory to the Seller;
 - (e) all other documentation reasonably requested by the Seller.

ARTICLE III REPRESENTATIONS AND WARRANTIES OF THE SELLER

The Seller hereby represents and warrants to the Purchaser that the statements contained in this Article III are true, correct and complete as of the date hereof.

- **3.01.** Organization; Power; Capacity. Seller is a limited liability company duly organized, validly existing and in good standing under the laws of its jurisdiction of formation, organization or incorporation, as applicable. The Seller has all requisite power and authority, and legal capacity, to: (a) execute and deliver this Agreement and the Related Agreements to which the Seller is a party; (b) to carry out the Seller's obligations hereunder and thereunder; (c) to comply with and fulfill the terms and conditions of this Agreement and the Related Agreements to which the Seller is a party; and (d) to consummate the transactions contemplated hereby and thereby.
- **3.02.** Authorization and Validity of Agreement. The execution, delivery and performance of this Agreement and the other Related Agreements to which the Seller is a party have been duly authorized by the Seller. This Agreement has been, and each other Related Agreement to which the Seller is a party has been, duly executed and delivered by the Seller and constitutes the Seller's valid and binding obligation, enforceable against the Seller in accordance with its terms and conditions, except as enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar Laws affecting the enforcement of creditor's rights generally or by general principles of equity (whether applied in a Proceeding at Law or equity).

4

- 3.03. <u>Title to the Contributed Interests</u>. The Seller has good and marketable title to, and is the sole record and beneficial owner of, all of the Contributed Interests, free and clear of all Encumbrances or any restrictions on transfer (other than any restrictions under the Securities Act or applicable state securities Laws or restrictions contained in the Governing Documents of the Company). The Seller has complete and unrestricted power and the unqualified right to contribute, convey, assign, transfer and deliver the Contributed Interests, and the instruments of assignment and transfer to be executed and delivered by the Seller to the Purchaser at the Closing shall be valid and binding obligations of the Seller. At the Closing, the Seller shall transfer to the Purchaser good and marketable title to the Contributed Interests to be sold by the Seller pursuant to this Agreement, free and clear of all Encumbrances or any restrictions on transfer, other than any restrictions under the Securities Act or applicable state securities Laws.
- 3.04. No Conflict; Required Filings and Consents. The execution, delivery and consummation of this Agreement by the Seller does not, and the execution, delivery and consummation of the Related Agreements to which the Seller is a party and the performance of this Agreement and such Related Agreements will not: (a) violate any Law applicable to the Seller or which affects the Contributed Interests to be contributed by the Seller; (b)(i) require any consent or approval other than as set forth in this Agreement or (ii) violate or result in any breach of or constitute (with or without due notice or the passage of time or both) a default under any judicial consent, order, decree or any Contract to which the Seller is a party or to which the Contributed Interests to be sold by the Seller are subject; or (c) result in the imposition of any Encumbrance or restriction on the Seller's Contributed Interests (with or without due notice or the passage of time or both).
- 3.05. <u>Litigation</u>. To the Seller's Knowledge, there are no Proceedings pending or threatened against or affecting the Seller, which would affect the ability of the Seller to consummate the transactions contemplated by this Agreement or any Related Agreement to which the Seller is a party. There are no currently existing events, facts or circumstances which could reasonably be expected to form the basis for any Proceeding or order, or decree of any court or Governmental Entity which would affect the ability of the Seller to consummate the transactions contemplated by this Agreement or any Related Agreement to which the Seller is a party.
- 3.06. Acquisition for Own Account. The Seller is acquiring the Greenlane Interests for the Seller's own account, or the account of another Seller or Person identified to the Purchaser in writing, as principal, not as a nominee or agent, for investment purposes only, and not with a view to, or for, resale, distribution or fractionalization thereof in whole or in part and no other Person has a direct or indirect beneficial interest in such Greenlane Interests. Further, the Seller does not have any Contract, undertaking, agreement or arrangement with any Person to sell, transfer or grant participations to such Person or to any third person, with respect to any of the Greenlane Interests other than as may be set forth in the Greenlane Operating Agreement.
- **3.07.** Seller Diligence. The Seller has been given the opportunity for a reasonable time prior to the date hereof to ask questions of, and receive answers from, the Purchaser or its representatives concerning the terms and conditions of this Agreement, and other matters pertaining to its receipt of the Greenlane Interests, and has been given the opportunity for a reasonable time prior to the date hereof to obtain such additional information in connection with the Purchaser in order for the Seller to evaluate the merits and risks of its acquisition of the

Greenlane Interests, to the extent the Purchaser possesses such information or can acquire it without unreasonable effort or expense. The Seller has such knowledge and experience in financial and business matters as to be capable of evaluating the merits and risks of holding the Greenlane Interests.

- **3.08.** <u>Accredited Investor</u>. The Seller is an "accredited investor" within the meaning of Rule 501 of Regulation D promulgated under the Securities Act. The Seller agrees to furnish any additional information requested by the Purchaser or any of its Affiliates to assure compliance with applicable U.S. federal and state securities Laws in connection with the acquisition and transfer of the Greenlane Interests.
- **3.09.** No Bad Actor Disqualification. The Seller and its Rule 506(d) Related Parties (defined below) are not subject to any "Bad Actor" disqualifying events described in Rule 506(d)(1)(i) to (viii) under the Securities Act (a "Disqualification Event"), except, if applicable, for a Disqualification Event as to which Rule 506(d)(2)(ii) or (iii) or (d)(3) is applicable. The Seller hereby agrees that it shall notify the Purchaser promptly in writing in the event a Disqualification Event becomes applicable to the Seller or any of its Rule 506(d) Related Parties, except, if applicable, for a Disqualification Event as to which Rule 506(d)(2)(ii) or (iii) or (d)(3) is applicable. For purposes of this Section 3.09, "Rule 506(d) Related Party" shall mean a Person that is a beneficial owner of the Seller's securities for purposes of Rule 506(d) of the Securities Act.
- 3.10. <u>Limited Liquidity; Economic Risk</u>. The Seller understands that (a) it may not be able to readily resell any of the Greenlane Interests acquired by the Seller pursuant to this Agreement because (i) there may only be a limited market, if any exists, for any of the Greenlane Interests and (ii) none of the Greenlane Interests have been registered under the Securities Act or any applicable state "blue sky" Laws; and (b) the Purchaser has the absolute right to refuse to consent to the transfer or assignment of the Greenlane Interests if such transfer or assignment does not comply with applicable Laws, including the Securities Act and any applicable state "blue sky" Laws, or the Governing Documents of the Purchaser. The Seller has the financial ability to bear the economic risk of a total diminution in value of the Greenlane Interests, has adequate means for providing for the Seller's current needs and personal contingencies and has no need for liquidity with respect to the Seller's acquisition of the Greenlane Interests.
- 3.11. <u>Anti-Terrorism and Money Laundering Activities</u>. The Seller acknowledges that the Purchaser is required by United States Federal Law to obtain, verify and record information that identifies each Person who makes contributions to the Purchaser in consideration for the Greenlane Interests. The Seller acknowledges and agrees that it will furnish to the Purchaser upon request a copy of the Seller's identifying documents that will assist the Purchaser to properly identify the Seller as required by Federal Law. Such documents may include, without limitation, a copy of the Seller's Governing Documents and evidence of the authority of the Person executing this Agreement on behalf of such the Seller that such Person has full authority to execute and deliver this Agreement on behalf of the Seller and otherwise to act on behalf of the Seller in connection with the Seller's receipt of the Greenlane Interests.
- **3.12.** Broker's and Finder's Fees. No broker, finder or other Person is entitled to any commission or finder's fee in connection with this Agreement or with the transactions contemplated hereby as a result of any actions or commitments of the Seller.

- **3.13.** No Reliance on IPO. The Seller acknowledges and agrees that in entering into this Agreement, it is not relying on any express or implied representation or other statement of the Purchaser regarding the Purchaser consummating or otherwise participating in an IPO.
- **3.14.** Seller Capitalization. The capitalization of the Seller is as set forth on Exhibit B (the "Seller Interests"). The Seller Interests constitute all of the issued and outstanding limited liability company interests, Equity Securities, or similar interests of the Seller and are duly authorized and were validly issued in compliance with the Seller's Governing Documents and all applicable securities Laws. There are no, and as of immediately following the Closing after giving effect to the transactions contemplated by this Agreement, there will not be, any Contracts, options, warrants, call rights, puts, convertible securities, exchangeable securities, understandings or other rights, arrangements or understandings of any kind to issue, repurchase, redeem, sell, deliver or otherwise acquire or cause to be issued, repurchased, redeemed, sold, delivered or acquired, any limited liability company interests, Equity Securities, Debt or similar interests in the Seller. The Seller has no Subsidiaries. Except as contemplated by the Governing Documents of the Seller, there are no voting trusts, limited liability company agreements, proxies or other agreements, understandings or obligations in effect with respect to the voting, transfer or sale (including any rights of first refusal, rights of first offer or drag-along rights), issuance (including any pre-emptive or anti-dilution rights), redemption or repurchase (including any put or call or buy-sell rights), or registration (including any related lock-up or market standoff agreements) of any membership interests of Seller.

ARTICLE IV REPRESENTATIONS AND WARRANTIES REGARDING THE COMPANY

The Company and the Seller, jointly and severally, hereby represent and warrant to the Purchaser, subject to such exceptions as are specifically disclosed in the Disclosure Schedules delivered by the Seller concurrently with the execution of this Agreement (the "Disclosure Schedules"); that the statements contained in this Article IV are true, correct and complete as of the date hereof.

4.01. Organization; Power. The Company is a limited liability company duly organized and validly existing under the Laws of the State of Delaware. The Company is qualified as a foreign entity to transact business in, and is validly existing under the Laws of, the states listed on Schedule 4.01, and is not required to be qualified as a foreign entity in any other states or jurisdictions, except where such non-qualification would not reasonably be expected to have a Material Adverse Effect. The Company has all requisite power and authority to own, operate, lease and encumber its properties and carry on its business as now conducted and currently proposed to be conducted and to enter into this Agreement and each of the other Related Agreements to which it is a party and to carry out its obligations hereunder and thereunder.

4.02. <u>Capitalization</u>.

(a) The Contributed Interests constitute all of the authorized limited liability company membership interests of the Company. All of the Contributed Interests are held of record and beneficially owned by the Seller. The Contributed Interests constitute all of the issued and outstanding limited liability company interests, Equity Securities, or similar interests of the

Company and are duly authorized and were validly issued in compliance with the Company's Governing Documents and all applicable securities Laws.

- (b) Except as set forth on <u>Schedule 4.02(b)</u>, there are no, and as of immediately following the Closing after giving effect to the transactions contemplated by this Agreement, there will not be, any Contracts, options, warrants, call rights, puts, convertible securities, exchangeable securities, understandings or other rights, arrangements or understandings of any kind to issue, repurchase, redeem, sell, deliver or otherwise acquire or cause to be issued, repurchased, redeemed, sold, delivered or acquired, any limited liability company interests, Equity Securities, Debt or similar interests in the Company.
- (c) Except as contemplated by the Governing Documents of the Company, there are no voting trusts, limited liability company agreements, proxies or other agreements, understandings or obligations in effect with respect to the voting, transfer or sale (including any rights of first refusal, rights of first offer or drag-along rights), issuance (including any pre-emptive or anti-dilution rights), redemption or repurchase (including any put or call or buy-sell rights), or registration (including any related lock-up or market standoff agreements) of any membership interests of the Company.
- Schedule 4.02(d) sets forth a true and complete list of all direct or indirect (d) Subsidiaries of the Company, listing for each Subsidiary its name, its jurisdiction of organization or formation, and the current record and beneficial ownership of its Equity Securities. All of the issued and outstanding Equity Interests of each Subsidiary are validly issued, credited as fullypaid and non-assessable, have not been issued in violation of any preemptive or similar rights, and are owned by the Company free and clear of any Liens. The Equity Securities of each Subsidiary were issued in compliance with applicable Laws. There are no outstanding subscriptions, options, rights, warrants or other commitments entitling any Person to purchase or otherwise subscribe for or acquire any Equity Interests of any Subsidiary of the Company, nor is there presently outstanding any security convertible into or exchangeable for Equity Interests of any Subsidiary of the Company, nor has the Company, any of its Subsidiaries or the Seller entered into any agreement with respect to any of the foregoing. Other than the Governing Documents of each Subsidiary of the Company, there are no voting trusts, proxies or other agreements or understandings in effect with respect to the voting or transfer of any of the Equity Securities of such Subsidiary. Except as set forth on Schedule 4.02(d), neither the Company nor any of its Subsidiaries, directly or indirectly, own or have any interest in the Equity Interest in any Person.
- **4.03.** No Conflict or Violation. Except as set forth on Schedule 4.03, the execution, delivery, consummation and performance of this Agreement and each of the Related Agreements does not and shall not: (a) violate or conflict with any provision of the Governing Documents of the Company, (b) violate in any material respect any provision of Law applicable to the Company, (c) violate or result in a breach of or constitute (with or without due notice or the passage of time, or both) a default under any judicial consent, order, decree or material Contract to which the Company is a party, or by which the Company's material Assets or properties may be bound, or (d) result in the imposition of any Encumbrance or restriction on the Business (with or without due notice or the passage of time, or both).

4.04. Consents and Approvals. Except as set forth on Schedule 4.04, to the Knowledge of the Company, no consent, waiver, authorization or approval of any Governmental Entity, or of any other Person, or declaration or notice to or filing or registration with any Governmental Entity or other Person, is required in connection with the consummation of the transactions contemplated hereby.

4.05. Financial Statements; No Undisclosed Liabilities.

- (a) Attached hereto as <u>Schedule 4.05(a)</u> are copies of (i) the consolidated audited financial statements of the Company as of and for the year ended December 31, 2017 and December 31, 2016, which are comprised of the balance sheets as of December 31, 2017 and December 31, 2016, 2015, respectively, and the related statements of comprehensive income and owners' equity and cash flows for the years then ended and (ii) unaudited statements of the Company as of November 30, 2018 (such date, the "*Balance Sheet Date*"), which are comprised of the balance sheet as of the Balance Sheet Date, and the related statements of comprehensive income, owners' equity and cash flows of the Company for the ten (10) period then ended (collectively, the "*Financial Statements*").
- (b) There are no material Liabilities against, relating to or affecting the Company or the Business of a nature required to be disclosed by GAAP except for Liabilities fully disclosed in the Financial Statements and as set forth on <u>Schedule 4.05(b)</u>.

4.06. Tax Matters.

- (a) The Company has timely filed or caused to be timely filed all Tax Returns that are or were required to be filed by the Company pursuant to applicable Law. All Tax Returns filed by the Company are true, correct and complete in all material respects and were prepared in compliance with all applicable Laws. The Company has paid all Taxes owed by the Company (whether or not shown on any Tax Returns), except such Taxes, if any, which are not yet delinquent. The Company is not currently the beneficiary of any extension of time within which to file any income Tax Return. The Company has received no written claim made by any Governmental Entity in a jurisdiction where the Company does not file Tax Returns asserting that the Company is or may be subject to taxation by that jurisdiction. There are no Encumbrances for Taxes (other than for Taxes not yet due and payable) upon any of the Assets of the Company.
- (b) The Company has delivered or made available to the Purchaser copies of all Tax Returns filed by the Company and any audit materials, examination reports and statements of deficiencies with respect to Taxes of the Company, in each case, received by the Company or the Seller from a Governmental Entity, with respect to all tax years of the Company from inception. Neither the Seller nor the Company has received written notice from a Governmental Entity indicating that such Governmental Entity intends to assess any additional Taxes against the Company for any period for which Tax Returns have been filed. There are no pending audits, assessments, disputes or claims concerning any Taxes of the Company. The Company has not waived any statute of limitations in respect of Taxes or agreed to any extension of time with respect to a Tax assessment or deficiency, in each case, that are either in force or outstanding. Schedule 4.06(b) sets forth all of the jurisdictions where the Company files Tax Returns.

- (c) There is no Tax sharing agreement, Tax allocation agreement, Tax indemnity obligation or similar Contract with respect to Taxes (including any advance pricing agreement, closing agreement or other arrangement relating to Taxes) that shall require any payment by the Company after the Closing Date (other than customary commercial agreements the primary subject of which is not Taxes).
- (d) The Company has never been a member of an affiliated group filing a consolidated federal income Tax Return or any similar group for federal, state, local or foreign Tax purposes. The Company has no Liability for Taxes of any other Person under Treasury Regulations Section 1.1502-6 (or any similar provision of state, local or foreign Law), as a transferee or successor, by Contract or otherwise (other than customary commercial agreements the primary subject of which is not Taxes).
- (e) The Company will not be required to include any item of income in, or exclude any item of deduction from, taxable income for any taxable period (or portion thereof) ending after the Closing Date as a result of any (i) change in method of accounting for a taxable period ending on or prior to the Closing Date, (ii) "closing agreement" as described in Section 7121 of the Code (or any similar provision of state, local or foreign Law) executed on or prior to the Closing Date, (iii) installment sale or open transaction disposition made on or prior to the Closing Date, (iv) prepaid income received on or prior to the Closing Date, (v) method of accounting that defers recognition of income to any period ending after the Closing Date, or (vi) election under Section 108(i) of the Code.
- (f) The Company has timely withheld and paid all amounts of Taxes required by Law to have been withheld and paid by it (i) in connection with any amounts paid as compensation by the Company to its employees and independent contractors (including distributors) and (ii) to its members, creditors, holders of securities or other third parties. The Company has complied with all information reporting and backup withholding requirements of Law in all material respects.
- (g) The Company is not the subject of any private ruling from a taxing authority or Contract with a taxing authority.
- (h) The Company has not: (i) taken a reporting position on a Tax Return that, if not sustained, would give rise to a penalty for substantial understatement of federal income Tax under Section 6662 of the Code (or any similar provision of state, local or foreign Law); or (ii) entered into any transaction identified as a (x) "listed transaction," within the meaning of Treasury Regulations Section 1.6011-4(b)(2), (y) a "transaction of interest," within the meaning of Treasury Regulations Section 1.6011-4(b)(6), or (z) any transaction that is "substantially similar" (within the meaning of Treasury Regulations Section 1.6011-4(c)(4)) to a "listed transaction" or "transaction of interest."
- (i) The Company and each Subsidiary is, and at all times since its formation has been, and will be at all times through the Closing, properly classified as either a partnership or an entity disregarded as separate from its owner for U.S. federal income Tax purposes.

- **4.07.** <u>Absence of Certain Changes</u>. Since June 30, 2018 there has not occurred any event, change or circumstance that, individually or in the aggregate, has had or would reasonably be expected to have a Material Adverse Effect. Except as set forth on <u>Schedule 4.07</u>, since June 30, 2018 there has not been any:
- (a) damage or destruction affecting any portion of the material Assets or properties of the Company;
 - (b) change in the Company's accounting policies, procedures or methodologies;
- (c) sale or transfer of any tangible or intangible Asset of the Company, except in the Ordinary Course of Business;
- (d) mortgage, pledge or imposition of any Encumbrances (except for Permitted Encumbrances) on any Asset of the Company;
- (e) declaration or payment of any dividend or distribution in respect of any Equity Securities of the Company or, directly or indirectly, any purchase, redemption, issuance, or other acquisition or disposition by the Company of any of their respective Equity Securities;
- (f) increase in the salary, benefits or other compensation payable to any of the employees or consultants or officers or directors of the Company, or commitment to pay any bonus or other additional salary, benefits or compensation to any of the employees or consultants or officers or directors of the Company, or any entry into, grant, adoption, amendment or termination of any Employee Plan in any manner, except as otherwise required by Law;
- (g) incurrence of any capital expenditure, obligation or other liability in connection therewith by the Company other than in the Ordinary Course of Business not in excess of \$25,000;
- (h) acquisition by the Company of a Person (including by merger, consolidation or stock purchase), or any acquisition of a substantial portion of the Assets of any business of any other Person;
- (i) discharge or satisfaction by the Company of any material Encumbrance or material liability, other than Liabilities discharged or satisfied in the Ordinary Course of Business;
 - (j) amendment to the Governing Documents of the Company;
- (k) incurrence, assumption or prepayment of any Debt by the Company, including long term Debt or the issuance of any debt securities;
- (l) assumption, guarantee, endorsement or otherwise becoming liable or responsible, whether directly, contingently or otherwise, by the Company for the Debt of any other Person;
- (m) making of any loans, advances or capital contributions to or investments in any other Person by the Company;

- (n) entry into, amendment or termination of, or waiver of any rights under, any Material Contract:
- (o) initiation, settlement or compromise by or against the Company of any pending or threatened Proceeding;
- (p) making, changing or rescinding by the Company of any election relating to Taxes, settlement or compromise by the Company of any claim, action, suit, litigation, Proceeding, arbitration, investigation, or audit controversy relating to Taxes, consent by the Company to any extension or waiver of the limitation period applicable to any claim or assessment in respect of Taxes, filing by the Company of any Tax Return, amendment of any Tax Return by the Company or making by the Company of any change to any of its respective methods of accounting in respect of Taxes;
- (q) excess sales of inventory or return privileges granted to any customer of the Company with respect thereto (whether explicitly or through favorable concessions granted to the customer); or
- (r) agreement or commitment entered into by the Company to do any act described in clauses (a) through (q) above.
- **4.08.** Owned Real Property. The Company does not own, nor has ever owned, any real property or any interest therein (other than its lease interest in any Leased Real Property as set forth in Schedule 4.09(a)).

4.09. Leased Real Property.

- Schedule 4.09(a) identifies all real property leased or subleased or used by the Company as of the date hereof, including the landlord's name (the "Leased Real Property"). All Leased Real Property is leased to the Company, pursuant to written leases, true, correct and complete copies of which have been previously delivered to the Purchaser (collectively the "Real Property Leases"). The Company has a valid leasehold interest in the Leased Real Property, free and clear of all Encumbrances. Other than as set forth in Schedule 4.09(a), the Company has not subleased any Leased Real Property and the Leased Real Property is not otherwise subject to any third-party licenses, concessions, leases or tenancies of any kind. The Real Property Leases are in full force and effect and there are no other amendments, agreements or understandings relating to the Real Property Leases. All rent, additional rent and other charges due under the Real Property Leases were paid in full through the end of the month applicable to the Closing Date. There are no material defaults on the part of the Company or the landlord under the Real Property Leases. The Company has performed all of its obligations to be performed under the Real Property Leases. To the Knowledge of the Company, there are no claims by any landlord against the Company under the Real Property Leases. There are no rent concessions, abatements, or contributions owed to the Company under any Real Property Leases.
- (b) Except as set forth on <u>Schedule 4.09(b)</u>, the Company has not received written notice that the use or occupancy of the Leased Real Property violates in any material respect any covenants, conditions or restrictions that encumber such property, or that any such

property is subject to any restriction for which any material permits necessary to the current use thereof have not been obtained.

- (c) There are no pending or threatened condemnation Proceedings with respect to any portion of the Leased Real Property. There are no actual or threatened or imminent changes in the present zoning of any Leased Real Property or any part thereof or any restrictions, limitations or regulations issued, or proposed or under consideration by any Governmental Entity having or asserting jurisdiction over the Leased Real Property.
- **4.10.** Assets. The Company has good and marketable title (or valid, binding and enforceable leasehold interest with respect to leased Assets) to all properties and Assets used in its business, free and clear of all Encumbrances, except for Permitted Encumbrances listed on Schedule 4.10. Except as set forth in Schedule 4.10, such tangible Assets are in good operating condition and in a state of good maintenance and repair (ordinary wear and tear excepted) and are suitable for the purposes used. No such tangible Assets are (a) in need of repair or replacement other than as part of routine maintenance consistent with historical practices, or (b) owned, used or shared by the Seller. Such material tangible Assets and properties constitute all of the Assets used in or held for use in the business of the Company and are sufficient for the Purchaser to conduct the business of the Company from and after the Closing Date without interruption and in the Ordinary Course of Business.
- 4.11. Accounts Receivable. The accounts receivable reflected on the Financial Statements and the accounts receivable arising after the Balance Sheet Date: (a) arose from bona fide sales transactions in the Ordinary Course of Business and are payable in the Ordinary Course of Business on terms consistent with the Company's past practices; (b) are legal, valid and binding obligations of the respective debtors enforceable in accordance with their terms; (c) to the Knowledge of the Company, are not subject to any valid set-off or counterclaim by the debtor; (d) do not represent obligations for goods sold on consignment, on approval or on a sale-or-return basis or subject to any other repurchase or return arrangement; (e) to the Knowledge of the Company, are collectible in full, but in no event later than ninety (90) days of the applicable invoice date thereof, in the Ordinary Course of Business in the aggregate recorded amounts thereof; (f) are not owed by any Affiliate of the Company; and (g) are not the subject of any Proceeding. Except as set forth on Schedule 4.11, the Company has not received any notice from any account debtor regarding any dispute over any of the accounts receivable. None of the accounts receivable constitutes duplicate billings of other accounts receivable. There are no security arrangements or collateral securing the repayment or other satisfaction of the accounts receivable.

4.12. Intellectual Property.

(a) Schedule 4.12(a) sets forth a list of all utility and design patents and patent applications; registered and common law trademarks and service marks; trade names; domain names; registered copyrights; and software (other than commercial-off-the-shelf software licensed on a "click-through" or similar basis for the internal use of the Company) owned by or licensed to the Company pursuant to any Contract and used by the Company in the operation of the Business as presently conducted. The Company owns, free and clear of Encumbrances, or has the right to use pursuant to valid and effective Contracts, all such Intellectual Property, and all software licenses, trade secrets, technical knowledge, know-how and other confidential proprietary

information used to conduct the Business (collectively referred to as "Company Intellectual Property"). None of the Company Intellectual Property is owned by or licensed to the Company by any Affiliate, officer, director, contractor, or Employee of the Company. The Company Intellectual Property constitutes all Intellectual Property necessary for the continued conduct of the Business of the Company in substantially the same manner as conducted before the date of this Agreement. For each item of Intellectual Property licensed from third parties, Schedule 4.12(a) lists the Contract under which the Company has acquired rights in such Intellectual Property (each, an "Intellectual Property License"), including the date, title and parties for each such Intellectual Property License. Each Intellectual Property License is valid and binding on the Company and the applicable licensor in accordance with its terms and is in full force and effect. Neither the Company nor, to the Knowledge of the Company, any other party thereto is in breach of or default under, or has provided or received any notice of breach or default of or any intention to terminate, any Intellectual Property License. No claims are pending or, to the Knowledge of the Company, threatened, and the Company has received no communication alleging that the Company violated any rights relating to Intellectual Property of any third party. To the Knowledge of the Company, no third party is misappropriating, infringing, diluting, or violating any Company rights in Intellectual Property.

- (b) Except as set forth on <u>Schedule 4.12(b)</u>, to the Knowledge of the Company, neither the Company, nor any Company Product, nor the marketing, distribution, sale or use of any Company Product for its intended purpose, infringes, violates, dilutes or misappropriates any Intellectual Property rights of another Person.
- (c) Except as set forth in <u>Schedule 4.12(c)</u>, the Company has not made any claim of a violation, infringement, misuse or misappropriation by any Person (including any Employee, former employee or independent contractor of the Company or any of its Subsidiaries) of its rights to, or in connection with, any Company Intellectual Property, and to the Knowledge of the Company, no basis for such a claim exists. Except as set forth in <u>Schedule 4.12(c)</u>, neither the Company nor any of its Subsidiaries has entered into any agreement to indemnify any other Person against any charge of infringement of any Intellectual Property, other than indemnification provisions contained in Contracts entered into in the Ordinary Course of Business.
- (d) The Company has taken all commercially reasonable measures to maintain and protect the proprietary nature of the Company Intellectual Property, including the signing by all Persons hired by the Company and all of its Subsidiaries of nondisclosure and non-competition agreements, and the signing of valid and binding non-disclosure agreement by all third parties with responsibility for the conception, reduction to practice, authoring or other creation or development of, or having access to, or to whom a disclosure has been made of, know-how, trade secret information or other Company Intellectual Property. Except as set forth in Schedule 4.12(d), the Company has secured valid and binding written assignments from all consultants, contractors and employees and all other Persons who contributed to the conception, reduction to practice, authorship, creation or development of any Company Intellectual Property by or on behalf of the Company or any of its Subsidiaries of all rights to such contributions that the Company or its Subsidiary, as applicable, does not already own by operation of Law.

- (e) Except as set forth on <u>Schedule 4.12(e)</u>, no Person (other than the Company) has contributed to or participated in the conception and development of Intellectual Property that is necessary to or used by the Company in the operation of the Business.
- (f) All Persons who have access to Confidential Information of the Company are as set forth on Schedule 4.12(f).
- (g) Neither the Company nor any of its Subsidiaries has granted to any Person an exclusive license or equivalent right with respect to any Company Intellectual Property, or assigned or conveyed to any Person any ownership interest (including joint ownership rights) therein, and no third party owns or holds any such right, license or interest.
- (h) Other than as listed on <u>Schedule 4.12(h)</u> or as integrated into third-party software licensed to the Company for its internal use, neither the Company nor any of its Subsidiaries use, and no Company Products use or require the use of, any "open source" code (as defined by the Open Source Initiative) or "Free" code (as defined by the Free Software Foundation).
- (i) Except as set forth on <u>Schedule 4.12(i)</u>, neither the Company nor any of its Subsidiaries has disclosed or delivered to any escrow agent or any other Person any of the source code relating to any software covered by any Company Intellectual Property, and no other Person has the right, contingent or otherwise, to obtain access to or use any such source code. No event has occurred, and no circumstance or condition exists, that (with or without notice or lapse of time or both) will, or could reasonably be expected to, result in the delivery, license, or disclosure of any such source code to any Person who is not, as of the date of this Agreement, a current Employee.
- (j) The execution, delivery and performance of this Agreement and the consummation of the transactions contemplated hereby will not (i) materially alter or impair any rights of the Company in any Intellectual Property used in the Business or owned by the Company, or (ii) result in the Company being obligated to license any Intellectual Property used in the Business or owned by the Company to any other Person, or to pay any royalties or other material amounts, to accelerate the payment of any royalties or any other material amounts, or to offer any discounts to any third party in excess of those payable by, or required to be offered by, the Company in the absence of this Agreement.
- (k) The information technology systems owned, leased or licensed by the Company, including the software, firmware, hardware (whether general or special purpose), networks and interfaces (collectively, the "Company Systems") are sufficient for the current needs of the Company for its operation of the Business, including as to peak volume capacity and processing ability. In the twelve month period prior to the date of this Agreement, there have been no material failures, breakdowns, or continued substandard performance of any Company Systems which have caused the substantial disruption or interruption in the use of the Company Systems or the operation of the Business of the Company.
- (l) With respect to sensitive personally identifiable information, the Company has taken all commercially reasonable steps (including implementing and monitoring compliance with adequate measures with respect to technical and physical security) to ensure that the

information is protected against loss and against unauthorized access, use, modification, disclosure or other misuse. To the Knowledge of the Company, there has been no unauthorized access to or other misuse of such information.

4.13. Employee Benefit Plans.

- (a) Schedule 4.13(a) contains a list of all Employee Plans. Each Employee Plan is being administered in accordance with its terms and with all applicable Laws, and contributions required to be made under the terms of any of the Employee Plans, if any, as of the date of this Agreement have been timely made. With respect to each Employee Plan, the Company has remained in material compliance with all Tax, annual reporting and other governmental filing requirements under applicable Law, and such Taxes, reports and other filings have, in all material respects, been timely filed with the appropriate Governmental Entity and all notices and disclosures have been timely provided to participants. Each Employee Plan which is intended to qualify under Section 401(a) of the Code has received a favorable determination letter from the IRS as to its qualified status or may rely upon a prototype opinion letter, and that the trust established in connection with such Employee Plan which is intended to be exempt from federal income taxation under Section 501(a) of the Code is so exempt, and no fact or event has occurred that could reasonably be expected to adversely affect the qualified status of any such Employee Plan or the exempt status of any such trust. Neither the Company nor any Subsidiary has any express or implied commitment, whether legally enforceable or not, to (i) create, incur liability with respect to, or cause to exist any Employee Plan (or any plan, program or arrangement which would be an Employee Plan if in effect on the date hereof), (ii) to enter into any Contract or agreement to provide compensation or benefits to any individual, or (iii) to modify, change or terminate any Employee Plan, other than with respect to a modification, change or termination required by ERISA or by the Code.
- With respect to each Employee Plan, the Company has furnished to the Purchaser, as applicable: (i) a true and complete copy of each Employee Plan and underlying trust (or, in the case of an unwritten arrangement, a written description of its terms and conditions); (ii) copies of the most recent summary plan description and all summaries of material modifications; (iii) copies of the three (3) most recently filed Form 5500 annual reports and accompanying schedules, if any; (iv) a copy of the most recently received IRS determination letter or opinion letter; (v) copies of the non-discrimination testing results, audited financial statements, actuarial reports, and attorney's responses to an auditor's request for information, if any, for the three (3) most recent plan years; (vi) all material correspondence to or from any Governmental Entity in the past three (3) years relating to any Employee Plan; (vii) all material communications relating to any established or proposed Employee Plan that relates to any material amendments, terminations, increases or decreases in benefits, acceleration of payments or vesting schedules or other events which would result in any liability to the Company or its Subsidiaries; and (viii) all prospectuses prepared in connection with each Employee Plan. The Company has not made any material express or implied proposal, assurance or commitment, to establish, modify, change or terminate any Employee Plan (other than with respect to a modification, change or termination required by ERISA or the Code), or to any Employee or other service provider of the Company regarding any improvement to terms of employment or regarding the increase or improvement in the rate or quantum of remuneration, benefits or other compensation.

- (c) (i) No event has occurred and no condition exists that would subject the Company or any ERISA Affiliate, to any Tax, fine, lien, penalty or other liability imposed by ERISA, the Code or other applicable Laws, rules and regulations; (ii) for each Employee Plan with respect to which a Form 5500 has been filed, no material change has occurred with respect to the matters covered by the most recent Form since the date thereof; (iii) no "reportable event" (as such term is defined in ERISA Section 4043), "prohibited transaction" (as such term is defined in ERISA Section 302 and Code Section 412 (whether or not waived)) has occurred with respect to any Employee Plan; (iv) all awards, grants or bonuses made pursuant to any Employee Plan have been, or will be, fully deductible to the Company or its Subsidiaries notwithstanding the provisions of Section 162(m) of the Internal Revenue Code and the regulations promulgated thereunder; and (v) except to the extent limited by applicable Law, each Employee Plan may be amended, terminated or otherwise discontinued after the Closing Date in accordance with its terms without liability to the Company or any Subsidiary (other than ordinary administration expenses).
- (d) Full payment has been made of all amounts (other than current outstanding routine claims for benefits) that the Company and any Subsidiary is required to contribute or pay under the terms of any Employee Plan, if any, and all contributions to any Employee Plan that are required or recommended with respect to any period of time prior to the Closing have been made or such amounts have been accrued in accordance with GAAP. There are no funded benefit obligations for which contributions have not been made or properly accrued and there are no unfunded benefit obligations that have not been accounted for by reserves, or otherwise properly footnoted in accordance with GAAP on the Financial Statements.
- (e) No Employee Plan is a Multiemployer plan or a pension plan within the meaning of Section 3(2) of ERISA that is subject to Title IV of ERISA or similar minimum funding requirements under applicable foreign Law (each such arrangement being a "*Pension Plan*"), and neither the Company nor any ERISA Affiliate has ever sponsored or contributed to or been required to contribute to a Multiemployer Plan or Pension Plan. No material liability under Title IV of ERISA or similar applicable foreign Law has been incurred by the Company or any ERISA Affiliate that has not been satisfied in full, and no condition exists that presents a risk to the Company or any ERISA Affiliate of incurring or being subject (whether primarily, jointly or secondarily) to a liability thereunder or to any lien arising under ERISA.
- (f) With respect to any Employee Plan, (i) no actions, suits or claims (other than routine claims for benefits in the ordinary course) are pending or threatened and (ii) no facts or circumstances exist that could give rise to any such actions, suits or claims. There are no audits, inquiries or Proceedings pending or to the Knowledge of the Company, threatened by the Internal Revenue Service, the Department of Labor, or any similar Governmental Entity with respect to any Employee Plan.
- (g) Except as set forth on Schedule 4.13(g), the Company has not granted any loans or advances in excess of \$1,000, or provided any guarantees or financial assistance in excess of \$1,000, to any of its officers or directors (past or present), which are currently outstanding. For the avoidance of doubt, this representation does not apply to any loans or advances (if any) which are (or were) made in connection with any Employee Plan which is intended to be qualified under Section 401(a) of the Code.

- (h) Except as set out on <u>Schedule 4.13(h)</u>, there is no term of employment for any Employee which provides that a change of control (i) shall be a deemed a breach of his or her service or employment contract, or (ii) would entitle the Employee concerned to the vesting or acceleration of any payment or benefit whatsoever or entitle such Employee to be treated as redundant or otherwise dismissed or released from any such obligation.
- (i) Except as required by the Consolidated Omnibus Budget Reconciliation Act of 1985, as amended, and the regulations (including proposed regulations) thereunder and any similar state Law, no Employee Plan provides any post-termination or retiree medical or retiree welfare benefits to any Person.
- (j) All "nonqualified deferred compensation plans" (as defined in Section 409A of the Code and the Treasury Regulations promulgated thereunder) of the Company have been operated in material compliance with Section 409A of the Code and all applicable guidance (including the Treasury Regulations) promulgated thereunder. The Company is not a party to, or otherwise obligated under, any Employee Plan that provides for the gross-up of the Tax imposed by Section 409A(a)(1)(B) of the Code.
- (k) No payment or benefit provided pursuant to any Employee Plan, including the grant, vesting or exercise of any equity-based award, will or may provide for the deferral of compensation subject to Section 409A of the Code, whether pursuant to the execution and delivery of this Agreement or the consummation of the transactions (either alone or upon the occurrence of any additional or subsequent events) or otherwise. The Company is not a party to, or otherwise obligated under, any Employee Plan that provides for the gross-up of the Tax imposed by Section 409A(a)(1)(B) of the Code. The execution and delivery of this Agreement by the Company and the consummation of the transactions contemplated by this Agreement and the Related Agreements will not (either alone or upon the occurrence of any additional or subsequent events) constitute an event under any Employee Plan or Contract that will or may result in any payment of deferred compensation which will not be in compliance with Section 409A of the Code.
- (l) The Company has no employees who are providing services at a location which are subject to the Laws of any jurisdiction outside of the United States.
- (m) The execution of this Agreement and the consummation of the transactions contemplated hereby will not (either alone or upon the occurrence of any other event, such as individual's termination of employment) constitute an event under any Employee Plan that will or may result in any payment (whether of severance pay or otherwise), acceleration, forgiveness of indebtedness, vesting, distribution, increase in benefits or obligation to fund benefits with respect to any employee of the Company or any member of the Company's "Controlled Group". There is no Employee Plan, contract, plan or arrangement to which the Company (or any member of its Controlled Group) is a party or by which it is bound, that requires the Company (or any member of the its Controlled Group) to compensate any employee, former employee or any person providing services to the Company (or any member of its Controlled Group) for excise taxes paid pursuant to Code Section 4999.

4.14. Personnel; Labor Relations.

- (a) <u>Schedule 4.14</u> lists the name of each employee of the Company ("*Employee*") and, with respect to each such Employee, his or her: (i) work location; (ii) position; (iii) hire date; (iv) classification (i.e., exempt or non-exempt); (v) rate of compensation (base salary or hourly rate of pay); (vi) bonus (or commission) opportunity; and (vii) visa or green card status.
- (b) The Company is not a party to or bound by any collective bargaining or labor contract, voluntary recognition agreement or other binding commitment to any labor union, trade union, works council or employee organization in respect of any of its employees. There are not currently, and in the five (5) years preceding the date hereof there have not been, nor are there now threatened, any: (i) strikes, work stoppages, slowdowns, lockouts or arbitrations; or (ii) employee or union grievances, claims, charges, unfair labor practice charges, grievances or complaints or other labor disputes with respect to the Company. During the five (5) years preceding the date hereof, none of the employees of the Company is or has been represented by any labor union or other employee collective bargaining organization, was a party to, or bound by, any labor or other collective bargaining agreement in connection with such employment or has been subject to or involved in, or threatened, any union elections, petitions or other organizational or recruiting activities, nor are any such labor organizing activities now pending or threatened against the Company.
- The Company is in compliance in all material respects with all applicable Laws relating to employment or termination of employment, including those related to wages, hours, compensation, terms and conditions of employment, workplace health and safety, discrimination or harassment, retaliation, human rights, pay equity, notice of termination, classification of workers (i.e., as employees versus independent contractors, or as exempt versus non-exempt employees), immigration, collective bargaining and the payment and withholding of Taxes and other sums as required by the appropriate Governmental Entity. The Company has paid in full to all employees, or adequately accrued for in accordance with GAAP consistently, applied all wages, salaries, commissions, bonuses, benefits and other compensation due to or on behalf of such employees through the pay period preceding the date hereof. Other than as set forth in Schedule 4.14(c), there is no claim with respect to employment or termination of employment, or payment of wages, salary or overtime pay that has been asserted or is now pending or threatened before any Governmental Entity, and no audit or investigation by any Governmental Entity is currently pending or threatened. The Company has no liability, whether direct or indirect, absolute or contingent, including any obligations under any Employee Plans, with respect to any misclassification of a Person performing services as an independent contractor or consultant rather than as an employee. To the Knowledge of the Company, no group of Employees and no key Employee, manager or executive has any current plans to terminate employment in connection with the Closing.

4.15. Environmental Compliance.

(a) To the Knowledge of the Company, the Company is in compliance, in all respects, with all applicable Environmental Laws. The Company does not possess any Environmental Permits for the operation of the Business. There is no Environmental Claim pending or threatened against the Company.

- (b) The Company has not treated, stored, disposed of, arranged for or permitted the disposal of, transported, handled, released or exposed any Person to any Hazardous Substances or owned, used or operated any property or facility (and no such property or facility is contaminated by any Hazardous Substance), so as to give rise to any Environmental Claim.
- (c) Without limiting the generality of the foregoing, the Company has no outstanding legal or contractual obligation under any applicable Environmental Law, or any unresolved enforcement action, Liability or other Proceeding pursuant to any Environmental Law, including any outstanding investigation, cleanup, removal, response activity, remediation, or corrective action obligation under any applicable Environmental Law or any outstanding indemnification obligation owed to any third party under any applicable Environmental Law relating to the Leased Real Property, any formerly owned real property, any formerly owned, used or operated property, or any offsite disposal location.
- (d) Neither this Agreement nor the consummation of the transactions contemplated hereby will result in any obligations for site investigation or cleanup, or notification to or consent of any Governmental Entity or third parties, pursuant to any of the so-called "transaction-triggered" or "responsible property transfer" Environmental Laws.
- (e) To the Knowledge of the Company, there are no current or any abandoned or former underground storage tanks (USTs) located at any real property owned, used or operated by the Company and any such USTs that do exist are in compliance with applicable Environmental Laws.
- 4.16. <u>Licenses and Permits</u>. <u>Schedule 4.16</u> sets forth a list of all Licenses and Permits currently held by the Company, and no suspension or cancellation of any of the Licenses and Permits is (a) pending, (b) threatened in writing or (c) to the Knowledge of the Company, threatened orally by a Governmental Entity, or will result due to the consummation of the transaction contemplated by this Agreement. The Company has obtained and maintained, and currently maintains, in full force and effect all Licenses and Permits required to operate the Business as presently conducted in the Ordinary Course of Business and as currently proposed to be conducted, each of which is set forth on <u>Schedule 4.16</u>. The consummation of the transactions contemplated hereby shall not give any Governmental Entity the right to terminate any of the Licenses and Permits or the conduct of the Business or require any amendments, registration, or renewal of any such Licenses and Permits. The Company is in compliance in all material respects with all terms, conditions and requirements of all Licenses and Permits and no Proceeding is pending or threatened relating to the revocation or limitation of any of such Licenses and Permits.
- **4.17.** <u>Insurance</u>. <u>Schedule 4.17</u> sets forth a list of all policies of title, liability, fire, casualty, business interruption, workers' compensation and all other forms of insurance (including self-insurance arrangements) (collectively, the "*Policies*" and individually, each a "*Policy*") insuring the properties, Assets or other operations of the Company. A true, correct and complete copy of each Policy has been made available to the Purchaser. Each of the Policies is in full force and effect. The Company is not in default under any material provisions of any Policy, and the Company has not received notice of cancellation of any Policy. There is no claim by the Company pending under any Policy as to which coverage has been denied or disputed by the underwriters of any Policy. The Company has not received any notice from or on behalf of any

insurance carrier issuing any Policy that insurance rates therefor shall hereafter be materially increased or that there shall hereafter be a cancellation or an increase in a deductible (or a material increase in premiums in order to maintain an existing deductible) or non-renewal of any Policy.

- **4.18.** Payment Card Standards. Except as set forth on Schedule 4.18, the Company and its Subsidiaries have collected, stored, maintained, used, shared and processed Personal Information in accordance with all Applicable Privacy and Data Security Laws and have taken commercially reasonable steps to protect against any anticipated or actual threats or hazards to the security or integrity of Personal Information, and from the loss of Personal Information. To the Knowledge of the Company, the Company's and its Subsidiaries' practices, policies and procedures with regard to payment instrument information are in full compliance with all rules, regulations, standards and guidelines adopted or required (a) by all payment card brands that are accepted as a form of payment by, or whose instrument information is otherwise handled by, the Company, and (b) by the Payment Card Industry Security Standards Council, in either case relating to privacy, data security or the safeguarding, disclosure or handling of payment instrument information, including but not limited to (1) the Payment Card Industry Data Security Standards, (2) the Payment Card Industry's Payment Application Data Security Standard, (3) the Payment Card Industry's PIN Transaction Security requirements, (4) Visa's Cardholder Information Security Program and Payment Application Best Practices, (5) American Express's Data Security Operating Policy, (6) MasterCard's Site Data Protection Program and POS Terminal Security program, and (7) the analogous security programs implemented by other card brands, in each case referenced in this sentence as they may be amended from time to time (collectively referred to herein as the "PCI Requirements"). Other than as set forth on Schedule 4.18, to the Knowledge of the Company, neither the Company nor its Subsidiaries have suffered a breach of Personal Information that was required to be reported to a data subject or a data owner or licensee pursuant to any Applicable Privacy and Data Security Laws or any other Applicable Laws. The Company and its Subsidiaries have written agreements with each third party service provider or partner having access to Personal Information requiring compliance with Applicable Privacy and Data Security Laws, including the PCI Requirements to the extent applicable. The Company and its Subsidiaries maintain records of their customers' communications preferences, such as opt-ins and opt-outs for various forms of direct marketing, behavioral advertising, and customer tracking, sufficient for the Company and its Subsidiaries to honor such preferences and comply with all Applicable Privacy and Data Security Laws. The Company and its Subsidiaries are and have always been in compliance with their published privacy policies.
- **4.19.** <u>Contracts and Commitments</u>. <u>Schedule 4.19</u> contains a list of all of the following Contracts (collectively, the "*Material Contracts*"):
- (a) each written employment agreement and consulting agreement currently in effect, along with all bonus, profit-sharing, percentage compensation, deferred compensation, pension, welfare, retirement, stock purchase or stock option plans or other Contracts with or relating to the Personnel of the Company; further, the Company is not party to any employment agreements other than at-will employment agreements;
- (b) each Contract currently in effect with a customer representing annual revenues in excess of \$50,000;

- (c) Contracts currently in effect evidencing any Debt of the Company, including Contracts for the repayment or borrowing of money by the Company, or for a line of credit (including credit card agreements), as well as guarantees of, indemnification for or agreements to acquire any obligations of others, and all security or pledge agreements related thereto;
- (d) Contracts currently in effect relating to any joint venture, partnership, strategic alliance or sharing of profits or losses with any Person to which the Company is a party or by which it or any of its Assets is bound;
- (e) Contracts currently in effect that evidence or relate to any obligations of the Company with respect to the issuance, sale, repurchase or redemption Equity Securities;
- (f) Contracts that relate to any Proceeding involving the Company at any time during the last four years;
- (g) Contracts relating to the acquisition or disposition of any Equity Securities, business or product line of any other Person pursuant to which any economic obligations (whether or not contingent) remain outstanding;
- (h) Contracts currently in effect that contain covenants limiting the freedom of the Company to compete in any business in any material respect or in any geographic area;
- (i) Contracts currently in effect with respect to any Intellectual Property owned or licensed by the Company other than off-the-shelf software licenses;
- (j) Contracts currently in effect pursuant to which the Company has granted any exclusive agency, marketing, sales representative relationship or distribution right to any third party;
- (k) Contracts currently in effect providing for capital expenditures by the Company in excess of \$25,000;
- (l) Contracts currently in effect that require the Company to make other payments equal to more than \$25,000 in any calendar year;
- (m) Contracts currently in effect to which any manufacturers of the Company are party; and
 - (n) Contracts currently in effect not made in the Ordinary Course of Business.

The Company has made available to the Purchaser true, correct and complete copies of all Material Contracts. All of the Material Contracts are in full force and effect. Neither the Company, nor any other party thereto, has breached any material provision of, or is in material default under the terms of, nor does any condition exist which (with or without due notice or the passage of time, or both), would cause the Company or any other party to be in default under any of the Material Contracts. Except as set forth on Schedule 4.19, the consummation of the transactions contemplated by this Agreement shall not afford any other party the right to terminate any such

Material Contract or require notice to or consent of any Person party to a Material Contract, or result in any increase or acceleration of any obligation under any Material Contract or the payment by the Company of any amount under any Material Contract.

4.20. <u>Customers and Suppliers</u>.

- (a) Schedule 4.20(a) sets forth a list of the names of the Company's top ten (10) customers for the twelve-month period ended December 31, 2017 and the nine-month period ended September 30, 2018 based on total revenues for such period. Since December 31, 2017, no customer set forth on Schedule 4.20(a) has terminated or adversely modified its relationship with the Company.
- (b) Schedule 4.20(b) sets forth a list of the names of the Company's top ten (10) suppliers for the twelve-month period ended December 31, 2017 and the nine-month period ended September 30, 2018 based on the dollar amount of expenditures by the Company for such period. Since December 31, 2017, no supplier set forth on Schedule 4.20(b) has terminated or adversely modified its relationship with the Company.
- (c) Since December 31, 2017, there has been no written communication from any customer set forth on Schedule 4.20(a) or any supplier set forth on Schedule 4.20(b) that would lead the Company reasonably to believe that such customer or supplier, as applicable, is planning to terminate or materially reduce or modify the terms of its business relationship with the Company.

4.21. Compliance with Law.

- (a) The Company has, since January 1, 2016, complied in all material respects with all Laws applicable to the Business, including but not limited to, as applicable, the Controlled Substances Act and the Laws and regulations of each U.S. state in which it conducts business concerning cannabis.
- (b) The Company is in compliance in all material respects with all applicable Laws. The Company is not in default or violation with respect to any order, writ, judgment, award, injunction or decree of any Governmental Entity or arbitrator applicable to it, or any of its Assets. The Company has not received, at any time during the prior four (4) years from the date of this Agreement, any written notice from any Governmental Entity regarding any actual, alleged, or potential violation of, or failure to comply with, any term or requirement of any Law applicable to the Company.
- (c) Neither the Company nor any of its managers, directors, officers, equity holders, agents and employees has: (i) used any organizational funds of the Company for unlawful contributions, gifts, entertainment or other unlawful expenses related to political activity, (ii) made any unlawful payments to foreign or domestic government officials or employees or to foreign or domestic political parties or campaigns from corporate funds or violated any provision of the Foreign Corrupt Practices Act of 1977, as amended, or (iii) made or received any other payment prohibited under any applicable Law. Neither the Company, nor any of its managers, directors, officers, equity holders, nor, to the Knowledge of the Company, any of its or their respective agents or employees, is or has been the subject of any investigation, inquiry or enforcement Proceeding by any Governmental Entity regarding any offense or alleged offense under anti-bribery, anti-

corruption or anti-fraud legislation in any jurisdiction and no such investigation, inquiry or Proceedings has been threatened.

- **4.22.** <u>Litigation</u>. Except as set forth on <u>Schedule 4.22</u>, (a) there are no Proceedings pending or, to the Knowledge of the Company, threatened by or against the Company or any of its Assets or Personnel (with respect to Personnel, in such individual's capacity as such), (b) there are no unsatisfied judgments of any kind against or in favor of the Company or any of its Assets or Personnel (with respect to Personnel, in such individual's capacity as such), and (c) neither the Company nor any of its Personnel (with respect to Personnel, in such individual's capacity as such) is subject to any outstanding judgment, order, or decree of any court or Governmental Entity. The Company is not currently planning to initiate any Proceeding. Except as set forth on <u>Schedule 4.22</u>, to the Knowledge of the Company, there are no currently existing events, facts or circumstances which could reasonably be expected to form the basis for any Proceeding or order, or decree of any court or Governmental Entity by or against the Company, or any of its Personnel (with respect to Personnel, in such individual's capacity as such).
- 4.23. <u>Title to and Sufficiency of Assets and Related Matters</u>. The Company has good and marketable title to all of the respective Assets owned by it and reflected on the Financial Statements, free and clear of all Encumbrances (other than Permitted Encumbrances), except as disposed of since June 30, 2018 in the Ordinary Course of Business. The equipment currently used in the Business is in reasonable working order. The Assets and properties owned and leased by the Company comprise all the Assets and properties that are necessary or advisable for the operation of the Business as presently conducted and as presently contemplated to be conducted without restriction, interruption or limitation, other than any restriction or limitation under any applicable Law.
- **4.24.** Broker's and Finder's Fees. No broker, finder or other Person is entitled to any commission or finder's fee in connection with this Agreement or with the transactions contemplated by this Agreement as a result of any actions or commitments of the Company.
- **4.25.** Affiliate Transactions. Except as disclosed on Schedule 4.14 or Schedule 4.25, the Company is not presently a party to any Contract with any owner, equity holder, manager, director, officer or Employee of the Company (or any relative or other Affiliate of such Persons), nor does any of the foregoing have any interest in any of the properties or Assets owned or used by the Company in connection with the operation of the Business. The Company does not provide or cause to be provided any Assets, services, or facilities to any any manager, director, officer, Employee or Affiliate (other than the Company) of the Seller.
- **4.26.** <u>Inventory</u>. All items of inventory reflected on the Financial Statements or acquired after December 31, 2017 and prior to the Closing Date consist of a quality and quantity usable and saleable in the Ordinary Course of Business except for obsolete items and work-in-process goods, all of which have been written off or written down to current fair market value on the Financial Statements or on the accounting records of the Company as of June 30, 2018, as the case may be.
- **4.27. Product Matters**. With respect to the Company Products, the Company does not have any liability, whether based on strict liability, gross negligence, breach of Contract

or otherwise, with respect to any product, component or other item designed, manufactured, distributed, assembled, produced, leased or sold by the Company to others, other than standard warranty obligations (to replace, repair, or refund) made by the Company in the Ordinary Course of Business consistent with past practice to the Purchasers of its products. Since January 1, 2016, the Company has not received written notice as to any claim or allegation of any material defect or material failure of any Company Product, of personal injury, death, or property or economic damages, any claim for punitive or exemplary damages, any claim for contribution or indemnification, or any claim for injunctive relief in connection with any Company Product sold or distributed by, or in connection with any service provided by, or based on any error or omission or negligent act in the performance of services by, the Company, and there is no basis for any such claim and no such claim is threatened. Schedule 4.27 completely and correctly describes all such claims since January 1, 2016, together in each case with the date such claim was made, the amount claimed, the disposition or status of such claim (including settlement or judgment amount), and the amount of attorney's fees incurred in connection with such claim. The Company has not had a recall of any Company Products.

- **4.28.** Bank Accounts. Schedule 4.28 sets forth a list of all of the bank accounts, investment accounts, safe deposit boxes, lock boxes and safes held by, or in the name of, the Company, and the names of all managers, directors, officers, employees or other individuals who have access thereto or are authorized to make withdrawals therefrom or dispositions thereof.
- **4.29.** Plans and Designs. True, correct and complete copies of the plans, designs, test reports, other reports, specifications, description and manuals relating to the products and services sold, provided or otherwise distributed by the Company (such products and services, "Company Products" and such plans, designs, test reports, other reports, specifications, descriptions and manuals, collectively the "Product Plans") have been provided to the Purchaser.

4.30. Privacy.

- (a) <u>Schedule 4.30</u> sets forth a list of all Personal Information held by the Company. The Company undertakes commercially reasonable efforts to adequately secure all Personal Information held by the Company.
- (b) The Company has not received any notice of any claims, investigations, or alleged violations of Law with respect to Personal Information possessed by or otherwise subject to the control of the Company, and except as set forth on <u>Schedule 4.30</u>, there are no facts or circumstances which could form the basis for any such violation.
- (c) To the Knowledge of the Company, there have been no data breaches involving any Personal Information of any of the Company's customers, suppliers or employees.
- **4.31. Full Disclosure**. No representation, warranty, covenant or agreement made by the Company or the Seller in this Agreement or in any Related Agreements contains any false or misleading statement of a material fact, or omits any material fact required to be stated therein or necessary in order to make the statements therein not false or misleading.

ARTICLE V REPRESENTATIONS AND WARRANTIES OF THE PURCHASER

The Purchaser hereby represents and warrants to the Seller that the statements contained in this <u>Article V</u> are true, correct and complete as of the date hereof.

- **5.01.** Organization; Power. The Purchaser is a limited liability company duly organized, validly existing, and in good standing under the Laws of the State of Delaware and has all requisite limited liability company power and authority to own its properties and Assets and to conduct its business as it is now conducted.
- **5.02.** <u>Title to the Greenlane Interests</u>. The issuance of the Greenlane Interests to the Seller is duly authorized in compliance with the Purchaser's Governing Documents and all applicable securities Laws. Once issued, the Seller shall possess the Greenlane Interests free and clear of all Encumbrances or any restrictions on transfer other than as set forth in the Purchaser's Governing Documents, under the Securities Act, or under applicable state securities Laws.
- 5.03. Authorization and Validity of Agreement. The Purchaser has all requisite limited liability company power and authority to enter into this Agreement and each of the Related Agreements to which it is a party and to perform its obligations hereunder and thereunder. The execution and delivery of this Agreement and each of the Related Agreements to which it is a party and the performance of the obligations of the Purchaser hereunder and thereunder have been duly authorized by all necessary limited liability company action of the Purchaser, and no other limited liability company proceedings on the part of the Purchaser are necessary to authorize the execution, delivery or performance of this Agreement and each of the Related Agreements to which it is a party. This Agreement and each of the Related Agreements to which it is a party has been duly executed and delivered by the Purchaser and constitutes the Purchaser's valid and binding obligation, enforceable against the Purchaser in accordance with its terms and conditions, except as enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar Laws affecting the enforcement of creditor's rights generally or by general principles of equity (whether applied in a Proceeding at Law or equity).
- **5.04.** No Conflict or Violation. The execution, delivery and performance of this Agreement and each of the Related Agreements to which it is a party by the Purchaser does not and shall not: (a) violate or conflict with any provision of its Governing Documents; (b) violate in any material respect any applicable provision of Law; or (c)(i) require any consent or approval or (ii) violate or result in a breach of or constitute (with or without due notice or the passage of time, or both) a default under any judicial consent, order or decree or any Contract to which the Purchaser is a party or by which it or any of its Assets or properties are bound.
- submitted to the Seller the Purchaser's initial draft registration statement on Form S-1 that has been submitted to the SEC on a confidential basis, and all amendments thereto (the "Purchaser SEC Documents"). The Purchaser agrees to make available to Seller all exhibits to the Purchaser SEC Documents submitted on a confidential basis subsequent to the date hereof that are attached to the Purchaser SEC Documents ("Requested Confidential Exhibits") and will promptly make available to the Seller all Requested Confidential Exhibits to any additional Purchaser SEC Documents submitted on a confidential basis prior to the Closing Date. As of their respective dates of submission, none of the Purchaser SEC Documents contained any untrue statement of a material

fact or omitted to state a material fact required to be stated therein or necessary to make the statements made therein, in light of the circumstances in which they were made, not misleading. In no event shall changes to or the contents of subsequently filed amendments to the Purchaser SEC Documents be considered evidence that, or otherwise be the basis for a determination that, the Purchaser SEC Documents contained any untrue statement of a material fact or omitted to state a material fact required to be stated therein or necessary to make the statements made therein, in light of the circumstances in which they were made, not misleading. The financial statements of the Purchaser, including the notes thereto, included in the Purchaser SEC Documents (the "Purchaser Financial Statements"), complied as to form in all material respects with applicable accounting requirements and with the published rules and regulations of the SEC with respect thereto as of their respective dates, and have been prepared in accordance with GAAP applied on a basis consistent throughout the periods indicated and consistent with each other. The Purchaser Financial Statements fairly present the consolidated financial condition, operating results and cash flow of the Purchaser and its Subsidiaries at the dates and during the periods presented therein (subject, in the case of unaudited statements, to normal, recurring year-end adjustments). There has been no change in the Purchaser accounting policies except as described in the notes to the Purchaser Financial Statements.

- **5.06.** Broker's and Finder's Fees. No broker, finder or other Person is entitled to any commission or finder's fee in connection with this Agreement or the transactions contemplated by this Agreement as a result of any actions or commitments of the Purchaser or its Affiliates.
- **5.07.** Consents and Approvals. No consent, waiver, authorization or approval of any Governmental Entity, or of any other Person, or declaration to or filing or registration with any Governmental Entity, is required in connection with (a) the execution and delivery of this Agreement or any of the Related Agreements by the Purchaser, or any agreement, document or instrument contemplated hereby or thereby by the Purchaser, or (b) the performance by the Purchaser of its obligations hereunder or thereunder, other than the consent and approval of Fifth Third Bank under that Credit Agreement dated October 4, 2017 between Fifth Third Bank and Jacoby & Co. Inc. (the "Purchaser Required Consent").
- **5.08.** <u>Independent Investigation</u>. The Purchaser acknowledges and agrees that: (a) in making its decision to enter into this Agreement and to consummate the transactions contemplated hereby, the Purchaser has relied solely upon its own investigation and the express representations and warranties of the Seller and the Company set forth in <u>Articles III</u> and <u>IV</u> of this Agreement (including the related portions) and (b) none of the Seller, the Company, or any other Person has made any representation or warranty as to the Seller, the Company, or this Agreement, except as expressly set forth in <u>Articles III</u> and <u>IV</u> of this Agreement (including the related portions).

ARTICLE VI INDEMNIFICATION; SURVIVAL

6.01. <u>Indemnification by the Seller</u>.

- (a) Subject to the applicable provisions of this <u>Article VI</u> and the last paragraph of this <u>Section 6.01</u>, the Seller shall indemnify and hold harmless the Purchaser and its successors and assigns, members, directors, managers, partners, Personnel, representatives and agents, and those of its Affiliates (including the Company, on and after the Closing Date) (collectively, the "*Purchaser Indemnified Parties*"), from and against any and all Indemnity Losses directly or indirectly arising from:
- (i) any misrepresentation or breach of any warranty regarding the Company contained in this Agreement, including, without limitation, as set forth in Article IV;
- (ii) (x) any Taxes (or the non-payment thereof) of the Company, for any Pre-Closing Tax Period (including that portion of any Straddle Period ending on the Closing Date, apportioned in accordance with Section 7.04(b)); and (y) any Taxes of any Person imposed on the Company, as transferee or successor, by Contract, pursuant to any Law (including Treasury Regulation Section 1.1502-6 or any analogous or similar state, local or foreign Law or regulation) or otherwise, which Taxes relate to an event or transaction occurring before the Closing;
- (iii) any claim based on actual fraud or willful misconduct by or on behalf of the Company arising out of factors or circumstances existing prior to or as of the Closing; and/or
 - (iv) the matters set forth on Schedule 6.01.
- (b) Notwithstanding anything to the contrary contained herein, the Seller shall indemnify the Purchaser Indemnified Parties for any Indemnity Loss arising from:
- (i) any misrepresentation or breach of any warranty of the Seller contained in this Agreement or any Related Agreement, including, without limitation, as set forth in Article III;
- (ii) any breach or failure to perform by the Seller of the Seller's covenants, obligations or agreements contained in this Agreement or any Related Agreement;
- (iii) any Taxes (or non-payment thereof) of the Seller, including the Transfer Taxes under Section 7.03; and/or
- (iv) any claim based on actual fraud or willful misconduct by or on behalf of the Seller.
- **6.02.** <u>Indemnification by the Purchaser</u>. The Purchaser shall indemnify and hold harmless the Seller and its successors, assigns, heirs, representatives and agents (collectively, the "*Seller Indemnified Parties*") from and against any and all Indemnity Losses directly or indirectly arising from or relating to (a) any misrepresentation or breach of any warranty of the

Purchaser contained in this Agreement, (b) any breach or failure to perform by the Purchaser of any of its covenants or obligations contained in this Agreement or (c) (x) any Taxes (or the non-payment thereof) of the Purchaser, for any Pre-Closing Tax Period; and (y) any Taxes of any Person imposed on the Purchaser, as transferee or successor, by Contract, pursuant to any Law (including Treasury Regulation Section 1.1502-6 or any analogous or similar state, local or foreign Law or regulation) or otherwise, which Taxes relate to an event or transaction occurring before the Closing.

6.03. Indemnification Notice; Litigation Notice.

- (a) If a party believes that it has suffered or incurred any Indemnity Loss for which it is entitled to indemnification pursuant to Section 6.01 or Section 6.02 (such party, the "Claimant"), such Claimant shall notify, as the case may be, (i) the Purchaser, in the event such Claimant is a Seller Indemnified Party, or (ii) the Seller, in the event such Claimant is a Purchaser Indemnified Party, promptly in writing (x) identifying the party or parties which such Claimant believes has or have an obligation to indemnify (the "Indemnifying Party") and (y) describing such Indemnity Loss in reasonable detail, including the amount thereof, if known (or estimated amount as necessary) (such written notice, the "Indemnification Notice"). If any Proceeding is instituted by a third party against the Claimant with respect to which the Claimant intends to claim any Liability or expense as an Indemnity Loss under this Article VI (a "Third Party Claim"), such Claimant shall promptly notify the Indemnifying Party in writing of such Third Party Claim describing such Indemnity Loss in reasonable detail, including the amount thereof, if known (or estimated amount as necessary) (such written notice, a "Litigation Notice"). For the avoidance of doubt, in the event that a Third Party Claim arises prior to the time an Indemnification Notice is issued by a Claimant, the Indemnification Notice and Litigation Notice may be combined into a single notice so long as such notice contains the information required in both an Indemnification Notice and a Litigation Notice. Notwithstanding the foregoing, the failure or delay to notifying the Indemnifying Party of any Indemnity Loss or Third Party Claim shall not affect the Claimant's rights or the Indemnifying Party's obligations hereunder, except to the extent that the Indemnifying Party demonstrates that it was materially and adversely prejudiced thereby.
- If a claim is one that is asserted directly by the Claimant against an Indemnifying Party, within thirty (30) calendar days after receipt of the applicable Indemnification Notice, the Indemnifying Party shall, by written notice to the Claimant (a "Claim Response"), either concede or deny liability for the claim set forth in such Indemnification Notice. If an Indemnifying Party shall deny liability, in whole or in part, such Claim Response shall be accompanied by a reasonably detailed description of the basis for such denial. If an Indemnifying Party fails to deliver a Claim Response within such thirty (30) calendar day period by 5:00 p.m., Eastern time, on the last day of such period, such Indemnifying Party shall be deemed to have conceded, subject only to the limitations set forth herein, the entire amount of such claim and, subject to the limitations set forth in this Article VI, the Claimant shall be entitled to the entire amount of such Indemnity Loss. If an Indemnifying Party denies liability for a claim, in whole or in part, the Purchaser and the Seller shall attempt to resolve such dispute as promptly as possible. If the Purchaser and the Seller fail to resolve such dispute within thirty (30) calendar days after receipt of the Claim Response corresponding to such dispute, any party may commence appropriate legal Proceedings in order to obtain a final judgment of a court of competent jurisdiction that is not subject to further appeal as provided in Section 10.12.

6.04. Defense of Third Party Claims. Upon receipt of a Litigation Notice, the applicable Indemnifying Party shall have thirty (30) calendar days after receipt of a Litigation Notice to notify the Claimant in writing that it elects to conduct and control any Proceeding with respect to an identifiable claim (the "Election Notice") with legal counsel reasonably satisfactory to the applicable Indemnified Parties so long as the Third Party Claim (i) seeks solely money damages (and not injunctive or other equitable relief) and, in the event that the Seller or any of its Affiliates would be the Indemnifying Party, (ii) will have no continuing material adverse effect on the Business or the Company. In the event the Indemnifying Party so assumes the conduct and control of any such Third Party Claim, such assumption of the conduct and control by the Indemnifying Party shall conclusively establish for purposes of this Agreement that all Indemnity Losses incurred by the Indemnified Parties in connection with such Third Party Claim are within the scope of and subject to indemnification hereunder subject to the limitations set forth in this Article VI. If the Indemnifying Party does not give the foregoing Election Notice during such thirty (30) day period, then the Claimant shall have the right (but not the obligation) to defend, contest, settle or compromise such Third Party Claim in the exercise of its reasonable discretion. If the Indemnifying Party timely gives the foregoing Election Notice, then the Indemnifying Party shall have the right to undertake, conduct and control, at the Indemnifying Party's sole reasonable cost and expense, the conduct and settlement of such Third Party Claim, and the Claimant shall cooperate, at the Indemnifying Party's sole reasonable cost and expense, including by providing reasonable access during regular business hours to records and Personnel of the Company, as applicable, to the Indemnifying Party in connection therewith; provided, however, that (i) the Indemnifying Party shall permit the Claimant to participate in such conduct or settlement through legal counsel chosen by the Claimant, but the fees and expenses of such legal counsel shall be borne solely by the Claimant, and (ii) the Indemnifying Party shall have authority to compromise or settle any such claim without the prior written consent of the Claimant if such compromise or settlement (1) contains an unconditional release from all Liability of the Claimant and its Affiliates with respect to such Third Party Claim, (2) does not result in any Liability to or equitable relief against the Claimant and its Affiliates, (3) would not restrict the future activity of the Claimant or any of its Affiliates and (4) would not result in the admission or finding of a violation of Law by the Claimant or any of its Affiliates.

6.05. Survival.

- (a) Claims for indemnification under <u>6.01(a)(i)</u>, <u>6.01(b)(i)</u>, <u>6.01(b)(ii)</u>, and <u>6.02(a)</u> shall only be valid to the extent that such claims are made prior to the twelve (12) month anniversary of the Closing Date (the "*Survival Period*"). If an Indemnification Notice or Litigation Notice is provided with respect to such claim prior to the expiration of such period, then the applicable representations and/or warranties shall survive only as to such claim until such claim has been fully resolved.
- (b) Claims arising under the other provisions of <u>Sections 6.01</u> and <u>6.02</u>, including, without limitation, <u>Section 6.01(a)(iv)</u>, or from any breach of the Fundamental Representations may be made up to the applicable statute of limitation for such claim, without limitation.
- (c) The Purchaser shall use its best efforts to (i) maintain the Policies for the Survival Period or (ii) add the Seller as insureds under Purchaser's existing insurance policies of

comparable limits and scope of coverage, and shall cooperate with the Seller in filing any claims thereunder should any such claims require indemnification pursuant to <u>Section 6.01</u>.

6.06. Additional Indemnification Provisions.

- (a) The Seller indemnity obligations for Indemnity Losses arising under Sections 6.01(a)(i), 6.01(b)(i) or 6.01(b)(ii), shall not exceed 12.5% of the Greenlane Interests (the "Indemnification Cap"); provided, however, that the Indemnification Cap shall not apply to indemnification for Indemnity Losses the Purchaser Indemnified Party may suffer resulting from, arising out of, relating to, the breach of any of the following (the "Specified Provisions"); (i) the Fundamental Representations, (ii) the representations and warranties set forth in Section 4.06 (Tax Matters) or (iii) the covenants of the Seller contained in this Agreement, and no such Indemnity Losses shall be taken into account to determine whether the Indemnification Cap has been exceeded with respect to claims for indemnification not referred to in this proviso.
- (b) No Indemnifying Party shall be required to indemnify applicable Indemnified Parties for Indemnity Losses arising under Sections 6.01(a)(i), 6.01(b)(i), or 6.01(b)(ii) as applicable, unless and until, and only to the extent that the aggregate amount of all such Indemnity Losses for which such Indemnified Parties are otherwise entitled to indemnification pursuant to this Article VI exceeds an amount equal to 0.75% of the Greenlane Interests (the "Basket Amount"), following which the Indemnified Parties shall be entitled to recover all of their respective Indemnity Losses after the Basket Amount; provided, however, that the limitations in this Section 6.06(a) shall not apply to Indemnity Losses from claims for indemnification arising out of the Specified Provisions.
- (c) For purposes of this <u>Section 6.06</u>, (x) if any obligation to pay a claim for indemnification under this <u>Article VI</u> arises prior to the consummation of an IPO of Purchaser, then the value of the Greenlane Interests for purposes of determining the Indemnification Cap and Basket Amount shall be \$9,760,000 and (y) if any claim for indemnification under this <u>Article VI</u> arises on or subsequent to the consummation of an IPO of Purchaser, then the value of the Greenlane Interests for purposes of determining the Indemnification Cap and Basket Amount shall be based upon Purchaser's IPO valuation prior to the beginning of trading.
- (d) For purposes of this <u>Article VI</u>, any inaccuracy in or breach of any representation or warranty (and any Indemnity Losses arising therefrom or related thereto) shall be determined without regard to any materiality, "*Material Adverse Effect*" or similar qualification contained in or otherwise applicable to such representation or warranty.
- **6.07.** Special Rule for Fraud. Notwithstanding anything to the contrary contained in this Agreement, in the event of any breach of a representation or warranty by any party hereto that constitutes a criminal, fraudulent, or otherwise intentionally wrongful action or omission, by or on behalf of the Seller, on the one hand, or the Purchaser, on the other hand, then (a) such representation or warranty shall survive indefinitely, and (b) the limitations set forth in Section 6.06, as applicable, shall not apply to any Indemnity Loss that the Purchaser Indemnified Parties with respect to the Seller that committed the fraud or the Seller Indemnified Parties, as the case may be, may suffer, sustain or become subject to, as a result of, arising out of, relating to or in connection with any such breach.

- 6.08. Sole Remedy. Subject to Section 6.07, the right to indemnification under this Article VI, subject to all of the terms, conditions and limitations hereof, shall constitute the sole and exclusive right and remedy available to any party hereto (or any specified third party) for any actual or threatened breach of this Agreement, and none of the parties hereto shall initiate or maintain any Proceeding against any other party hereto which is directly or indirectly related to any breach or threatened breach of this Agreement, except that any party may pursue legal or equitable relief against any other party for any claim based upon fraud or intentional misconduct by or on behalf of the party that committed such fraud or intentional misconduct. The foregoing shall not limit the rights of a party to seek or obtain injunctive relief based upon the actual breach of any covenant contained herein and/or to enforce each of the covenants contained herein, pursuant to the terms of this Agreement (including pursuant to Section 6.07).
- **6.09.** Determination of Loss Amount. The amount of any and all Indemnity Losses under this Article VI will be (a) determined net of any amounts actually recovered by any Indemnified Party or any of such of Indemnified Party's Affiliate or pursuant to any insurance policy or title insurance policy pursuant to which or under which such Indemnified Party or such Indemnified Party's Affiliates is a party or has rights (collectively, "Alternative Arrangements") less any Indemnity Losses incurred in obtaining the amount recovered under such Alternative Arrangements and (b) reduced to take account of any Tax benefit actually realized by the Indemnified Party arising from such Indemnity Losses (determined on a with and without basis).
- **6.10.** Adjustments to Contribution Consideration. Except as required by applicable Law, all indemnification payments under this Article VI shall be treated as an adjustment to the Contribution Consideration for all Tax purposes.

ARTICLE VII OTHER AGREEMENTS

- **7.01.** Conduct of the Business. Except as may be otherwise expressly contemplated by this Agreement or required by applicable Law, or as the Purchaser may otherwise consent to in writing, from the date hereof until the Closing, the Company shall, and the Seller shall cause the Company to:
 - (a) conduct its business only in the Ordinary Course of Business; and
- (b) maintain and preserve intact its current organization, business and operations and to preserve the rights, goodwill and relationships of its employees, customers, lenders, suppliers, regulators and others having business relationships with such Person and its Subsidiaries, which shall include, without limitation:
 - (i) the preservation and maintenance of all of its Licenses and Permits;
 - (ii) the payment of its Debt, Taxes and other obligations when due;
- (iii) the maintenance of the properties and Assets owned, operated or used by it in the same condition as they were on the date of this Agreement, subject to reasonable wear and tear;

- (iv) the continuance in full force and effect, without modification, of the Policies, except as required by applicable Law;
- (v) the maintenance of its books and records in accordance with the Ordinary Course of Business;
- (vi) compliance in all material respects with all applicable Laws in respect of employees, workers, independent contractors and consultants;
- (vii) the performance of all of its obligations under all Material Contracts relating to or affecting its properties, Assets or Business;
 - (viii) compliance in all material respects with all applicable Laws; and
- (ix) not taking or permitting any action that would cause any of the changes, events or conditions described in <u>Section 4.07</u> to occur.

7.02. Confidential Information.

- The Seller acknowledges and agrees that the Confidential Information of (a) the Company is an Asset that the Purchaser will acquire pursuant to this Agreement. For purposes of this Agreement, "Confidential Information" shall mean the Company's trade secrets, other Intellectual Property and other information regarding the Company, the Business and the other business operations of the Company, which information: (i) was used in the Business and was proprietary to, about or created by the Company (including the Company's Personnel) for use in the Business; (ii) is used in the Business as of the Closing Date and is proprietary to, about or created by the Company (including the Company's Personnel) for use in the Business; (iii) is designated and/or, in fact, treated as confidential by the Company; or (iv) is not generally known by any Persons other than Personnel. The Seller agrees to maintain the confidentiality of, and refrain from using or disclosing to any Person, all Confidential Information, except to the extent disclosure of any such information is required by Law or in connection with any claims, disputes or Proceedings against the Purchaser. Notwithstanding the foregoing, "Confidential Information" shall not include information which: (1) was in the public domain on the date hereof or comes into the public domain other than through the fault or negligence of the Seller; (2) was or is independently developed by the Seller after the Closing Date without making use of any Confidential Information; (3) is required to be disclosed during the course of pursuing or defending indemnification claims (or the matters underlying such indemnification claims) or in connection with any disputes between the Purchaser, on the one hand, and the Seller, on the other hand; or (4) is required to be disclosed pursuant to applicable Laws or regulations or the order of any court or Governmental Entity, provided that the Seller shall first notify the Purchaser and the Company of any such order and afford the Purchaser and/or the Company the opportunity to seek a protective order relating to any such disclosure.
- (b) If the Seller or any of its Affiliates (other than the Company) is required by interrogatories, requests for information or documents, subpoenas or similar processes to disclose any Confidential Information, such Person shall provide the Purchaser with prompt prior written notice of such request or requirement so that the Purchaser may seek an appropriate protective order (and if the Purchaser seeks such an order, the Seller will, and will cause the Seller's

representatives to, provide such cooperation, at the expense of the Purchaser, as such the Purchaser shall reasonably request). If, in the absence of a protective order, the Seller or the Seller's representative(s) is nonetheless required to disclose Confidential Information, the Seller or representative(s), as the case may be: (i) may, and will cause each of the Seller's representatives to, disclose only that portion of the Confidential Information that they are legally compelled to disclose; and (ii) shall, and shall cause each of the Seller's representatives to, at the request of the Purchaser, use its commercially reasonable efforts, at the expense of the Purchaser, to obtain assurance that confidential treatment will be accorded to such Confidential Information.

7.03. Transfer Taxes. All transfer, documentary, sales, use, stamp, duty, recording, registration, value added and other such similar Taxes and fees (including any penalties, interest and additions to Tax) (collectively, "Transfer Taxes") incurred in connection with this Agreement and Related Agreements shall be borne and paid by the Seller. The Seller shall, at its own expense, prepare and timely file any Tax Return or other document required to be filed by it (if any) with respect to such Taxes or fees to the extent permitted by applicable Law; provided, however that the Purchaser shall cooperate with the Seller in the preparation and filing of all such Tax Returns or other applicable documents for or with respect to Transfer Taxes, including timely signing and delivering such Tax Returns and documents as may be necessary or appropriate to file such Tax Returns or establish an exemption from (or otherwise reduce) Transfer Taxes.

7.04. Preparation of Tax Returns; Payment of Taxes.

- (a) The Seller shall, at the Seller's expense, prepare, or cause to be prepared, all income Tax Returns with respect to the Company for the Tax period ending on the Closing Date ("Pre-Closing Income Tax Returns"). Such Pre-Closing Income Tax Returns shall be prepared in a manner that is consistent with the prior practice of the Company, except as required by applicable Law. At least twenty (20) days prior to filing such Pre-Closing Income Tax Returns (taking into account any extension), the Seller shall submit a copy of such Pre-Closing Income Tax Returns to the Purchaser for the Purchaser's review, comment and approval. The Seller shall revise, or cause to be revised, such Pre-Closing Income Tax Returns to reflect the Purchaser's comments to such Pre-Closing Income Tax Returns, if any, prior to filing each such Pre-Closing Income Tax Return with the applicable Governmental Entity. The Company shall timely pay to the appropriate Governmental Entity the full amount of any Taxes due and payable by the Company with respect to such Pre-Closing Income Tax Returns. The Seller shall pay to the Purchaser no later than five (5) Business Days before the due date of such Pre-Closing Income Tax Return (taking into account any extension) the amount equal to the Taxes payable by the Company with respect to such Pre-Closing Income Tax Return.
- (b) The Purchaser shall, at its expense, prepare and timely file, or cause to be prepared and timely filed, (i) all Tax Returns with respect to the Company for any Tax period ending on or prior to the Closing Date but that are required to be filed after the Closing Date (other than Pre-Closing Income Tax Returns, which are governed by Section 7.04(a)), and (ii) any Tax Return required to be filed by the Company for a Straddle Period (a "Straddle Period Tax Return"). All such Tax Returns shall be prepared and filed in a manner that is consistent with the prior practice of the Company, except as required by applicable Law. With respect to Taxes of the Company relating to a Straddle Period, the parties agree that the portion of such Tax that relates to the portion of such Straddle Period ending on the Closing Date shall (1) in the case of any Taxes

other than Taxes based upon or related to income, receipts, profits, wages, capital, net worth or expenses, be deemed to be the amount of such Tax for the entire Straddle Period *multiplied* by a fraction (A) the numerator of which is the number of days in the portion of the Straddle Period ending on the Closing Date and (B) the denominator of which is the total number of days in the entire Straddle Period, and (2) in the case of any Tax based upon or related to income, receipts, profits, wages, capital, net worth or expenses, be determined as though the taxable year of the Company terminated at the close of business on the Closing Date. The Company shall pay to the Purchaser at least five (5) days before the filing of such Tax Return (taking into account any extension) the portion of the Taxes shown as due on such Tax Return (or, with respect to a Straddle Period Tax Return, the portion of the Taxes shown as due on such Tax Return that relate to the portion of such Straddle Period ending on the Closing Date (as determined pursuant to this Section 7.04(b))).

7.05. <u>Cooperation on Tax Matters</u>.

- The parties hereto shall cooperate, and shall cause their respective (a) representatives to cooperate, including by agreeing to furnish or cause to be furnished to the other, upon request, as promptly as practicable, such information and assistance relating to Taxes, including access to books and records, as is reasonably necessary in preparing and filing all Tax Returns, in making any election relating to Taxes, in handling audits, examinations, investigations and administrative, court or other Proceedings relating to Taxes, in resolving all disputes, audits and refund claims with respect to Tax Returns and Taxes and in all other relevant Tax matters. Any information obtained by any party or its Affiliates from another party or its Affiliates in connection with any Tax matters to which this Agreement relates shall be kept confidential, except: (i) as may be otherwise necessary (A) in connection with the filing of Tax Returns or an audit or other Proceeding relating to Taxes or as may be otherwise required by applicable Law, (B) to enforce rights under this Agreement or (C) to pursue any claim for refund or contest any proposed Tax assessment; or (ii) for any external disclosure in audited financial statements or regulatory filings which a party reasonably believes is required by applicable Law or stock exchange or similar applicable rules.
- (b) Notwithstanding the provisions of Section 7.05(a), and in addition to all other obligations imposed by this Section 7.05, the Seller and the Purchaser agree to give the other party reasonable written notice prior to transferring, destroying or discarding any Files and Records with respect to Tax matters and, if the other party so requests, shall allow the other party to take possession of such Files and Records.

7.06. Tax Contests

- (a) The Purchaser or the Company, on the one hand, and the Seller, on the other hand, shall promptly notify each other upon receipt by such party of written notice of any inquiries, claims, assessments, audits, Proceeding or similar events with respect to Taxes or Tax Returns of the Company relating to a Pre-Closing Tax Period (any such inquiry, claim, assessment, audit, Proceeding or similar event, a "*Tax Matter*").
- (b) The Purchaser shall have sole control of the conduct of all Tax Matters, including any conduct, control, settlement or compromise thereof; provided, however, that the

Purchaser shall not settle or compromise any such Tax Matter without the prior written consent of the Seller (not to be unreasonably withheld, conditioned or delayed).

7.07. Release. Effective as of the Closing, the Seller, on behalf of itself, and the Seller's Affiliates and each of its and their respective heirs, successors and assigns (collectively, the "Releasing Parties"), hereby releases, acquits and forever discharges the Company, and any and all of its successors and assigns, together with all their present and former equity holders, directors, managers, officers and employees (collectively, the "Released Parties"), from any and all manner of claims, actions, suits, damages, demands and Liabilities whatsoever in Law or equity, whether known or unknown, liquidated or unliquidated, fixed, contingent, direct or indirect, which the Releasing Party ever had, has or may have against any of the Released Parties for, upon, or by reason of any matter, transaction, act, omission or thing whatsoever arising under or in connection with any of the Released Parties, from facts or circumstances existing from the beginning of time to and including the Closing Date, other than obligations arising under this Agreement or any transactions or documents contemplated thereby or executed in connection therewith.

7.08. Employees; Employee Benefits.

- (a) The Purchaser may, in its sole and absolute discretion, make offers of employment to such Employees, on such terms and conditions, as the Purchaser shall determine (such Employees who accept the Purchaser's offer of employment are hereinafter collectively referred to as "*Transferred Employees*"). Following the Closing, the Seller shall provide the Purchaser with reasonable cooperation and information in connection with the foregoing.
- (b) No agreement, understanding or arrangement entered into by an Employee and the Company prohibits or restricts (or shall prohibit or restrict) an Employee who is subsequently employed by the Purchaser from disclosing Confidential Information of the Company to the Purchaser or its Affiliates after the Closing, or if any such agreements do prohibit or restrict disclosure, the Company hereby releases the Employees from such restrictions so as to allow disclosures to be made to the Purchaser after the Closing.
- (c) The Seller and the Company shall be responsible for, and shall jointly and severally indemnify and hold harmless (as set forth in <u>Article VII</u>) the Purchaser from and against, any and all severance, termination, retention, "golden parachute," unemployment compensation or any similar payment or other Liabilities or obligations with respect to any Employee attributable to the termination of their employment with the Company, relating to the period of time that the Company employed any Employee up to the Closing Date, whether pursuant to corporate policy, any benefit plan, or by Law (domestic or foreign and including, but not limited to, any liability under the WARN Act and the provisions of Section 4980B of the Code and Part 6 of the Subtitle B of Title I of ERISA), and whether or not pursuant to individual agreement or commitment or group plan.
- (d) The Purchaser shall not assume any obligation or liability for and the Company shall remain responsible for (i) any vested benefits accrued by Transferred Employees, Employees and former employees under any benefit plans, whether or not set forth in any employment agreement with the Company, including, without limitation, under any equity appreciation or stock option plans, (ii) any and all obligations and Liabilities to Employees and

former employees of the Company related to any employment or service performed or otherwise, which were incurred or accrued prior to the Closing, including, without limitation, under any benefit plans that the Company is or becomes obligated to provide prior to or after the Closing, including, without limitation, retirement benefits, disability payments and the obligation to provide COBRA continuation coverage to such former employees and their beneficiaries, whether payable prior to or after the Closing, and (iii) severance or any other Liabilities arising or resulting from the consummation of the transactions contemplated by this Agreement or the termination of any Employees in connection therewith under the WARN Act, or any similar United States Law. The Purchaser is not the successor employer of the Company's employees for any purpose and is under no obligation to employ any such employees.

- **7.09.** Non-Competition; Non-Solicitation. Edward Kilduff hereby acknowledges that: (i) in addition to disposing of his indirect beneficial ownership interest in the Company as set forth in this Agreement, he is selling the goodwill of the Company associated with or attributable to the Contributed Interests; (ii) he has contributed to the development of the goodwill of the Company; and (iii) the parties hereto have agreed upon the consideration for the Contributed Interests to specifically include and reflect such sale of goodwill. In consideration of the sale of Edward Kilduff's indirect beneficial ownership in the Company, Edward Kilduff agrees that:
- (a) Except as provided in this <u>Section 7.09</u>, during the period commencing at the Closing and up to and through the later of (x) the date arising three (3) years after the Closing Date and (y) the date arising two (2) years after the termination of his employment by the Purchaser or any of its Affiliates, (the "*Restricted Period*"), Edward Kilduff, whether directly or indirectly, shall not, whether for himself or on behalf of or in conjunction with any other Person in any capacity (as a principal, equity holder, joint-venturer, partner, director, officer, agent, executive, consultant, contractor, employee, lender or otherwise) (collectively, the "*Covenanting Party*"):
- (i) induce, solicit, hire, recruit or attempt to persuade any Person to terminate such Person's employment or other relationship with the Company, the Purchaser or any of their Affiliates (collectively, "Company Parties") or not to establish an employment or other relationship with any Company Party, whether or not such Person is or would be an employee, consultant, contractor, manager, director, officer and/or employee, whether or not such relationship is or would be pursuant to a written or oral agreement and whether or not such relationship is for a specific period of time or is at-will;
- (ii) employ or establish a business relationship with (or attempt to employ or establish a business relationship with), or encourage or assist any Person to employ or establish a business relationship with, any individual who is, was at any time within the six (6) month period prior to the date hereof, or will be at any time during the Restricted Period, an employee, consultant, contractor, manager, officer, director or employee of any Company Party;
- (iii) direct or engage in any act which may interfere with or materially and adversely affect, alter or change the relationship (contractual or otherwise) of any Company Party with any Person that is a Client, Prospective Client, vendor, supplier or contractor of any Company Party, or otherwise induce or attempt to induce any such Person to cease doing business, reduce or otherwise limit its business with any Company Party;

- (iv) solicit business from any Client or Prospective Client, or do business with any Client or Prospective Client, involving the Business or any business that is competitive, directly or indirectly, with the Business; or
- (v) engage or participate in, manage, operate, be employed by, consult with, advise, or be financially interested in, any Person engaged in the Business anywhere where any Company Party transacts the Business during the three (3) year period immediately prior to the Closing Date (provided, however, that nothing contained in this Section 7.09 shall prevent Edward Kilduff from holding for passive investment less than five percent (5%) of any class of equity securities of a company whose securities are publicly traded on a national securities exchange or in a national market system).
- (b) For purposes of this <u>Section 7.09</u>, "*Client*" means a Person for whom or which any Company Party performed services or to whom or which any Company Party sold or licensed its products, during the prior twelve (12) months. "*Prospective Client*" means Persons whose business was solicited by any Company Party during the prior twelve (12) months.
- (c) This Section 7.09 shall not restrict or limit Edward Kilduff from: (i) soliciting or hiring (x) any employee or former employee (1) whose employment or relationship with any Company Party was terminated at least (A) 180 days before such solicitation in the event such employment or relationship was terminated by the applicable Company Party or (B) one year before such solicitation in the event such employment or relationship was terminated by the former employee, or (2) by general solicitations not specifically directed at any such employee; or (ii) performing services for the Purchaser, the Company or any Affiliate thereof pursuant to any employment agreement to be entered into at Closing.
- (d) Edward Kilduff acknowledges that the restrictions contained in this Section 7.09 are reasonable and necessary to protect the legitimate interests of the Purchaser and its Affiliates (including the Company) and constitute a material inducement to the Purchaser to enter into this Agreement and the Related Agreements and to consummate the transactions contemplated by this Agreement and the Related Agreements. In the event that any covenant contained in this Section 7.09 should ever be adjudicated to exceed the time, geographic, product or service, or other limitations permitted by applicable Laws in any jurisdiction, then any court is expressly empowered to reform such covenant, and such covenant shall be deemed reformed, in such jurisdiction to the maximum time, geographic, product or service, or other limitations permitted by applicable Laws. The covenants contained in this Section 7.09 and each provision hereof are severable and distinct covenants and provisions. The invalidity or unenforceability of any such covenants or provision as written shall not invalidate or render unenforceable the remaining covenants or provisions hereof, and any such invalidity or unenforceability in any jurisdiction shall not invalidate or render unenforceable such covenant or provision in any other jurisdiction.
- **7.10.** Remedies. The Seller acknowledges that remedies at Law may be inadequate to protect the Purchaser and the Company against any actual or threatened breach of Section 7.02 and Section 7.09 by the Seller. Without limiting any other rights or remedies available to the Purchaser, the Purchaser will, in addition to any and all other rights and remedies that may be available to it in respect of such breach, be entitled to obtain equitable relief from an

actual or threatened violation of <u>Section 7.02</u> and <u>Section 7.09</u>, including specific performance and temporary or permanent injunctive relief. To obtain any such equitable relief, the Purchaser need not post a bond or other security or prove actual damages.

ARTICLE VIII CONDITIONS TO CLOSING

- **8.01.** Conditions to the Obligations of the Purchaser. The obligation of the Purchaser to contribute the Contribution Consideration is conditioned on the satisfaction of the following conditions on or prior to the Closing Date (any one or more of which can be waived by a written waiver executed by the Seller):
- (a) The representations and warranties contained in <u>Article III</u> and <u>Article IV</u> of this Agreement shall be true and correct in all respects (in the case of any representation or warranty qualified by materiality or Material Adverse Effect) or in all material respects (in the case of any representation or warranty not qualified by materiality or Material Adverse Effect) as of the date hereof and as of the Closing Date as though made at and as of the Closing Date (except those representations and warranties that address matters only as of a specified date, the accuracy of which shall be determined as of that specified date in all respects).
- (b) The Company and the Seller shall have performed and complied in all material respects with their respective covenants and agreements required to be performed, satisfied or complied with by them hereunder on or prior to the Closing Date.
- (c) From the date of this Agreement, there shall have been no Material Adverse Effect on the Company.
- (d) No Proceeding shall have been instituted or threatened or claim or demand made against the Company, the Seller or the Purchaser seeking to restrain or prohibit, or to obtain damages with respect to, the consummation of the transactions contemplated hereby, and no Law, order, decree or ruling shall be in effect, or shall have been issued, enacted, entered, promulgated or enforced by a Governmental Entity, that restrains, enjoins or otherwise prohibits the consummation of the transactions contemplated hereby.
 - (e) The Company shall have obtained the consents listed on <u>Schedule 4.04</u>;
- (f) The Company shall have received executed non-disclosure agreements from each of its members, in a form approved by Purchaser, and, following Purchaser's approval of such non-disclosure agreements and delivery by the Company to Purchaser of the executed non-disclosure agreements, the Company shall have delivered an Information Statement to each of its members, in a form reviewed and approved by Purchaser;
- (g) The Purchaser shall have received all of the Closing deliveries set forth in Section 2.02.
- (h) The representations and warranties of the Company contained in this Agreement and any Related Agreement shall be true and correct, in all material respects, as of the Closing, with the same force and effect as if made as of the Closing, other than such representations

and warranties as are specifically made as of another date, and all the covenants contained in this Agreement and any Related Agreement to be complied with by the Company on or before the Closing Date shall have been complied with, in all material respects, and the Purchaser shall have received a certificate of a duly authorized Person on behalf of the Company to such effect signed by a duly authorized Person thereof.

- **8.02.** Conditions to the Obligations of the Seller and the Company. The obligation of the Seller and the Company to contribute the Contributed Interests is conditioned on the satisfaction of the following conditions on or prior to the Closing Date (any one or more of which can be waived by a written waiver executed by the Seller):
- (a) The representations and warranties contained in <u>Article V</u> of this Agreement shall be true and correct in all respects (in the case of any representation or warranty qualified by materiality or Material Adverse Effect) or in all material respects (in the case of any representation or warranty not qualified by materiality or Material Adverse Effect) as of the date hereof and as of the Closing Date as though made at and as of the Closing Date (except those representations and warranties that address matters only as of a specified date, the accuracy of which shall be determined as of that specified date in all respects).
- (b) The Purchaser shall have performed and complied in all material respects with the covenants and agreements required to be performed, satisfied or complied with by it hereunder on or prior to the Closing Date.
- (c) No Proceeding shall have been instituted or threatened or claim or demand made against the Company, the Seller or the Purchaser seeking to restrain or prohibit, or to obtain damages with respect to, the consummation of the transactions contemplated hereby, and no Law, order, decree or ruling shall be in effect, or shall have been issued, enacted, entered, promulgated or enforced by a Governmental Entity, that restrains, enjoins or otherwise prohibits the consummation of the transactions contemplated hereby.
 - (d) The Purchaser shall have obtained the Purchaser Required Consent;
- (e) From the date of this Agreement, there shall have been no Material Adverse Effect on the Purchaser.
- (f) The Seller shall have received all of the Closing deliveries set forth in Section 2.03.
- (g) The Seller shall have received a certificate of an executive officer or Person with appropriate authority of the Purchaser as to the incumbency and signature of the Persons executing this Agreement and the Related Agreements.

ARTICLE IX TERMINATION

9.01. <u>Termination</u>. This Agreement may be terminated, and the transactions contemplated hereby may be abandoned, at any time prior to the Closing:

- (a) By mutual written consent of the Purchaser and the Seller (on behalf of itself and the Company);
- (b) By the Purchaser, if any of the conditions set forth in <u>Section 8.01</u> shall have become incapable of fulfillment, and shall not have been waived by the Purchaser;
- (c) By the Purchaser, if the Closing shall not have occurred prior to the initial public filing of the Purchaser's registration statement on Form S-1 in connection with an IPO by Purchaser;
- (d) By the Seller (on behalf of itself and the Company), if any of the conditions set forth in Section 8.02 shall have become incapable of fulfillment, and shall not have been waived by the Company;
- (e) By either the Purchaser or the Seller (on behalf of itself and the Company), if (i) any Governmental Entity shall have issued an order, decree or ruling or taken any other action permanently restraining, enjoining or otherwise prohibiting the transactions contemplated hereby, and such order, decree, ruling or other action shall have become final and non-appealable, or (ii) a Law shall be in effect that makes consummation of the transactions contemplated hereby illegal or otherwise prohibits or prevents consummation of the transactions contemplated hereby;

<u>provided</u>, <u>however</u>, that the party seeking termination pursuant to any of clauses (b) through (d) is not then in material breach of any of its representations, warranties, covenants or agreements contained in this Agreement.

9.02. Effect of Termination. In the event of termination by the Seller or the Purchaser pursuant to this Article IX, written notice thereof shall forthwith be given to the other and the transactions contemplated by this Agreement shall be terminated, without further action by any party. Termination of this Agreement pursuant to this Article IX shall terminate all Liabilities and obligations of the parties and there shall be no Liability hereunder on the part of any party, except that Section 7.02, this Article IX and Article X shall survive any termination of this Agreement. Notwithstanding the foregoing, the termination of this Agreement pursuant to this Article IX shall not relieve any party of any Liability for any intentional inaccuracy or breach of any representation or warranty or any intentional breach or nonperformance of any covenant or obligation hereunder prior to such termination and any Indemnity Losses in connection therewith, and any such termination shall not be deemed to be a waiver of any available remedy for any such intentional inaccuracy, breach or nonperformance.

ARTICLE X MISCELLANEOUS

10.01. Public Announcements. No party to this Agreement, other than the Purchaser, shall make any public announcement of the transactions provided for in, or contemplated by, this Agreement or any of the Related Agreements unless the form and substance of the announcement is agreed upon by the Purchaser at the Purchaser's sole, absolute and unfettered discretion, or unless public disclosure is necessary to comply with applicable Laws, provided the Person required to make such disclosure gives the Purchaser reasonable prior notice thereof and cooperates in good faith with the Purchaser's efforts to prevent or limit such disclosure,

and shall then only make such disclosure as is necessary to comply with the applicable Laws, as so modified, if at all, by the Purchaser. The Purchaser shall not make any public announcement prior to the Closing Date without the prior consent of the Seller, which consent shall not be unreasonably withheld, conditioned or delayed. The Purchaser may make any public announcement at any time following the Closing Date.

10.02. Costs and Expenses. The Purchaser shall at its sole cost and expense bear all expenses and costs incurred by the parties herein in connection with this Agreement and the Related Agreements and the transactions contemplated by any of them, including the fees and disbursements of any legal counsel, independent accountants or any other Person or representative whose services have been used by the parties.

10.03. Further Assurances. From and after the date of this Agreement, the parties shall cooperate reasonably with each other in connection with any steps required to be taken as part of their respective obligations under this Agreement or any of the Related Agreements, and shall: (a) furnish upon request to each other such further information, (b) execute and deliver to each other such other documents, and (c) do such other acts and things, all as the other party may reasonably request for the purpose of carrying out the intent of transactions contemplated by this Agreement and the Related Agreements.

10.04. Addresses for Notices, Etc.. All notices, requests, demands and other communications that are required or may be given pursuant to the terms of this Agreement shall be in writing, and delivery shall be deemed sufficient in all respects and to have been duly given, as follows: (a) on the actual date of service if delivered personally, (b) at the time of receipt of confirmation by the transmitting party if by electronic transmission, (c) at the time of receipt if given by electronic mail to the e-mail addresses set forth in this Section 10.04, provided that a party sending notice by electronic delivery shall bear the burden of authentication and of proving transmittal, receipt and time of receipt, or (d) on the day after delivery to a nationally recognized overnight courier service during its business hours or the Express Mail service maintained by the United States Postal Service during its business hours for overnight delivery against receipt, and properly addressed as set forth in this Section 10.04:

If to the Seller or the Company (prior to the Edward Kilduff Closing Date):

Pollen Gear LLC

601 Cypress Ave., No. 405 Hermosa Beach, CA 90254 E-mail: edkilduff@mac.com

With a copy to (which copy shall not constitute notice hereunder):

Foundation Law Group LLP 445 S. Figueroa Street, Suite 3100

Los Angeles, CA 90071 Attn: Armen S. Martin

E-mail: armen@foundationlaw.com

If to the Purchaser: Greenlane Holdings, LLC

1095 Broken Sound Parkway, Suite 300

Boca Raton, FL 33487 Attn: Aaron LoCascio E-mail: aaron@gnln.com

With a copy to (which copy shall not constitute notice hereunder):

Pryor Cashman LLP 7 Times Square New York, NY 10036

Attn: Jeffrey C. Johnson Facsimile: (212) 326-0118

E-mail: jjohnson@pryorcashman.com

Any party may change its address or other contact information for notice by giving notice to each other party in accordance with the terms of this <u>Section 10.04</u>.

10.05. <u>Headings</u>. The Article, Section and paragraph headings in this Agreement are for reference purposes only and shall not affect the meaning or interpretation of this Agreement.

10.06. Construction.

- (a) The parties have participated jointly in the negotiation and drafting of this Agreement and the Related Agreements, and, in the event of an ambiguity or a question of intent or a need for interpretation arises, this Agreement and the Related Agreements shall be construed as if drafted jointly by the parties and no presumption or burden of proof shall arise favoring or disfavoring any party by virtue of the authorship of any of the provisions of this Agreement or any of the Related Agreements.
- (b) Except as otherwise specifically provided in this Agreement or any of the Related Agreements (such as by use of the words "sole", "absolute discretion", "complete discretion" or words of similar import), if any provision of this Agreement or any of the Related Agreements requires or provides for the consent, waiver or approval of a party, such consent, waiver or approval shall not be unreasonably withheld, conditioned or delayed.
- (c) The Disclosure Schedules referred to herein shall be construed with and as an integral part of this Agreement to the same extent as if they were set forth verbatim herein.
- (d) Words of any gender used in this Agreement or any of the Related Agreements shall be held and construed to include any other gender; words in the singular shall be held to include the plural and words in the plural shall be held to include the singular, unless and only to the extent the context indicates otherwise.
- (e) "Hereunder," "hereof," "hereto," "herein," and words of similar import shall be deemed references to this Agreement as a whole and not to any particular Article, Section or other provision hereof.
- (f) "Including" (and with correlative meaning "includes" or "include") means including without limiting the generality of any description preceding such term.

- (g) References to documents, instruments or agreements shall be deemed to refer as well to all addenda, appendices, Exhibits, Schedules or amendments thereto.
- 10.07. Severability. The invalidity or unenforceability of any provision of this Agreement or any of the Related Agreements shall in no way affect the validity or enforceability of any other provision of this Agreement or any of the Related Agreements. Wherever possible, each provision hereof shall be interpreted in such a manner as to be effective and valid under applicable Law. In case any one or more of the provisions contained herein shall, for any reason, be held to be invalid, illegal or unenforceable in any respect, such provision or provisions shall be ineffective to the extent, but only to the extent, of such invalidity, illegality or unenforceability, without invalidating the remainder of such invalid, illegal or unenforceable provision or provisions or any other provisions hereof, unless such a construction would be unreasonable.
- Agreements, including the Exhibits and Schedules referred to and incorporated by reference herein and therein that form a part of this Agreement and the Related Agreements, contain the entire understanding of the parties with respect to the subject matter of this Agreement and the Related Agreements. This Agreement and the Related Agreements supersede all prior agreements and understandings among the parties hereto with respect to the transactions contemplated by this Agreement and the Related Agreements, including any and all letters of intent, memoranda of understanding, term sheets or the like. This Agreement may not be amended, supplemented or otherwise modified except by a written agreement executed by each of the Purchaser and the Seller, and any such amendment, supplement or modification set forth in such executed written agreement shall be binding on all of the parties hereto.
- 10.09. No Waiver; Cumulative Remedies. Except as specifically set forth herein, the rights and remedies of the parties to this Agreement are cumulative and not alternative. No failure or delay on the part of any party in exercising any right, power or remedy under this Agreement or any of the Related Agreements shall operate as a waiver of such right, power or remedy, and no single or partial exercise of any such right, power or remedy shall preclude any other or further exercise of such right, power or remedy or the exercise of any other right, power or remedy. To the maximum extent permitted by applicable Law: (a) no claim or right arising out of this Agreement or any of the Related Agreements can be discharged by one party, in whole or in part, by a waiver or renunciation of the claim or right unless in writing signed by the other party, (b) no waiver that may be given by a party shall be applicable except in the specific instance for which it is given, and (c) no notice to or demand on one party shall be deemed to be a waiver of any obligation of that party or of the right of the party giving such notice or demand to take further action without notice or demand as provided in this Agreement or any of the Related Agreements.
- **10.10.** Parties in Interest. Nothing in this Agreement is intended to confer any rights or remedies under or by reason of this Agreement on any Person other than the Purchaser and the Seller and their respective successors and permitted assigns and the Purchaser Indemnified Parties and the Seller Indemnified Parties under Article VI; provided, however, that the Company shall be a third party beneficiary of the covenants and agreements set forth in Sections 7.02 and 7.10.

- 10.11. Successors and Assigns; Assignment. This Agreement shall be binding upon and inure to the benefit of each of the parties hereto and their respective successors and permitted assigns. The Seller shall not assign or delegate the Seller's rights or duties hereunder or under any of the Related Agreements, in whole or in part, without the prior written consent of the Purchaser. The Seller hereby consents to the Purchaser's assignment of this Agreement and the rights hereunder to its Affiliates and to the collateral assignment of the Purchaser's rights under this Agreement and the Related Agreements to lenders of the Purchaser or its Affiliates. Any purported assignment made in contravention of this Section 10.11 shall be null and void.
- **10.12.** Governing Law; Jurisdiction and Venue. This Agreement, and all claims or causes of action (whether at Law, in contract, in tort or otherwise) that may be based upon, arise out of or relate to this Agreement or the negotiation, execution or performance hereof, shall be governed by and construed in accordance with the Laws of the State of Delaware, without giving effect to any choice or conflict of Law provision or rule (whether of the State of Delaware or any other jurisdiction) that would cause the application of the Laws of any jurisdiction other than the State of Delaware. Each of the parties hereto irrevocably (a) consents to submit itself to the personal jurisdiction of the United States District Court for the District of Delaware in the event any dispute arises out of this Agreement or any of the transactions contemplated hereby, and, in connection with any such matter, to service of process by notice as otherwise provided herein, (b) agrees that it will not attempt to deny or defeat such personal jurisdiction by motion or other request for leave from any such court and (c) agrees that it will not bring any action relating to this Agreement or any of the transactions contemplated hereby in any court other than in the United States District Court for the District of Delaware. Any party may make service on another party by sending or delivering a copy of the process to the party to be served at the address and in the manner provided for the giving of notices in Section 10.04.
- 10.13. Waiver of Jury Trial. EACH PARTY HEREBY KNOWINGLY, VOLUNTARILY AND IRREVOCABLY WAIVES ALL RIGHT TO TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM (WHETHER BASED ON LAW, CONTRACT, TORT OR OTHERWISE) ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE ACTIONS OF THE PARTIES IN THE NEGOTIATION, ADMINISTRATION, PERFORMANCE AND ENFORCEMENT HEREOF.
- **10.14.** Counterparts. This Agreement may be executed in multiple original, electronic or facsimile counterparts, each of which will be deemed an original, but all of which when taken together shall constitute one and the same agreement.
- 10.15. Privileged and Confidential Information. In order to induce the Seller to contribute the Contributed Interests and to enter into this Agreement, and as a material part of the consideration for this transaction, the Purchaser agrees that the attorney-client privilege with respect to communications between the Company and Foundation Law Group LLP ("Counsel") prior to Closing, and any work product of Counsel relating to the contribution of the Contributed Interests and any and all matters up to and including the Closing (collectively, "Privileged Information") shall be deemed the Seller's and not the Company's privilege, and such work product shall be deemed to have been prepared on behalf of the Seller and not for the Company for purposes of any dispute that may arise between (i) the Seller and (ii) the Purchaser and/or the Company. The Purchaser further knowingly and irrevocably agrees to waive and to cause the

Company to waive any claim that Counsel is disqualified from representing, and agrees that Counsel may represent, the Seller (and may not represent the Company or the Purchaser) in any such dispute. For purposes of this paragraph, the term also includes the Company's predecessors and successors. All such Privileged Information, whether expressly so labeled or not, shall be delivered to the Seller at the Closing, and to the extent not so delivered shall be deemed to be held in trust by the Company and the Purchaser for the Seller's sole and exclusive benefit. In addition, all Company documents and communications regarding this Agreement, the subject matter hereof and all transactions related thereto that are not delivered by the Seller or the Company to the Purchaser or any other prospective buyer shall be deemed to be confidential information and the sole and exclusive property of the Seller ("Seller Confidential Information"). Seller Confidential Information shall be delivered to the Seller at Closing and, to the extent not so delivered, shall be deemed to be held in trust by the Company and the Purchaser for the Seller's sole and exclusive benefit. After Closing, Seller Confidential Information shall not be used by the Purchaser or the Company for any purpose detrimental to the Seller. The Purchaser and the Company acknowledge that the restrictions contained in this Section 10.15 are reasonable and necessary in order to protect the Seller's legitimate interests and that any violation thereof would result in irreparable injury to the Seller. The Purchaser and the Company therefore acknowledge and agree that, in the event of any violation thereof, the Seller shall be authorized and entitled to obtain, from any court of competent jurisdiction, preliminary and permanent injunctive relief as well as an equitable accounting of all profits or benefits arising out of such violation, which rights and remedies shall be cumulative and in addition to any other rights or remedies to which the Seller may be entitled. By way of clarification and not limitation, the terms and conditions of this Section 10.15 shall survive the Closing.

[Signatures Begin on Following Page]

IN WITNESS WHEREOF, the Parties have executed this Agreement as of the date first set forth above.

SELLER:

Pollen Gear Holdings LLC

By: dward Kilduff

Its: Manager

COMPANY:

Pollen Gear LAC

By: Edward Kilduff

Its: Manager

PURCHASER:

Greenlane Holdings, LLC

By: Jacoby & Co. Inc.

Its: Managing Member

By;_____

Name: Aaron LoCascio

Title: Co-President

By:_____

Name: Adam Schoenfeld

Title: Co-President

Signature Page to Contribution Spreement

IN WITNESS WHEREOF, the Parties have executed this Agreement as of the date first set forth above.

SEI	\mathbf{L}	\mathbf{E}	R
-----	--------------	--------------	---

Pollen Gear Holdings LLC

By: Edward Kilduff

Its: Manager

COMPANY:

Pollen Gear LLC

By: Edward Kilduff

Its: Manager

PURCHASER:

Greenlane Holdings, LLC By: Jacoby & Co. Inc. Its: Managing Member

Jane Strategick

Name: Aaron LoCascio Title: Co-President

Name: Adam Schoenfeld

Title: Co-President

REEL: 059126 FRAME: 0667

EXHIBIT A

Definitions

As used in this Agreement, the following terms have the meanings indicated below:

"Affiliate" means, with respect to any Person, any other Person that directly or indirectly controls, is controlled by or is under common control with such Person, as the case may be. As used in this definition, "control" (including, its correlative meanings "controlled by" and "under common control with") means possession, directly or indirectly, of power to direct or cause the direction of management or policies (whether through ownership of ten percent (10%) or more of outstanding voting securities or partnership or other ownership interests, by Contract or otherwise).

"Applicable Privacy and Data Security Laws" means (a) all privacy, security, data collection, data protection, data sharing, direct marketing, consumer protection, location tracking, customer tracking, behavioral marketing, and workplace privacy Laws, rules and regulations of any applicable jurisdiction and all then-current industry standards, guidelines and practices with respect to privacy, security, data protection, data sharing, direct marketing, consumer protection, location tracking, customer tracking, behavioral marketing, and workplace privacy, including the collection, processing, storage, protection and disclosure of Personal Information, and (b) the applicable data security and privacy policies of the Company and its Subsidiaries.

"Assets" means all properties, assets and rights of every kind, nature and description whatsoever whether tangible or intangible, real, personal or mixed, wherever located, (including cash, cash equivalents, accounts receivable, inventory, equipment, improvements, Intellectual Property, Contracts, real estate, claims and defenses).

- "Balance Sheet Date" has the meaning set forth in Section 4.05(a).
- "Basket Amount" has the meaning set forth in Section 6.06(b).
- "Business" means the sale and distribution of custom glass, child-resistant jars and other similar accessories for the storage of smoking/consumption products.
- "Business Day" means any day other than Saturday, Sunday and any day on which commercial banks in the State of New York are authorized by Law to be closed.
 - "Claim Response" has the meaning set forth in Section 6.03(b).
 - "Claimant" has the meaning set forth in Section 6.03(a).
 - "Closing" has the meaning set forth in Section 2.01.
 - "Closing Date" has the meaning set forth in Section 2.01.
 - "Code" means the Internal Revenue Code of 1986, as amended.

- "Company" has the meaning set forth in the Preamble.
- "Company Intellectual Property" has the meaning set forth in Section 4.12(a).
- "Company Products" has the meaning set forth in Section 4.29.
- "Company Systems" has the meaning set forth in Section 4.12(k).
- "Confidential Information" has the meaning set forth in Section 7.02(a).
- "Contract" means any contract, agreement, indenture, note, bond, loan, instrument, lease, conditional sale contract, mortgage, license, franchise, insurance policy, commitment or other arrangement or agreement, whether written or oral.
 - "Contributed Interests" has the meaning set forth in the Recitals.
 - "Contribution Consideration" has the meaning set forth in Section 1.02.
- "Convertible Notes" means the convertible notes of the Company set forth on Schedule 4.02(a).
 - "Counsel" has the meaning set forth in Section 10.15.
- "Debt" means, with respect to any Person: (a) all indebtedness of such Person for borrowed money, amounts payable under debt or like instruments, including outstanding promissory notes or letter of credit facilities and any principal, interest, overdrafts, premiums, make whole premiums or payments, fees and prepayment, termination and other penalties and expenses with respect to the foregoing; (b) for the reimbursement of amounts drawn on any letter of credit and in respect of bankers' acceptances or similar transactions; (c) all obligations of such Person under conditional sale or other title retention agreements relating to property or Assets purchased by such Person; (d) all obligations of such Person issued or assumed as the deferred purchase price of property, goods or services (including earn outs but excluding trade payables or accruals incurred in the Ordinary Course of Business); (e) all indebtedness of any other Person with respect to borrowed money, notes payable or amounts outstanding under letter of credit facilities, which amounts are secured by (or for which the holder of such indebtedness has an existing right, contingent or otherwise, to be secured by) any Encumbrance on property or Assets owned by such Person, whether or not the obligations secured thereby have been assumed; (f) all guarantees, whether direct or indirect, by such Person of indebtedness of any other Person with respect to borrowed money, notes payable or amounts outstanding under letter of credit facilities; (g) all capital lease obligations that have or should have been capitalized in accordance with GAAP; (h) customer deposits and sums received in advance from customers; (i) all amounts owed by such Person to any Person under any noncompetition, bonus, and severance agreements or retirement and termination arrangements (to the extent any amounts owed pursuant to such agreements or arrangements do not become payable as a result of any action taken by the Purchaser or any of its Affiliates post-Closing), consulting or deferred compensation arrangements arising in connection with a transaction not in the Ordinary Course of Business, (including the transaction contemplated under this Agreement); (j) any Company credit card balances that are unrelated to the Business; and (k) all negative cash and obligations arising from cash/book overdrafts. For the avoidance of doubt, all of the Company's

debt obligations to the Seller or other equityholder of the Company shall be considered Debt of the Company.

"Disclosure Schedules" has the meaning set forth in Article IV.

"Election Notice" has the meaning set forth in Section 6.04.

"Employee Plans" means any "employee benefit plan" as defined in Section 3(3) of ERISA (whether or not subject to ERISA) and each other plan, policy, program, practice, agreement, understanding or arrangement (whether written or oral) providing compensation or other benefits to any current or former director, manager, officer, employee or consultant (or to any dependent or beneficiary thereof) of the Company or any ERISA Affiliate of the Company, which is now, or was maintained, sponsored or contributed to by the Company or any ERISA Affiliate of the Company, or under which the Company or any ERISA Affiliate of the Company has or may have any obligation or liability, whether actual or contingent, including all incentive, bonus, deferred compensation, change in control, employment, severance, retirement, vacation, holiday, cafeteria, fringe benefit, medical, disability, stock purchase, sick leave, option, stock appreciation, phantom stock, restricted stock or other stock-based compensation plans, policies, programs, practices or arrangements.

"Employment Agreements" has the meaning set forth in Section 2.02(m).

"Encumbrance" means all liens (statutory or other), leases, mortgages, pledges, security interests, conditional sales agreements, charges, claims, options, easements, rights of way (other than easements of record) and other encumbrances of any kind or nature whatsoever.

"Environmental Claim" means any and all administrative, regulatory or judicial actions, suits or Proceedings as well as any actions, suits or Proceedings initiated by a third party, public or private, alleging liability arising out of or resulting from: (a) the presence or Release into the environment of any Hazardous Substance at the real property that is within the Company's possession, use or control; or (b) any violation or alleged violation of Environmental Law.

"Environmental Laws" means all federal, state or local statutes, Laws, regulations, judgments and orders relating to protection of human health or the environment, including Laws and regulations relating to Releases or threatened Releases of Hazardous Substances, or otherwise relating to the manufacture, processing, distribution, use, treatment, storage, disposal, transport or handling of Hazardous Substances.

"Environmental Permits" means all Licenses and Permits issued pursuant to Environmental Law.

"Equity Securities" means, with respect to a Person that is an entity, any shares of capital stock, limited liability company interests, options, warrants, phantom equity, convertible notes or other convertible debt instruments or other equity securities of such Person which have ever been offered or sold by such Person.

"*ERISA*" means the Employee Retirement Income Security Act of 1974 and the rules of regulations promulgated thereunder from time to time.

"ERISA Affiliate" means any Person that, together with the Company, is required to be treated as a single employer under Section 414 of the Code.

"Exchange Act" means the Securities Exchange Act of 1934, as amended.

"Files and Records" means all financial and accounting files, records and other information of the Company.

"Financial Statements" has the meaning set forth in Section 4.05(a).

"Fundamental Representation" means any representation or warranty set forth in <u>Section 3.01</u> (Organization; Power; Capacity), <u>Section 3.02</u> (Authorization and Validity of Agreement), <u>Section 3.03</u> (Title to the Contributed Interests), <u>Section 4.01</u> (Organization; Power), <u>Section 4.02</u> (Capitalization), <u>Section 5.01</u> (Organization; Power), <u>Section 5.02</u> (Title to the Greenlane Interests), and <u>Section 5.03</u> (Authorization and Validity of Agreement).

"GAAP" means the prevailing generally accepted accounting principles in the United States, in effect from time to time, consistently applied with past practices of the Company.

"Governing Documents" means, with respect to any particular entity: (a) if a corporation, the articles or certificate of incorporation and the bylaws of such entity; (b) if a general partnership, the partnership agreement and any statement of partnership; (c) if a limited partnership, the limited partnership agreement and the certificate of limited partnership; (d) if a limited liability company, the articles or certificate of organization or formation and the limited liability company operating agreement; (e) if another type of Person, any other charter or similar document adopted or filed in connection with the creation, formation or organization of the Person; (f) all equity holders' agreements, voting agreements, voting trust agreements, joint venture agreements, registration rights agreements or other agreements or documents relating to the organization, management or operation of any Person or relating to the rights, duties and obligations of the equity holders of any Person; and (g) any amendment, restatement or supplement to any of the foregoing.

"Governmental Entity" means any court, government agency, department, commission, board, bureau or instrumentality of the United States, any local, county, state, federal or political subdivision thereof, or any foreign governmental entity of any kind.

"Greenlane Interests" has the meaning set forth in Section 1.02.

"Greenlane Operating Agreement" has the meaning set forth in Section 2.02(b).

"Hazardous Substances" means any chemicals, materials or substances which are defined as or included in the definition of "hazardous substances," "hazardous wastes," "hazardous wastes," "toxic substances," "toxic pollutants" or similar terms under, or otherwise regulated under, any Environmental Law.

"Indemnification Cap" has the meaning set forth in Section 6.06(a).

"Indemnification Notice" has the meaning set forth in Section 6.03(a).

"*Indemnified Parties*" means the Purchaser Indemnified Parties or the Seller Indemnified Parties, as the context requires.

"Indemnifying Party" has the meaning set forth in Section 6.03(a).

"Indemnity Loss" means, net of any applicable insurance proceeds, actual damages, losses, obligations, Liabilities, Taxes, deficiencies, claims, Encumbrances, penalties, costs, disbursements and expenses, including reasonable costs of investigation and defense and reasonable attorneys' fees and expenses; provided, however, that "Indemnity Loss" shall not include consequential damages, indirect damages, exemplary damages, speculative damages, lost profits, diminution in value, or special or punitive damages (other than special or punitive damages payable to a third party).

"Intellectual Property" means shall mean, collectively, in the United States and all countries or jurisdictions foreign thereto, (a) all inventions (whether patentable or unpatentable and whether or not reduced to practice), all improvements thereto, and all utility and design patents and pending applications for patents of the United States and all countries and jurisdictions foreign thereto and all reissues, reexaminations, divisions, continuations, continuations-in-part, revisions, and extensions thereof; (b) all registered trademarks, registered service marks, trademark and service mark applications, unregistered trademarks and service marks, registered trade names and unregistered trade names, corporate names, fictitious names, trade dress, logos, slogans, Internet domain names, rights in telephone numbers, and other indicia of source, origin, endorsement, sponsorship or certification, together with all translations, adaptations, derivations, combinations and renewals thereof, all goodwill associated therewith; (c) all moral rights, copyrights and other rights in any work of authorship (including catalogues and related copy, databases, data, Software, and mask works), compilation, derivative work or mask work and all applications, registrations, and renewals in connection therewith; (d) all trade secrets and confidential business information (including confidential ideas, research and development, know-how, methods, formulas, compositions, manufacturing and production processes and techniques, technical and other data, designs, drawings, specifications, customer and supplier lists, pricing and cost information, and business and marketing plans and proposals); (e) all other intellectual and industrial property rights; (f) the exclusive right to display, perform, reproduce, make, use, sell, distribute, import, export and create derivative works or improvements based on any of the foregoing; and (g) all income, royalties, damages and payments related to any of the foregoing (including damages and payments for past, present or future infringements, misappropriations or other conflicts with any intellectual property), and the right to sue and recover for past, present or future infringements, misappropriations or other conflict with any intellectual property, and all other rights of any kind or nature in and to any of the forgoing.

"*IPO*" means an initial public offering of Equity Securities pursuant to an effective registration statement filed under the Securities Act or, if earlier, the registration of such Equity Securities pursuant to Section 12(b) or Section 12(g) of the Securities Exchange Act of 1934, as amended, or any Canadian Law equivalents.

"Knowledge" or words of similar import, means, with respect to any Person, the actual knowledge of such Person, in each case with such additional knowledge as such Person would

acquire after having undertaken reasonable due inquiry. With respect to the Company, "Knowledge" means the Knowledge of Edward Kilduff and Jason Brown.

"Law" means any local, county, state, federal, foreign or other law, statute, regulation, ordinance, rule, order, decree, judgment, consent decree, settlement agreement or governmental requirement enacted, promulgated, entered into, agreed or imposed by any Governmental Entity including, for the avoidance of doubt, the Controlled Substances Act of 1970, 21 U.S.C. Section 801, et seq., any regulations promulgated pursuant thereto, and any other law predicated on the violation thereof.

"Leased Real Property" has the meaning set forth in Section 4.09(a).

"Liability" with respect to any Person, means any Debt, liability or obligation of such Person of any kind, character or description, whether known or unknown, absolute or contingent, accrued or unaccrued, disputed or undisputed, liquidated or unliquidated, secured or unsecured, joint or several, due or to become due, vested or unvested, executory, determined, determinable or otherwise, and whether or not the same is required to be accrued on the financial statements of such Person. For the avoidance of doubt, with respect to the Company, Liability shall include Debt.

"Licenses and Permits" shall mean all licenses, registrations, franchises, qualifications, provider numbers, permits, orders, rights to indemnification, approvals and authorizations, if any, issued by any Governmental Entity which relate to the Business.

"Litigation Notice" has the meaning set forth in Section 6.03(a).

"Material Adverse Effect" when used with respect to the Company, means any fact, event, change, circumstance or effect that, individually or in the aggregate, has had, or is reasonably likely to have, a materially adverse effect upon the Assets, financial condition or results of operations of the Company; provided, however, that none of the following shall be deemed to constitute, and none of the following shall be taken into account in determining whether there has been, a Material Adverse Effect: (a) any adverse change, event, development or effect arising from or relating to (i) general business or economic conditions, including such conditions related to the Business, (ii) national or international political or social conditions, including the engagement by the United States in hostilities, whether or not pursuant to the declaration of a national emergency or war or the occurrence of any military or terrorist attack upon the United States, or any of its territories, possessions, or diplomatic or consular offices or upon any military installation, equipment or Personnel of the United States, (iii) financial, banking or securities markets (including any disruption thereof and any decline in the price of any security or any market index), (iv) changes in Law, (v) the taking of any action contemplated by this Agreement and the other agreements contemplated hereby, (vi) any "act of God," including weather, natural disasters and earthquakes, or (vii) changes resulting from the announcement of the execution of this Agreement or the transactions contemplated hereunder; except, with respect to clauses (i), (iii) or (iv), to the extent that such change, event, development or effect has a disproportionate effect on the business of the Company relative to other businesses in the industry in which the Company operates.

"Material Contracts" has the meaning set forth in Section 4.19.

"Multiemployer Plan" means any Employee Plan that is a "multiemployer plan" within the meaning of Section 4001(a)(3) or Section 3(37) of ERISA.

"Ordinary Course of Business" means any action taken by a Person if such action is consistent in nature, scope and magnitude with the past practices of such Person and is taken in the ordinary course of the normal day-to-day operations of such Person.

"PCI Requirements" has the meaning set forth in Section 4.18.

"Pension Plan" has the meaning set forth in Section 4.13(e).

"Permitted Encumbrance" means any of the following: (a) carriers, workmen, warehousemen, repairmen, mechanics, contractors, materialmen and other similar Persons and other liens imposed by applicable Laws; (b) with respect to the real property that is within the Company's possession, use or control, the provisions of all applicable zoning Laws; (c) purchase money liens securing rental payments under capital lease arrangements that will be released as of the Closing; (d) Encumbrances created by, or for the benefit of, the Purchaser; (e) Encumbrances for Taxes not yet due and payable or for Taxes that are being contested in good faith and by appropriate Proceedings for which adequate reserves have been provided on the books and records of the Company in accordance with GAAP; (f) Encumbrances incurred or deposits made in the Ordinary Course of Business in connection with workers' compensation, unemployment insurance and other types of social security, and mechanic's liens, carrier's Encumbrances and other Encumbrances to secure the performance of tenders, statutory obligations, contract bids, government Contracts, performance and return of money bonds and other similar obligations, incurred in the Ordinary Course of Business, whether pursuant to statutory requirements, common law or consensual arrangements; (g) Encumbrances which constitute rights of setoff of a customary nature or banker's liens, whether arising by Law or by Contract; or (h) Encumbrances on insurance proceeds in favor of insurance companies granted solely as security for financed premiums.

"Person" means any individual, corporation, partnership, joint venture, association, limited liability company, joint-stock company, trust or unincorporated organization, or any Governmental Entity, officer, department, commission, board, bureau or instrumentality thereof.

"Personal Information" means, in addition to any definitions provided by the Company for any similar term (e.g., "personally identifiable information" or "PII") in the Company's privacy policy or other public-facing statement, all information regarding or capable of being associated with an individual person or device, including (a) information that identifies, could be used to identify or is otherwise identifiable with an individual, including an individual's name, physical address, telephone number, email address, financial account number or government-issued identifier (including Social Security number and driver's license number), medical, biometric, health or insurance information, gender, date of birth, educational or employment information, religious or political views or affiliations, marital or other status, and any other data used or intended to be used to identify, contact or precisely locate an individual (e.g., geolocation data), (b) information that is created, maintained, or accessed by an individual (e.g., videos, audio or individual contact information), (c) any data regarding an individual's activities online or on a mobile device or other application (e.g., searches conducted, web pages or content visited or

viewed) and (d) Internet Protocol addresses, unique device identifiers or other persistent identifiers. Personal Information may relate to any individual, including a current, prospective or former customer or employee of any Person. Personal Information includes information in any form, including paper, electronic and other forms.

- "Personnel" means any manager, director, officer or employee of a particular Person.
- "*Pre-Closing Tax Period*" means any Tax period ending on or before the Closing Date and that portion of any Straddle Period ending on the Closing Date.
 - "Privileged Information" has the meaning set forth in Section 10.15.
- "*Proceeding*" means any judicial, administrative or arbitral actions, suits or proceedings (public or private) by or before any Governmental Entity or before any arbitrator, mediator or other alternative dispute resolution provider pursuant to any collective bargaining agreement, contractual agreement or Law, and including any audit or examination, or other administrative or court proceeding with respect to Taxes or Tax Returns.
 - "Product Plans" has the meaning set forth in Section 4.29.
 - "Purchaser" has the meaning set forth in the Preamble.
 - "Purchaser Financial Statements" has the meaning set forth in Section 5.05.
 - "Purchaser Indemnified Parties" has the meaning set forth in Section 6.01.
 - "Purchaser Required Consent" has the meaning set forth in Section 5.07.
 - "Purchaser SEC Documents" has the meaning set forth in Section 5.05.
 - "Real Property Leases" has the meaning set forth in Section 4.09(a).
- "Related Agreements" means all other agreements, documents and certificates entered into pursuant to this Agreement, except for the Employment Agreements.
- "Release" means any release, spill, emission, emptying, leaking, injection, deposit, disposal, discharge, dispersal, leaching, pumping, pouring, or migration into the atmosphere, soil, surface water, groundwater or property.
 - "Released Parties" has the meaning set forth in Section 7.07.
 - "Releasing Parties" has the meaning set forth in Section 7.07.
 - "Restricted Parties" has the meaning set forth in Section 7.09.
 - "Requested Confidential Exhibits" has the meaning set forth in Section 5.05.
 - "SEC" means the Securities and Exchange Commission.

"Securities Act" means the Securities Act of 1933, as amended and the rules and regulations promulgated thereunder from time to time.

"Seller" has the meaning set forth in the Preamble.

"Seller Confidential Information" has the meaning set forth in Section 10.15.

"Straddle Period" means any Tax period beginning on or before and ending after the Closing Date.

"Subsidiary" means a Person of which more than twenty-five percent (25%) of the voting power of the Equity Securities is owned, directly or indirectly, by the specified Person.

"Survival Period" has the meaning set forth in Section 6.05(a).

"Tax" or "Taxes" means all federal, state, local and foreign taxes (including income taxes, excise taxes, value added taxes, occupancy taxes, employment taxes, withholding taxes, escheat or unclaimed property, unemployment taxes, ad valorem taxes, custom duties and transfer taxes) and similar fees, levies, imposts, impositions, assessments and governmental charges imposed upon a Person, including all taxes and governmental charges imposed upon any of the personal properties, real properties, tangible or intangible Assets, income, receipts, payrolls, transactions, equity transfers, equity, net worth or franchises of a Person (including all sales, use, withholding or other taxes which a Person is required to collect or pay over to any government), and all related additions to tax, penalties or interest thereon.

"Tax Return" means and includes all returns, statements, declarations, estimates, forms, reports, information returns and any other documents (including all consolidated, affiliated, combined or unitary versions of the same) relating to Taxes, including all related and supporting information, in each case, filed or required by Law to be filed with any Governmental Entity in connection with the determination, assessment, reporting, payment, collection or administration of any Taxes, and including Treasury Form TD F 90-22.1 and FinCEN Form 114.

"Third Party Claim" has the meaning set forth in Section 6.03(a).

"Treasury Regulations" means the Treasury Regulations promulgated under the Code, as such Treasury Regulations may be amended from time to time. Any reference herein to a particular provision of the Treasury Regulations means, where appropriate, the corresponding successor provision.

"Warranty Claims" means any claims arising in respect of the Company's obligations under any extended warranties sold by the Company covering Company Products, but specifically excluding warranty claims arising from any manufacturer's warranty covering Company Products.

EXHIBIT B

SELLER CAPITALIZATION

Members	Units	% of Fully Diluted
30 N Capital, LLC	24,290	0.36%
Calum Wallace	9,882	0.14%
Charles Ng	49,379	0.72%
Chi Ming Tse	98,515	1.44%
CoopMill LLC	48,006	0.70%
Crimson Fund I, LLC	462,316	6.76%
Darren Plested	49,367	0.72%
David Cooper	24,735	0.36%
Dean Chapman	267,375	3.91%
Deron J. Haley	51,082	0.75%
Edward H. Kilduff	3,461,505	50.61%
Edward H. Kilduff, Trustee of the RRS 2018 Trust		
dated September 4, 2018	865,376	12.65%
George Mueller	76,377	1.12%
Harrison Future Funds LLC	49,379	0.72%
Jason Brown	397,500	5.81%
John Seitz	90,698	1.33%
Jonathann Kuo	50,954	0.74%
Joshua Roberts	23,530	0.34%
Mark Pieri	39,826	0.58%
Robert Goldstein	51,003	0.75%
Robert Scott	25,000	0.37%
Shane Blomberg	205,744	3.01%
Swamy Holdings LLC	24,668	0.36%
Thomas DesChamps	97,416	1.42%
Thomas E. Katis	102,007	1.49%
UNIS Venture Fund LLC	143,868	2.10%
VS Holdings LLC	50,000	0.73%
Totals	6,839,798	100.00%

EXHIBIT C-1

Kilduff Employment Agreement

(See attached.)

C-1

EMPLOYMENT AGREEMENT

This employment agreement (the "Agreement"), when duly executed, is made and entered into by and between Greenlane Holdings, LLC (the "Company") and Edward Kilduff ("you") (the "Company" and "you" are referred to herein in the collective as the "Parties").

- 1. <u>Title/Duties</u>. You shall be employed as President Pollen Gear LLC, and provide services and take direction from the Company and its related and affiliated entities. You will be responsible for the duties and obligations consistent with your position, which will be subject to the control and direction of the Company, and may change from time to time at the Company's sole discretion (the "Services"). For the avoidance of doubt, you shall not negotiate or enter into contracts on behalf of the Company with the Company's providers, clients or business partners other than in accordance with the Company's policies with respect to the same. You will devote your full time and best efforts to the performance of your duties for the Company. While employed by the Company, you shall not perform work on behalf of, or provide services to, third parties, whether as an employee, consultant, or otherwise. The foregoing shall not prevent you from making and expending reasonable amounts of time on passive personal investments, for educational or charitable activities, and/or serving as an advisor or a director on the boards of any non-competing companies, provided that, in each case, such activities do not individually or collectively interfere with or adversely affect the performance of your duties under this Agreement.
- 2. <u>Base Salary</u>. As full and complete consideration for the Services provided herein by you, and on the condition that you fully and faithfully perform the Services, duties and obligations required to be performed hereunder, and that you are not in breach of this Agreement, the Company shall pay you a base salary of Two Hundred and Fifty Thousand Dollars (\$250,000.00) gross, per annum (prorated for partial years) ("Base Salary") in accordance with the Company's payroll practices, and subject to customary tax withholdings and deductions.

3. Bonus and Benefits.

- 3.1 Bonus. The Company may provide a bonus program pursuant to which you may be eligible to receive additional compensation in the form of an annual cash and/or equity bonus, provided that all determinations relating to the performance goals applicable to you, your level of participation and payout opportunity, and the extent to which you have met the applicable goals, shall be made in the sole and reasonable discretion of the Board of Directors of the Company (the "Board") or such Member of the Company (as such term is defined in the Company's operating Agreement) with the authority to approve such bonus. In order to be eligible to earn any potential bonus, it is a condition precedent that you be employed on the date the bonus is paid. If you separate from the Company's employ for any reason prior to the bonus pay-out date, you will not be eligible for any bonus or any portion thereof. The Company may withhold from any bonus payments any amounts required to be withheld by law.
- 3.2 <u>Benefits</u>. If you are eligible, you may participate in the Company's health and welfare benefit programs for which the Company's other executives are generally eligible. However, nothing herein requires the Company to keep a health or welfare benefit plan or arrangement in place, or continue any health or welfare benefit plan or arrangement, and the Company may modify, amend or terminate such plan or program at any time in its sole discretion.

Furthermore, the Company may, in its sole discretion, amend, modify, or cease paying the portion (if any) of the health insurance premiums it pays on your behalf, including, but not limited to, in the event that the Company or any employee can become subject to any tax or penalty under the Patient Protection and Affordable Care Act (as amended from time to time) or Section 105(h) of the Internal Revenue Code of 1986, as amended, (the "Code"), or applicable regulations or guidance issued thereunder. The Company may withhold from any payments any amounts required to be withheld by law. Certain of the benefits described above which the Company intends to offer you may constitute taxable income to you. In the event that it does, you understand and agree that you shall be responsible for such additional taxes and the Company shall not pay any portion of such additional taxes.

4. Paid Time Off. You accrue paid time off for vacation and sick leave ("PTO") at the rate of four (4) weeks of PTO each year, subject to the policies and procedures of the Company. Except for emergencies or other unanticipated events, the days selected for your PTO must be mutually agreeable to the Company and you. You will have a six (6) week cap on PTO accruals. This cap means that once you have accrued six (6) weeks of PTO, you cannot accrue any more PTO until you use sufficient PTO to fall below the cap. Accrued but unused PTO will be paid out upon termination, pursuant to the terms of Section 5.2 below. Nothing contained herein shall be read to contravene any rights provided by applicable law, which shall govern and supersede in the event of a conflict between its terms, this Agreement and/or the Company's policies and procedures.

5. <u>Term and Termination</u>.

5.1 <u>Term</u>. This Agreement shall commence on January [__], 2019 and, subject to the terms hereof, shall terminate on December [__], 2021 (the "**Initial Term**"). This Agreement will be automatically extended for one (1) year periods subsequent to the Initial Term (together with the Initial Term, the "**Term**") unless either the Company or you provide the other written notice of your intent to terminate the Agreement thirty (30) days prior to the expiration of the Term.

5.2 <u>Termination and Termination Compensation.</u>

(a) *General*. This Agreement and your employment with the Company hereunder shall be terminated prior to the end of the Term (i) automatically upon your death, (ii) at any time by the Company in the event of your Disability, (iii) at any time by the Company upon written notice to you, or (iv) at any time by you upon written notice to the Company.

(b) *Definitions*. For purposes of this Agreement:

(i) "Cause" means any of the following: (A) your indictment for or plea of guilty or *nolo contendere* to any felony or your being changed with or entering a plea of guilty or *nolo contendere* to any crime involving moral turpitude or dishonesty, in each case other than for driving under the influence of alcohol; (B) your commission of any act or omission that, in the Company's reasonable judgement, subjects the Company to shame or ridicule, or otherwise materially diminishes the reputation and good name of the Company; (C) you are in violation of or engage in conduct which in the Company's reasonably exercised judgment, is

materially inconsistent with the Company's material policies, and procedures, business or professional interests; (D) you commit any act or omission that creates a material conflict of interest or places you in a posture materially adverse to the best interests of the Company; or (E) any breach by you of any other term or condition of this Agreement which remains uncured for a period of ten (10) days following your receipt of notification from the Company of such breach.

(ii) "**Disability**" means you becomes physically or mentally incapacitated so as to be unable to perform your usual duties for the Company for a period of more than thirty (30) consecutive days or for a period of sixty (60) days during any consecutive three hundred sixty-five (365) day period, as determined by the Board, and in accordance with the Americans with Disabilities Act, as applicable.

(iii) "Good Reason" means any one of the following without your prior written approval: (A) the assignment to you of duties and responsibilities that are materially inconsistent with your position with the Company; (B) a material diminution in your salary (other than a reduction which is part of a general salary reduction program affecting executives of the Company); (C) a material breach of this Agreement by the Company; or (D) relocation of your principal work location to a location outside the greater Los Angeles area; provided, however, that in each case "Good Reason" shall not exist unless: (i) you give the Company detailed written notice within ninety (90) days following the date when you first learn of the conduct or condition alleged to constitute Good Reason, setting forth the circumstances of such conduct or condition; (ii) the Company has thirty (30) days to cure such conduct or condition giving rise to Good Reason (the "Cure Period"); (iii) the Company fails to cure by the end of the Cure Period; and (iv) you resign for Good Reason within ten (10) days following the end of the Cure Period.

- (c) Termination by the Company for Cause or Resignation by You without Good Reason
- (i) If the Company terminates this Agreement and your employment hereunder for Cause or you terminate this Agreement and your employment hereunder without Good Reason, you shall be entitled to receive only the following compensation (collectively, the "Accrued Rights"):
 - (A) the Base Salary through the date of termination;
- (B) payment in respect of PTO (as described in Section 4) accrued, but not used, during the year in which you are terminated;
- (C) such employee benefits, if any, as to which you may be entitled under the employee benefit plans of the Company (as described in <u>Section 3.2</u>); and
- (D) such reimbursable business expenses as may be due and owing to you under Section $\underline{6}$.
- (d) Termination as a Result of Death or Disability. If this Agreement and your employment hereunder are terminated because of your death or Disability, you or your estate, as the case may be, shall receive as compensation the Accrued Rights.

(e) Termination by the Comp Reason.	pany without Cause or by You for Good		
(i) If the Company employment hereunder without Cause (other than by rea terminate this Agreement and your employment hereu only the following compensation:	· · · · · · · · · · · · · · · · · · ·		
(A) the Accrue	ed Rights; and		
(B) a payment	of a severance benefit, equal to:		
(I) one have earned, but for the termination, in respect of the rettime after December [], 2020, <u>PLUS</u>	e half (1/2) of the Base Salary you would maining Term if you are terminated at any		
Salary you would have earned, but for the termin December [], 2019 and December [], 2020, PLUS	venty-five percent (75%) of the Base nation, in respect of the time between		
(III) the for the termination, in respect of the time between January	Base Salary you would have earned, but ary [], 2019 and December [], 2019;		
which payment shall be made as salary continuation over the remaining Term in accordance with the Company's payroll practices, with such payments beginning within sixty (60) days following the termination of your employment with the Company; provided that you must satisfy the Release Condition (defined below) (the "Severance Benefit"). If you do not satisfy the Release Condition, the Company shall have no obligation to pay you the Severance Benefit.			
mitigate damages and/or the amount of any payment pro other employment or otherwise after the termination of any amounts earned by you, whether from self-emplotherwise, shall reduce the amount of any payments oth limited to, the Severance Benefit. In the event that you employed (including self-employed) or serve as a constryou shall promptly notify the Company in writing and (including details of your compensation) as it shall reason the terms of this provision, you agree to repay to the Severance Benefit received by you.	your employment with the Company, and loyment, as a common-law employee or herwise payable to you, including, but not accept employment or otherwise become altant or service provider during the Term, d provide it with such other information onably request. If you fail to comply with		
(g) General Release. You shat of claims in favor of the Company and its affiliate substantially the form attached hereto as <u>Exhibit B</u> by the release must become irrevocable by its terms in order to	the date specified in such release and such		

that in the event that the period you have to sign and/or revoke such release spans two calendar years, the Company will begin paying you the Severance Benefit as soon as possible but in no

event earlier than the beginning of such second calendar year (the "Release Condition").

- **6.** <u>Business Expenses</u>. The Company will reimburse you for reasonably-incurred business expenses in accordance with Company policies and procedures.
- **7.** Restrictive Covenants. While you are employed by the Company or any affiliate and for a period of twenty-four (24) months following the termination of your employment by the Company or any affiliate for any reason, regardless of whether the termination is initiated by the Company, the affiliate, or you, you will not without written consent from the Company directly or indirectly, on behalf of yourself or on behalf or any other person, firm, company, partnership or business entity:
 - (i) induce, solicit, hire, recruit or attempt to persuade any Person to terminate such Person's employment or other relationship with Pollen Gear LLC, the Company or any of their affiliates (collectively, "Company Parties") or not to establish an employment or other relationship with any Company Party, whether or not such Person is or would be an employee, consultant, contractor, manager, director, officer and/or employee, whether or not such relationship is or would be pursuant to a written or oral agreement and whether or not such relationship is for a specific period of time or is at-will; or
 - (ii) solicit business from any Client or Prospective Client, or do business with any Client or Prospective Client, involving the Business or any business that is competitive, directly or indirectly, with the Business.
 - (iii) For purposes of this <u>Section 7</u>, any capitalized terms used this Section which are not herein defined shall have the meanings ascribed to them in the Contribution Agreement dated as of January [__], 2019 by and among the Company, Pollen Gear Holdings LLC and Pollen Gear LLC (the "Contribution Agreement").
 - (iv) "Client" means a Person for whom or which any Company Party performed services or to whom or which any Company Party sold or licensed its products, during the Term. "Prospective Client" means Persons whose business was solicited by any Company Party during the Term. "Business" means the sale and distribution of vaporizers, cannabis packaging and supplies, or related goods to members of the public.
- 8. <u>Injunction for Breach</u>. In the event of a breach or threatened breach by you of any of the promises contained in this Agreement, you acknowledge that the Company's remedy at law for damages will be inadequate and that the Company will be entitled to an injunction, without the requirement of an injunction bond, to prevent and enjoin you from breaching or threatening to breach any of the provisions contained herein. The Company's application to a court of law for an injunction under this <u>Section 8</u> will not constitute a waiver by the Company or in any way limit or prohibit the Company from pursuing any other remedies available to it in law or equity. You further agree that:
 - (i) The obligations of <u>Section 7</u> of this Agreement is independent of any other obligations of the Company, and the existence of any claim or cause of action of yours against the Company, whether predicated upon this Agreement or otherwise,

- shall not constitute a defense to the enforcement by the Company of the provisions of Section 7;
- (ii) The restrictive covenant provisions of this Agreement are reasonable and necessary to protect the Company's legitimate business interests, including but not limited to the protection of the Company's trade secrets and confidential information, relationships with existing and prospective customers and other business relationships, goodwill, and the Company's investment in training you;
- (iii) If any portion of the restrictive covenants of this Agreement are held be a court to be unreasonable, arbitrary, or against public policy for any reason, the restrictive covenant shall be enforced as to a lesser scope, time or geographic area, to the fullest extent possible such that the provision shall not be considered unreasonable or unenforceable;
- (iv) If proceedings have to be brought by the Company against you to enforce the restrictive covenant provisions of this Agreement, the period of restriction shall be deemed to begin to run on the date of entry of an order granting the Company injunctive relief, and shall continue uninterrupted for the next succeeding twenty-four (24) months; and
- (v) You agree to forfeit all remaining unpaid portions of the Severance Benefit and to repay to the Company immediately all portions of the Severance Benefit received by you.
- 9. Employee Proprietary Rights and Confidentiality Agreement. As an inducement to the Company to hire you as an employee, you agree to be subject to the covenants and other provisions of the Employee Proprietary Rights and Confidentiality Agreement (the "NDA"), annexed hereto as Exhibit A. You agree that your employment with the Company is contingent upon your adherence to the NDA and to the Company's policies and procedures. In the event that your employment with the Company terminates for any reason or no reason, you agree that you will continue to be bound by the provisions of the NDA which by its terms continue in full force and effect after the termination of your employment with the Company. The restrictions in the NDA are in addition to and not in lieu of any other obligations you may be subject to, including, but not limited to, the restrictions Section 7.09 of the Contribution Agreement (the "Non-Competition Restrictions") which are specifically incorporated herein by reference.
- 10. Return of Materials. Upon termination of your employment with the Company, or at any time the Company so requests, (a) you shall return immediately to the Company all Company property and all materials (in written, electronic, or other form) containing or constituting confidential information or related to work, including any copies, reproductions, or other images, and (b) you shall not use confidential information in any way for any purpose.
- 11. <u>Section 409A</u>. This Agreement is intended to comply with the requirements of Section 409A of the Code ("Section 409A"), and the Parties hereby agree to amend this Agreement as and when necessary or desirable to conform to or otherwise properly reflect any guidance issued under Section 409A after the date hereof without violating Section 409A. In case any one or more

provisions of this Agreement fails to comply with the provisions of Section 409A, the remaining provisions of this Agreement shall remain in effect, and this Agreement shall be administered and applied as if the non-complying provisions were not part of this Agreement. The Parties in that event shall endeavor to agree upon a reasonable substitute for the non-complying provisions, to the extent that a substituted provision would not cause this Agreement to fail to comply with Section 409A, and, upon so agreeing, shall incorporate such substituted provisions into this Agreement. In no event whatsoever shall the Company be liable for any additional tax, interest or penalty that may be imposed on you by Section 409A or damages for failing to comply with Section 409A. A termination of your employment hereunder shall not be deemed to have occurred for purposes of any provision of this Agreement providing for the payment of any amount or benefit constituting "deferred compensation" under Section 409A upon or following a termination of employment unless such termination is also a "separation from service" within the meaning of Section 409A and, for purposes of any such provision of this Agreement, references to a "termination," "termination of employment" or like terms shall mean "separation from service." In the event that any payment or benefit made hereunder or under any compensation plan, program or arrangement of the Company would constitute payments or benefits pursuant to a non-qualified deferred compensation plan within the meaning of Section 409A and, at the time of your "separation from service" you are a "specified employee" within the meaning of Section 409A, then any such payments or benefits shall be delayed until the six-month anniversary of the date of your "separation from service." Each payment made under this Agreement shall be designated as a "separate payment" within the meaning of Section 409A. All reimbursements and in-kind benefits provided under this Agreement shall be made or provided in accordance with the requirements of Section 409A to the extent that such reimbursements or in-kind benefits are subject to Section 409A. All reimbursements for expenses paid pursuant hereto that constitute taxable income to you shall in no event be paid later than the end of the calendar year next following the calendar year in which you incur such expense or pays such related tax. Unless otherwise permitted by Section 409A, the right to reimbursement or in-kind benefits under this Agreement shall not be subject to liquidation or exchange for another benefit and the amount of expenses eligible for reimbursement, or in-kind benefits, provided during any taxable year shall not affect the expenses eligible for reimbursement, or in-kind benefits to be provided, respectively, in any other taxable year.

12. No Other Restrictions. You hereby represent that you are under no restrictions that would prevent you from being employed by the Company or performing the Services contemplated by this Agreement. You hereby represent and warrant that: (i) entering into employment with the Company under this Agreement does not constitute a breach of any agreement or other legal obligation with or to a third party; (ii) you are not bound by or subject to any agreement or other legal obligation with a third party that would adversely affect your employment with the Company, including, but not limited to, a prior employment agreement, confidentiality agreement, or covenant not to compete, not to solicit or other restriction against competition; (iii) you will not use in connection with your employment by the Company any confidential or proprietary information belonging to any third party without first obtaining a written release from that third party; (iv) all work that you shall provide to the Company shall be your original work product and shall not knowingly infringe upon the intellectual property rights of any third party; and (v) in performing services for the Company, you shall comply with all policies and procedures of the Company and all applicable laws.

- 13. <u>Background Check</u>. You agree to assist and cooperate with the Company on any background checks the Company wishes to perform on you, including any credit reports, and you will execute all appropriate authorizations to obtain any such background checks and/or reports. This offer of employment is conditioned upon a favorable outcome of such background checks.
- Governing Law and Arbitration. This Agreement shall be governed by, and construed and enforced in accordance with, the laws of the State of California. Other than a claimed violation by you of the separately executed NDA or the Non-Competition Restrictions (the "Injunctive Relief Exception"), any dispute or controversy arising out of or relating to this, and/or related to your employment, that could otherwise be resolved by a court shall be resolved through arbitration in accordance with the Employment Arbitration Rules and Mediation Procedures of the American Arbitration Association, in which neither class nor collective proceedings will be permitted. Judgment upon the award may be entered in any court having jurisdiction thereover. Except for the Injunctive Relief Exception, you and the Company give up and waive any right to resolve a controversy through any other means, and the right to sue in court in connection with claims related to this Agreement and/or your employment with the Company. This waiver of the right to sue in court includes, for example, claims based on federal statutes such as Title VII of the Civil Rights Act of 1964 and the Fair Labor Standards Act and claims based on statutes, common law causes of action and concerning compensation. If there is more than one dispute between you and the Company, all such disputes may be heard in a single proceeding. Disputes pertaining to different employees of the Company will be heard in separate proceedings. Any arbitration shall be held in the County of Los Angeles, State of California or in such other place as the Parties hereto may agree, unless applicable law requires otherwise.
- 15. <u>No Credit for Drafting</u>. You understand that this Agreement is deemed to have been drafted jointly by the Parties. Any uncertainty or ambiguity shall not be construed for or against any Party based on attribution of drafting to any Party.
- **16.** <u>Survival</u>. You agree that notwithstanding anything in this Agreement to the contrary, your obligations under this Agreement, specifically but without limitation, your obligations under Sections 5 and 7 above, the Non-Competition Restrictions, and the NDA, shall survive any termination of this Agreement and/or the relationship between you and the Company, to the extent necessary to give effect to the subject provisions according to their terms.
- 17. <u>Assignment</u>. This Agreement shall be binding upon and inure to the benefit of your heirs and representatives and the assigns and successors of the Company, but neither this Agreement nor any rights or obligations hereunder shall be assignable or otherwise subject to hypothecation by you. The Company shall have the right to assign this Agreement to any of its affiliates, successors, or related companies, whether now in existence or later formed.
- 18. <u>Severability</u>. Whenever possible, each provision of this Agreement shall be interpreted in such a manner as to be effective and valid under applicable law, but if any provision of this Agreement is held to be prohibited by or invalid under applicable law, such provision shall fail to be in effect only to the extent of such prohibition or invalidity, without invalidating the remainder of this Agreement or of any such provision.

- **19.** <u>Amendment and Modification</u>. No provision of this Agreement may be modified, amended, waived or terminated except by a writing signed by both you and the Company. No course of dealing between you and the Company will modify, amend, waive or terminate any provision of this Agreement or any rights or obligations of any Party under or by reason of this Agreement.
- **20.** Entire Agreement. This Agreement constitutes the entire Agreement between the Parties hereto relating to the subject matter hereof. All prior or contemporaneous agreements or understandings between the Parties relating to the subject matter hereof, whether oral or written, are superseded by and merged into this Agreement. You acknowledge that no other promises were made to you other than are or may be contained in this Agreement, and that no other inducement caused you to sign this letter.
- **21.** Counterparts. This Agreement may be executed in two or more counterparts, each of which shall be an original, but all of which shall constitute one and the same instrument.
- **22.** <u>Facsimile Signatures</u>. A facsimile or PDF scanned copy of signature shall be as valid and binding as an original.
- **23.** Acknowledgment. You acknowledge that you have had the opportunity to consult legal counsel in regard to this Agreement, have read and understand this Agreement, are fully aware of its legal effect, and have entered into it freely and voluntarily and not based on any representations of promises other than those contained in this Agreement.

[Signature page follows]

IN WITNESS WHEREOF, the Parties hereto have caused this Agreement to be duly executed.

Date:
By: Greenlane Holdings LLC By: Jacoby & Co. Inc. Its: Managing Member
By:
Aaron LoCascio
Co-President By:
Adam Schoenfeld
Co-President
Accepted and Agreed:
Edward Kilduff
Data

EXHIBIT A

EMPLOYEE PROPRIETARY RIGHTS AND CONFIDENTIALITY AGREEMENT

This Employee Proprietary Rights and Confidentiality Agreement, made the ____ day of December, 2018, is between Greenlane Holdings, LLC ("Company") and Edward Kilduff ("Employee").

WHEREAS, subject to Employee's agreement to, and compliance with, this Agreement Company has agreed to employee, and Employee has agreed to be employed by Company; and

WHEREAS, Company and Employee each acknowledge and agree that, as part of Employee's job duties, Employee will from time-to-time have access to Confidential Information as more specifically defined below and may from time-to-time conceive of, author, or otherwise create or assist in the conception, authorship or development of certain Inventions and Works of Authorship as more specifically defined below;

NOW THEREFORE, for good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, Company and Employee agree as follows:

1. <u>Intellectual Property</u>.

- a. Ownership and Works Made For Hire. Employee hereby agrees that Company shall be the sole and exclusive owner of any and all ideas, inventions, discoveries, computer hardware and software, algorithms, improvements and devices relating thereto, and other information and know-how heretofore or hereinafter conceived, developed or made by Employee, either alone or with others, during the course of Employee's employment by Company, or which are conceived, developed or made using Company equipment or resources, or by reason of knowledge gained from such employment (hereinafter "Inventions"), together with any graphic designs, narrative works, business plans, financial analyses, lab notes, designs, specifications, programming, source code, object code, flow charts, data, compilations of data and other information and works of authorship of any nature compiled, authored or otherwise developed by Employee and relating to such Inventions or otherwise created during the course of Employee's employment with Company or using Company equipment or resources (the "Works of Authorship") and all trademark, copyright, trade secret, patent and other intellectual and industrial property rights of whatever nature throughout the world covering or embodied by such Inventions and Works of Authorship, that the Works of Authorship constitute "works made for hire" (as that term is defined in the United States Copyright Act as amended) by Employee because they have either been prepared by Employee while acting within the scope of Employee's employment with Company or they have been specifically ordered or commissioned from Employee for use by Company and that, as such, Company is the author and owner of such Works of Authorship from the first instance of creation.
- b. <u>Assignment</u>. Employee hereby acknowledges and agrees that, to the extent any Inventions or Works of Authorship are not, for whatever reason, deemed to be "works made for hire" or otherwise solely and exclusively owned by Company, Employee hereby

grants, transfers, conveys and assigns to Company and its successors and assigns all of Employee's right, title, interest, ownership and all subsidiary rights in and to such Inventions and Works of Authorship and all trademark, trade secret, copyright, patent and other intellectual and industrial property rights of whatever nature throughout the world covering or embodied by such Inventions and Works of Authorship, including the right to apply for, register, obtain, maintain, enforce and defend the foregoing in any jurisdiction throughout the world.

- c. <u>Disclosure</u>. Employee hereby agrees to keep and maintain current and complete records of, and to disclose promptly and in writing to Company, all Inventions and Works of Authorship to which Company is or may arguably be entitled as provided above, and agrees not to (i) disclose such Inventions and Works of Authorship to others, except as required in accordance with Employee's duties as an employee of Company, or (ii) move copies of such written records to, or keep copies of such records at, locations other than the Company's premises, in either case without the express prior written consent of Company.
- d. Continuing Assistance. Employee hereby further agrees that during the term of Employee's employment with Company and at any time thereafter, Employee shall execute and deliver to Company any written documents necessary or appropriate, in Company's sole discretion, to effectuate the assignment to Company of any and all rights in such Inventions and Works of Authorship to which Company is entitled as provided above, and will execute all papers and perform any other lawful acts requested by Company for the preparation, prosecution, procurement, maintenance, enforcement and defense of any trademark, trade secret, copyright, patent and/or other intellectual or industrial property rights in and for such Inventions and Works of Authorship throughout the world, and will execute all documents and perform any other lawful act necessary to vest title in Company to such Inventions and Works of Authorship and all trademarks, trade secrets, copyrights, patents and other intellectual and industrial property rights relating thereto. To the extent that, for any reason or no reason, including without limitation Employee's physical or mental incapacity, Employee fails to comply with Employee's obligations described above in this Section 1(d), Employee hereby appoints Company, acting by and through Company's designated officers or representatives, as Employee's agent and attorney in fact, and grants to Company a power-of-attorney with full power of substitution, to execute, deliver and file any documents, and perform any acts, Employee would otherwise lawfully be obligated to execute, deliver, file or perform pursuant to this Section 1(d), with the same legal force and effect as if executed, delivered, filed or performed by Employee. Such power of attorney shall be deemed coupled with an interest for the purposes of effecting the foregoing assignments, prosecutions, procurement, maintenance, enforcement and defense.
- e. <u>Pre-existing Rights</u>. In the event that Employee uses or incorporates into any Inventions, Works of Authorship or other work product of Employee created in the course of Employee's employment with Company any (i) ideas, inventions, discoveries, computer hardware and software, algorithms, improvements and devices relating thereto conceived, developed or made by Employee prior to, or outside the scope of, Employee's employment with Company, or (ii) any graphic designs, narrative works, business plans, financial

analyses, lab notes, designs, specifications, programming, source code, object code, flow charts, data, compilations of data and other information compiled, authored, developed or otherwise created by Employee prior to, or outside the scope of, Employee's employment with Company (collectively, the "Pre-existing Inventions and Works"), then Employee shall be deemed to have granted to Company a perpetual, irrevocable, fully-paid, royalty-free, non-exclusive, transferable, assignable, sublicenseable right, under any and all trademarks, trade secrets, copyrights, patents and other intellectual and industrial property rights of whatever nature throughout the world covering or embodied by the Pre-existing Inventions and Works, to make, have made, use, copy, distribute, display, perform, modify, translate, offer to sell, sell, and otherwise transfer or dispose of the Pre-existing Inventions and Works and any products and services embodying or covered by such trademarks, trade secrets, copyrights, patents and other intellectual and industrial property rights. Employee covenants and agrees that it shall notify Company in writing of all Pre-existing Inventions and Works prior to incorporating the same into any Inventions, Works of Authorship or other work product created in the course of Employee's employment with the Company.

f. Moral Rights Waiver. Employee hereby releases, waives and to the extent allowed by law assigns in favor of Company any and all "moral" rights related to the Works of Authorship, including without limitation: the right of paternity (i.e., acknowledgment of authorship), disclosure, withdrawal and integrity (i.e., right to prevent changes to the work).

2. Confidentiality and Non-Disparagement.

- For so long as Employee is employed by Company and thereafter a. Non-Disclosure. for so long as the Confidential Information (as hereinafter defined) has not otherwise been publicly disclosed by Company, Employee hereby expressly agrees that (i) Employee shall not divulge, directly or indirectly, other than in the regular and proper course of business of Company, any Confidential Information, and (ii) Employee shall not use, directly or indirectly, any Confidential Information for the benefit of anyone other than Company; provided, however, that Employee shall have no obligation, express or implied, to refrain from using or disclosing to others any Confidential Information which is or hereafter shall become generally available to the public other than through Employee's breach of this Agreement. In the event Employee is or may foreseeably be compelled to disclose any Confidential Information pursuant to any law, regulation, order of a court of competent jurisdiction or other legal process, Employee shall give Company prior written notice, reasonable in light of the circumstances, of such compelled disclosure. In the event Employee makes or learns of any unauthorized disclosure of Confidential Information, Employee shall promptly notify Company of such unauthorized disclosure and in good faith cooperate with Company's efforts to ameliorate the effects of, or otherwise enforce its rights in connection with, such unauthorized disclosure.
- b. <u>Confidential Information</u>. For the purposes of this Agreement, the term "**Confidential Information**" shall mean any information not generally known in the relevant trade or industry (i.e. any trade or industry in which Company engages or is then engaged) or otherwise not generally available to the public, which was obtained from or disclosed to Employee by Company, its affiliates or their predecessors or which was learned,

discovered, developed, conceived, originated or prepared by Employee in the course of Employee's employment with Company, including, without limitation, Inventions, Works of Authorship, and any other product, service, customer, marketing, operating and financial information, computer software, source code, object codes, and flow charts proprietary to Company.

- c. <u>Possession</u>. All Confidential Information in Employee's possession at any time shall be the exclusive property of Company, shall not be copied, summarized, extracted from, or removed from the premises of Company, except in pursuit of the business of Company and at the direction of Company, and shall be delivered to Company, without Employee retaining any copies, upon the termination of Employee's employment with Company or at any earlier time as requested by Company.
- Employee warrants that the performance of its duties and d. Prior Obligations. obligations under this Agreement will not constitute a breach or default under any other agreement to which Employee is bound, that it is not bound by any pre-existing (i.e., before the date of this Agreement first set forth above) obligation of confidentiality or duty to assign rights in any Inventions or Works of Authorship (whether written, oral or otherwise) to any other person or entity that may foreseeably interfere with or conflict with Employee's job duties or compliance with this Agreement, nor is there any pending or threatened proceeding, investigation or action alleging (or to Employee's knowledge any basis for such a proceeding, investigation or action) that Employee is bound by any such obligation or duty. Employee covenants and agrees that, for so long as Employee is employed by Company, Employee shall (i) comply with any pre-existing confidentiality obligations, (ii) in the event Employee's job duties with Company may in the future potentially conflict with such obligations, inform Company of such conflict and refrain from making any disclosures or otherwise taking any action that violate Employee's preexisting confidentiality obligations, (iii) at all times in good faith work with Company to determine a course of action which will allow Employee to perform Employee' job duties without violating any pre-existing confidentiality obligations, and (iv) not hereafter enter into any agreement with any other person or entity that imposes upon Employee obligations of confidentiality that conflict with, or may foreseeably conflict with, the proper performance of Employee's job duties with Company. Employee hereby consents to Company notifying any subsequent employers of Employee that Employee is bound by obligations of confidentiality as set forth in this Agreement.
- e. <u>Defend Trade Secrets Act Notice</u>. Company hereby notifies Employee, and Employee hereby acknowledges, that nothing in this Agreement is intended to interfere with or restrain the immunity provided under the Defend Trade Secrets Act (18 U.S.C. Section 1833(b)). Specifically, an individual shall not be held criminally or civilly liable under any Federal or State trade secret law for the disclosure of a trade secret that is made in a complaint or other document filed in a lawsuit or other proceeding if such filing is made under seal. An individual who files a lawsuit for retaliation by an employer for reporting a suspected violation of law may disclose the trade secret to the attorney of the individual, and may use the trade secret information in the court proceeding, if the individual files any

document containing the trade secret under seal and does not disclose the trade secret except pursuant to court order.

f. Non-disparagement. Employee covenants and agrees that, for so long as Employee is employed by Company and for one year thereafter, Employee shall not make, publish or otherwise disseminate disparaging or derogatory statements or allegations, whether written, oral or otherwise, regarding Company or its affiliates, or any officers, directors, employees, products, services, policies or operations of Company or its affiliates. The provisions of this paragraph 2(f) in no way restrict or prohibit Employee from providing truthful testimony when such testimony is lawfully compelled pursuant to any court order or other legal process.

3. No Further Consideration.

Employee acknowledges and agrees that, except as compensated in accordance with Employee's status as an employee of Company, Employee shall not be entitled to any further or additional compensation in consideration of (i) the assignment by Employee of Employee's rights in any Inventions and/or Works of Authorship as provided for above in paragraphs 1 (a) and (b), (ii) the licenses (if any) granted to Company as provided for above in paragraph 1(e), (iii) complying with the confidentiality and non-disparagement obligations set forth above in paragraph 2, and (iv) the provision of any services as provided for above in paragraph 1(d); provided, however, that Employee shall be reimbursed for actual out-of-pocket expenses incurred in rendering to Company the services specified in paragraph 1(d).

- 4. <u>Term.</u> This Agreement shall commence and be effective as of the date of this Agreement first set forth above and shall terminate on the date Employee ceases to be an employee of Company. Notwithstanding the above, Employee's obligations arising pursuant to paragraphs 1(c) (d) and (e) and 2 (a), (c), (e) and (f) above, and paragraphs 5 and 6 (c), (d), (e) and (f) below, shall survive termination of this Agreement. In addition, those rights, duties and obligations of Company and Employee set forth in this Agreement that, by their nature, are intended to survive the termination of this Agreement shall survive the termination of this Agreement.
- 5. Originality. Employee hereby warrants and represents to Company that, to the best of Employee's knowledge after reasonable inquiry and except as expressly disclosed to Company in writing by Employee, all Inventions, Works of Authorship and Pre-existing Inventions and Works shall be original and shall not have been conceived, reduced to practice, authored or otherwise developed by misappropriating the trade secrets or other proprietary technology or intellectual property of any third parties. The foregoing representation and warranty shall be deemed first made on the date hereof and shall run continuously thereafter and apply to all Inventions, Works of Authorship and Pre-existing Inventions and Works governed by this Agreement.

6. Miscellaneous.

a. <u>Assignability</u>. Employee shall not assign this Agreement without the express prior written consent of Company, which consent may be withheld for any reason or no reason at Company's sole discretion. Any assignment in violation of this provision shall be void *ab*

initio and of no force or effect. Except as otherwise provided herein, this Agreement shall be binding upon the respective heirs, successors, and permitted assigns of the parties hereto. Company shall be free to assign its rights hereunder to any other party without the consent of Employee.

- b. <u>Severability</u>. Any provision of this Agreement which is determined to be illegal, prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such illegality, prohibition or unenforceability without invalidating the remaining provisions hereof which shall be severable and enforceable according to their terms and any such prohibition or unenforceability in any jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction.
- c. Specific Performance and Equitable Relief. Employee acknowledges and agrees that Company would suffer irreparable harm if Company's rights in the Inventions and Works of Authorship were not protected and/or the Confidential Information was disclosed to third parties without Company's consent. Accordingly, Employee hereby consents to and agrees that, in the event of a breach or threatened breach of this Agreement by Employee, Company shall have the right to exercise any and all rights by appropriate action either by law or in equity, including specific performance of Employee's obligations arising pursuant to paragraphs 1(c) and (d) above, and that Company shall be entitled to equitable relief, including injunctive relief, in connection with any breach or threatened breach, by Employee, of Employee's obligations arising pursuant to paragraphs 2(a), (c), (d), (e) and (f) above. Employee further agrees that Employee shall not request that Company post, nor shall Company be obligated to post, a bond in connection with any equitable relief authorized pursuant to this paragraph 6(c) and in fact requested by Company.
- d. <u>No Expectation of Privacy</u>. Employee acknowledges and agrees that Employee has no expectation of privacy in respect of any information (e.g., emails and voicemails), whether personal or otherwise, stored on Company's networking, telecommunications, computer and other equipment, which information may be monitored by Company at any time without notice.
- e. Governing Law. This Agreement shall be governed by, and construed and enforced in accordance with, the laws of the State of California and, to the extent applicable, the laws and regulations of the United States of America governing intellectual property matters, without giving effect to the provisions, policies or principles under California law respecting conflict or choice of law. EMPLOYEE ACKNOWLEDGES AND AGREES THAT, PRIOR TO EXECUTING THIS AGREEMENT, EMPLOYEE WAS AFFORDED AN OPPORTUNITY TO OBTAIN THE ADVICE OF INDEPENDENT LEGAL COUNSEL, AND HAS READ AND UNDERSTOOD ALL OF THIS AGREEMENT. ACCORDINGLY, THIS AGREEMENT SHALL NOT BE CONSTRUED AGAINST ANY PARTY BY REASON OF THE DRAFTING OR PREPARATION OF THIS AGREEMENT.
- f. <u>Choice of Forum</u>. Each of the parties hereby consents to the jurisdiction of any state or federal court located within the County of Los Angeles, State of California, and irrevocably

agrees that all actions or proceedings relating to this Agreement must be litigated in such courts, and each of the parties waives any objections which it may have based on improper venue or forum non conveniens to the conduct of any proceeding in such court.

- g. <u>Entire Agreement</u>. This Agreement, together with the written employment agreement (if any) to which the Company and Employee are a party, contains the entire understanding of the parties with respect to its subject matter, and supersedes all prior written or oral agreements and understandings between the parties with respect to its subject matter. This Agreement may not be changed orally, and any written change or amendment must be signed and accepted by Company.
- h. <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, and by the different parties hereto in separate counterparts, each of which shall be deemed to be an original but all of which taken together shall constitute one and the same agreement (and all signatures need not appear on any one counterpart), and this Agreement shall become effective when one or more counterparts has been signed by each of the parties hereto and delivered to each of the other parties hereto.

[Remainder of page intentionally left blank; signature page follows]

IN WITNESS WHEREOF, the parties have executed this Agreement and caused it to be binding and effective upon each of them in accordance with its terms.

Greenlane Holdings, LLC	Employee	
By: Jacoby & Co. Inc.		
Its: Managing Member		
By:		
Name: Aaron LoCascio	Edward Kilduff	
Title: Co-President		
By:		
Name: Adam Schoenfeld		
Title: Co-President		

EXHIBIT C-2

Brown Employment Agreement

(See attached.)

C-2

EMPLOYMENT AGREEMENT

This Employment Agreement ("Agreement") is entered into by and between Greenlane Holdings, LLC a Delaware Limited Liability Company and subsidiaries, hereinafter referred to as "Company" or "Employer" and Jason Brown, hereinafter referred to as "Employee," on the 14th day of January, 2019.

WITNESSETH:

WHEREAS, the Company and the Employee mutually desire to enter into an Agreement whereby the Employee agrees to provide certain services to the Company within accepted standards of care and under the terms and conditions set forth herein.

NOW, THEREFORE, in consideration of the mutual covenants and promises contained herein, and other good and valuable consideration, the receipt and adequacy of which is hereby acknowledged, the parties hereto, intending to be legally bound, hereby agree as follows:

- 1 *Recitations*. The recitations are true and correct and are incorporated herein by reference.
- 2 Definitions. Whenever used in this Agreement, the following terms shall have the definitions contained in this section.

Effective Date: means the last date specified on the signature portion of this Agreement.

Business: means the wholesale and distribution of vaporizers, cannabis packaging and cannabis accessories to members of the public, and any other lines of business the Company engages in during the term of Employee's employment with the Company.

- 3 Employment. The Company hereby employs the Employee as, and in the specific capacity of *Vice President, Strategic Growth* and the Employee accepts said employment upon the terms and conditions of this Agreement.
- 4 Employee's Duties and Responsibilities.
 - 4.1 The Employee shall be responsible for, but not limited to task and duties relating to the position of *Vice President, Strategic Growth*.
 - 4.2 The Employee shall at all times conduct himself or herself in compliance with all applicable federal, state, and local laws and their rules and regulations and in compliance with the Company's policies and procedures, and the terms of this Agreement.
 - 4.3 The Employee will occasionally be asked to work additional hours, including but not limited to working trade shows, weekends, and mandatory meetings.

Page 1

- 5 Employer's Responsibilities.
 - 5.1 <u>Administration</u>. The Company shall perform the appropriate administrative, regulatory, marketing, enrollment and other functions necessary for administration of the company, pursuant to applicable state and federal law. The Employer shall make all employment, termination, supervision, discipline and promotion decisions concerning its employees, and will assign employees, at its discretion, to assist the Employee with the Employee's performance hereunder. The Employer shall provide the Employee such office space, facilities, and supplies as reasonably necessary, in its discretion, to permit the Employee to provide the services described herein. The Employee shall abide by policies and procedures established by the Company.
 - 5.2 <u>Accounting</u>. The Company shall maintain, in accordance with generally accepted accounting principles, such financial accounting records as shall be necessary, appropriate and convenient to carry out the purposes of this Agreement. The Company may contract with a third party to carry out any or all of these functions.
 - 5.3 <u>Information</u>. The Company shall provide the Employee with information and data reasonably necessary to carry out the terms and conditions of this Agreement.
- 6 Compensation & Benefits. The Company shall provide compensation and benefits to the Employee for the continual and complete discharge of his or her duties under this Agreement as specified in attached Schedule 1.
- 7 Relocation Expense Reimbursements. The Company shall provide relocation expense reimbursements under this agreement as specified in attached Schedule 1, if applicable.
- 8 Term of Agreement/Termination.
 - 8.1 <u>Term</u>. The term of this Agreement shall commence on the Effective Date and shall continue for the period of one year thereafter. This Agreement shall automatically renew for successive one year periods, unless either party notifies the other of its desire not to renew the Agreement at least 60 days prior to the next scheduled renewal date. This Agreement may also be terminated in accordance with any applicable provision of this Agreement.
 - 8.2 <u>Termination Without Cause</u>. Either party may terminate this Agreement at any time without cause.
 - 8.3 <u>Employer Right To Terminate For Cause</u>. The Employer may terminate the Agreement for cause immediately in the event of the occurrence of any of the following events:
 - 8.3.1 Employee violates any applicable law;

- 8.3.2 Employee's commission of any act or omission that, in the Company's sole judgment, subjects the Company to shame or ridicule, or otherwise materially diminishes the reputation and good name of the Company;
- 8.3.3 Employee is in violation of or engages in conduct which in the Company's reasonably exercised judgment, is inconsistent with the Employer's policies, and procedures, business or professional interests:
- 8.3.4 Employee commits any act or omission that, in the Company's reasonably exercised judgment creates a conflict of interest or places the Employee in a posture adverse to the best interests of the Company;
- 8.3.5 Employee becomes unable to perform the Employee's duties and obligations pursuant to this Agreement, with reasonable accommodation, for a continuous period of more than thirty (30) days or for period of sixty (60) days during any consecutive three hundred sixtyfive (365) day period;
 - 8.3.6 Death of the Employee; and
- 8.3.7 A breach by the Employee of any other term or condition of this Agreement which remains uncured for a period of ten (10) days following receipt of notification from the Company of such breach.
- Obligations upon Termination. Upon termination of this Agreement, neither party shall have any further obligation hereunder except for obligations accruing prior to the date of termination, and those obligations, promises or covenants contained herein which are expressly made to extend beyond the term of the Agreement.
- Employee's Cooperation Upon Termination. The Employee agrees that for the period of his employment and thereafter for a period of twentyfour (24) months, the Company has the right to receive, open, and / or intercept all mail, and other communications (whether written, or in electronic or other form) addressed to the Employee at the Company's address or received by the Company by means of any electronic media, and the right to take possession, for the Company's own account, any mail or other communication provided such communication relates to the Company's business, and the Company shall have the further right to endorse the Employee's name on financial instruments "for deposit" and to deposit for the account of the Company any check, money order or other financial instrument payable to the Employee that is received from any person as payment for product or professional or other services rendered by the Company during the term of this Agreement.

9 Acknowledgements. The Company is engaged in the business of providing vaporizers and related equipment to members of the public and retailers and engagesin competition with other businesses that provide herbal vaporizers to members of the public and retailers. Through great time, energy and expense, the Company has developed expertise, programs, systems, methods, practices, and other information as well as contractual and business relationships with vendors and others that enable the Company to compete in the industry. The Employee acknowledges that:

- 9.1 The Company's business is highly specialized;
- 9.2 The expertise, programs, systems, methods, practices, and other information developed, used and maintained by the Company are not generally known by competitors in the industry;
- 9.3 The Company has a proprietary interest in its programs, systems, methods, practices, and other information developed and maintained by the Company;
- 9.4 Documents and other information regarding the Company's business methods, relationships, strategies, costs, programs, systems, methods, practices, and other information developed, used and maintained by the Company are highly confidential and constitute trade secrets; and
- 9.5 Disclosure of the Company's confidential and trade secret information would irreparably harm the Company and give unfair advantage to any other person competing with the Company in the industry.
- 10 Intellectual Property and Confidential Information. In order to induce the Company to hire and to thereafter continue to employ the Employee, the Employee and the Company have entered into a Proprietary Rights and Confidentiality Agreement (the "NDA"), annexed hereto as Exhibit A.
- 11 Restrictive Covenant. During the term of this Agreement, and for a period of eighteen (18) months after this Agreement has been terminated for any reason, regardless of whether the termination is initiated by the Company or the Employee, the Employee will not without written consent from the Company directly or indirectly, on behalf of himself or herself or on behalf of any other person, firm, company, partnership or business entity:
 - 11.1 solicit business in the Business from any person who is or was a customer or supplier of the Company during a period of five (5) years prior to the termination of the Employee's employment;
 - 11.2 recruit or solicit any employee of the Company, or person who was employed by the Company during a period of one (1) year prior to the termination of the Employee's employment, for employment with any other organization which engages directly or indirectly in the business of providing vaporizers and related goods to members of the public;
 - own, manage, operate, control, be employed by, participate in, or be connected in any manner whatsoever with the ownership, management,

operation, or control of any business engaged in the Business provided however that ownership of less than 5% of a company through a passive investment shall be permitted.

- Injunction for Breach. In the event of a breach or threatened breach by the Employee of any of the promises contained in this Agreement, the Employee acknowledges that the Company's remedy at law for damages will be inadequate and that the Company will be entitled to an injunction, without the requirement of an injunction bond, to prevent and enjoin the Employee from breaching or threatening to breach any of the provisions contained herein. The Company's application to a court of law for an injunction under this paragraph will not constitute a waiver by the Company or in any way limit or prohibit the Company from pursuing any other remedies available to it in law or equity. The Employee further agrees that:
 - 12.1 the obligations of paragraphs 10 and 11 of this Agreement are independent of any other obligations of the Company, and the existence of any claim or cause of action of the Employee against the Company, whether predicated upon this Agreement or otherwise, shall not constitute a defense to the enforcement by the Company of the provisions of paragraphs 10 and 11;
 - 12.2 the restrictive covenant provisions of this Agreement are reasonable and necessary to protect the Company's legitimate business interests, including but not limited to the protection of the Company's trade secrets and confidential information, relationships with existing and prospective customers and other business relationships, goodwill, and the Company's investment in training the Employee;
 - 12.3 if any portion of the restrictive covenants of this Agreement are held by a court to be unreasonable, arbitrary, or against public policy for any reason, the restrictive covenant shall be enforced as to a lesser scope, time or geographic area, to the fullest extent possible such that the provision shall not be considered unreasonable or unenforceable;
 - 12.4 if proceedings have to be brought by the Company against the Employee to enforce the restrictive covenant provisions of this Agreement, the period of restriction shall be deemed to begin to run on the date of entry of an order granting the Company injunctive relief, and shall continue uninterrupted for the next succeeding twelve (12) months.
- 13 Employee Proprietary Rights and Confidentiality Agreement. As an inducement to the Company to hire you as an employee, you agree to be subject to the covenants and other provisions of the Employee Proprietary Rights and Confidentiality Agreement. You agree that your employment with the Company is contingent upon your adherence to the NDA and to the Company's policies and procedures. In the event that your employment with the Company terminates for any reason or no reason, you agree that you will continue to be bound by the provisions of the NDA which by its terms continue in full force and effect after the termination of your employment with the Company.
- 14 Dispute Resolution. In the event of any dispute or claim regarding this Agreement or any

part thereof, the parties agree to submit such dispute or claim to binding arbitration. A single arbitrator shall be chosen by mutual agreement between the parties and their respective counsel, who shall render an award within 30 days of the arbitration hearing. The venue for any such arbitration shall be Palm Beach County, Florida. The determination or award rendered therein shall be binding and conclusive upon the parties, and judgment may be entered thereon in accordance with applicable law in any court having jurisdiction over the parties. The prevailing party in any such arbitration proceeding shall recover from the other party his/her/its attorneys fees and costs, which amounts shall be determined by the arbitrator. This paragraph will not be deemed a waiver of the Company's right to injunctive relief as provided for in this Agreement.

- 15 Representation. The Employee hereby represents and warrants to the Company that: (a) this Agreement constitutes the legal, valid and binding obligation to the Employee, enforceable against the Employee in accordance with its terms; (b) the execution, delivery and performance of this Agreement has been duly authorized by all required action of the Employee and (c) such execution, delivery and performance does not violate any provisions of any agreement to which the Employee is a party, or any laws or regulations applicable to the Employee.
- 16 Indemnification. The Employee shall indemnify and hold harmless, to the fullest extent permitted by law, the Company, its officers, directors, and shareholders with regard to any threatened, pending or contemplated claim, action, suit, or other type of proceeding, or any final judgment, assessment, fine, or penalty, including, without limitation: (a) any obligations to repay any sums previously paid, directly or indirectly, to the Company with respect to services rendered by the Employee on behalf of the Company if such overpayment results from or is based on the Employee: (i) the Internal Revenue Service upon audit or examination determine that business deductions to the Employee are disallowed as ineligible business deductions to the Company; (ii) professional or general liability claims or judgments are asserted against the Company with respect to acts or omissions of the Employee which exceed applicable insurance proceeds or which are determined to be non-covered by insurance; and (iii) any breach by the Employee of any of the representations and warranties of the Employee contained herein.
- No Ownership Interest. During the term of this Agreement, the Employee shall not be required to contribute any money towards the Company's equipment or operations, but likewise the Employee will have no financial interest in the accounts receivable, furniture, equipment, leasehold improvements, property, assets, goodwill or the like of the Company except as otherwise agreed in writing between the Employee and the Company. Notwithstanding the following, the Company and the Employee acknowledge (i) the Employee's indirect ownership of equity interests in the Company and (ii) the possibility that Employee shall receive equity from the Company's affiliates in their sole and absolute discretion.

18 Reporting Legal Actions.

18.1 <u>Summons or Complaint</u>. The Employee will notify the Company in writing within five (5) working days of being served with a summons, complaint, written claim, or other document alleging negligence or other injury, whether before or during employment by the Company.

Page 6

- 18.2 <u>Occurrence or Event</u>. The Employee will orally notify the Company within five (5) working days of any event which raises a reasonable probability that it will result in a legal claim or action against the Employee relating to his/her employment duties.
- 18.3 <u>Arrest, Indictment or Conviction</u>. The Employee will orally notify the Company immediately, and in writing within five (5) days, of any arrest, indictment or conviction for other than a misdemeanor motor vehicle license infraction.
- 18.4 Cooperation in Dispute Resolution. The Employee recognizes that certain disputes may arise between the Company and third parties, the resolution of which may require the cooperation of the Employee, including, but not limited to, the Employee's providing factual information and giving depositions and testimony in judicial and administrative proceedings. The Employee shall, during the term hereof and at all times after termination for a period of five (5) years, cooperate with the Company to allow it to advance its position with respect to such disputes. The Company shall provide reasonable compensation to the Employee and reimburse the Employee for all out-of-pocket expenses incurred in connection with such cooperation, provided that the Employee obtains the Company's agreement in advance for such expenses and provides the Company with an itemized written account of such reimbursable expenses. The terms and conditions of this Paragraph shall survive termination of this Agreement.

19 Miscellaneous.

- 19.1 <u>Applicable law</u>. This Agreement, and the rights and obligations of the parties herein, shall be construed, interpreted and enforced in accordance with, and governed by, the laws of the State of Florida, without reference to conflicts of laws principles as applied in that State. Subject to the arbitration provisions of Section 14 above, the parties each consent to the exclusive jurisdiction of any state or federal court located in the County of Palm Beach, State of Florida in all actions or proceedings relating to this Agreement.
- 19.2 <u>Confidentiality</u>. The parties hereto agree that each shall hold this Agreement and the information contained in this Agreement as confidential and shall not disclose this Agreement or any of its terms or information contained in this Agreement to any non-party, except as necessary to carry out the terms herein or as required by state or federal law.
- 19.3 <u>Heading and captions</u>. Headings and captions used in this Agreement are for convenience only and should not be considered in interpreting the provisions hereof.
- 19.4 <u>Federal and state law regulations</u>. This Agreement, and the performance thereof, is subject to the requirements of all applicable state and federal laws. Provisions required thereby to be in this Agreement shall

be incorporated herein by this reference and shall bind the parties to this Agreement whether or not specifically provided herein.

- 19.5 <u>Modification</u>. This Agreement, and the attachments hereto, constitute the entire understanding of the parties hereto and no changes, amendments or alterations shall be effective unless in writing and signed by both parties.
- 19.6 <u>Enforceability</u>. The invalidity or un-enforceability of any terms or conditions hereof shall in no way affect the validity or enforceability of any other term or provision.
- 19.7 <u>Notice</u>. Any notice required to be given under this Agreement shall be in writing and shall be delivered by certified mail, return receipt requested, to the Employee's last known residence or to Greenlane Holdings, LLC, Attention: Douglas Fischer, 1095 Broken Sound Parkways, Boca Raton, FL 33487.
- 19.8 <u>Complete Agreement</u>. This Agreement, together with the Proprietary Rights and Confidentiality Agreement entered into by the parties contemporaneously herewith, constitute the entire understanding of the parties in respect of its subject matter, and supersedes all prior oral agreements and understandings between the parties with respect to its subject matter.
- 19.9 <u>Waiver</u>. No waiver of any default in the performance of any of the duties or obligations arising under this Agreement shall be valid unless in writing and signed by the waiving party. Waiver of any one default shall not constitute or be construed as creating a waiver of any other default or defaults. No course of dealing between the parties shall operate as a waiver or preclude the exercise of any rights or remedies under this Agreement. Failure on the part of either party to object to any act or failure to act of the other party, or to declare the other party in default, regardless of the extent of such default, shall not constitute a waiver by such party of its rights hereunder.

IN WITNESS WHEREOF, the Parties hereto have caused this Agreement to be duly executed.

Date:
By: Greenlane Holdings, LLC By: Jacoby & Co. Inc. Its: Managing Member
By:Aaron LoCascio
Co-President
By:
Adam Schoenfeld Co-President
Accepted and Agreed:
Jason Brown
Data:

Schedule 1

Compensation The Company shall pay, to or for the benefit of the Employee as compensation ("Salary")in the amount of \$7,500 semi-monthly (\$180,000 per year). The Company shall withhold from each compensation payment made to the Employee the required withholding and other employment-related taxes, as well as any benefits to which the Employee contributes, and any personal expenses incurred by the Employee, the payment for which was advanced by the Company.

<u>Payroll Schedule</u> All employees are paid semi-monthly, on the 5th and the 20th of each month, for the corresponding previous work period.

Benefits The Employee will receive benefits based on employment status in accordance with the requirements as stated in the company's summary of benefits.

The Employee agrees to keep the terms of this Agreement confidential, and shall not disclose any of the terms and conditions of this Agreement to third parties or otherwise make the terms and conditions of this Agreement public except pursuant to the order of a court of competent jurisdiction or in connection with legal proceedings to enforce the terms and conditions of this Agreement. The Employee acknowledges that the Employee's breach of this provision is grounds for immediate termination, and further acknowledges that such a breach would cause the Company irreparable harm for which monetary compensation alone would be an inadequate remedy. Accordingly, without limiting any other remedies available to the Company in connection with a breach or threatened breach of this provision, the Company shall be entitled to injunctive relief if the Employee breaches or threatens to breach this provision.

EXHIBIT A

EMPLOYEE PROPRIETARY RIGHTS AND CONFIDENTIALITY AGREEMENT

This Employee Proprietary Rights and Confidentiality Agreement, made the 14th day of January, 2019, is between Greenlane Holdings, LLC ("Company") and Jason Brown ("Employee").

WHEREAS, subject to Employee's agreement to, and compliance with, this Agreement Company has agreed to employee, and Employee has agreed to be employed by Company; and

WHEREAS, Company and Employee each acknowledge and agree that, as part of Employee's job duties, Employee will from time-to-time have access to Confidential Information as more specifically defined below and may from time-to-time conceive of, author, or otherwise create or assist in the conception, authorship or development of certain Inventions and Works of Authorship as more specifically defined below;

NOW THEREFORE, for good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, Company and Employee agree as follows:

1. <u>Intellectual Property</u>.

- a. Ownership and Works Made For Hire. Employee hereby agrees that Company shall be the sole and exclusive owner of any and all ideas, inventions, discoveries, computer hardware and software, algorithms, improvements and devices relating thereto, and other information and know-how heretofore or hereinafter conceived, developed or made by Employee, either alone or with others, during the course of Employee's employment by Company, or which are conceived, developed or made using Company equipment or resources, or by reason of knowledge gained from such employment (hereinafter "Inventions"), together with any graphic designs, narrative works, business plans, financial analyses, lab notes, designs, specifications, programming, source code, object code, flow charts, data, compilations of data and other information and works of authorship of any nature compiled, authored or otherwise developed by Employee and relating to such Inventions or otherwise created during the course of Employee's employment with Company or using Company equipment or resources (the "Works of Authorship") and all trademark, copyright, trade secret, patent and other intellectual and industrial property rights of whatever nature throughout the world covering or embodied by such Inventions and Works of Authorship, that the Works of Authorship constitute "works made for hire" (as that term is defined in the United States Copyright Act as amended) by Employee because they have either been prepared by Employee while acting within the scope of Employee's employment with Company or they have been specifically ordered or commissioned from Employee for use by Company and that, as such, Company is the author and owner of such Works of Authorship from the first instance of creation.
- b. <u>Assignment</u>. Employee hereby acknowledges and agrees that, to the extent any Inventions or Works of Authorship are not, for whatever reason, deemed to be "works made for hire" or otherwise solely and exclusively owned by Company, Employee hereby

grants, transfers, conveys and assigns to Company and its successors and assigns all of Employee's right, title, interest, ownership and all subsidiary rights in and to such Inventions and Works of Authorship and all trademark, trade secret, copyright, patent and other intellectual and industrial property rights of whatever nature throughout the world covering or embodied by such Inventions and Works of Authorship, including the right to apply for, register, obtain, maintain, enforce and defend the foregoing in any jurisdiction throughout the world.

- c. <u>Disclosure</u>. Employee hereby agrees to keep and maintain current and complete records of, and to disclose promptly and in writing to Company, all Inventions and Works of Authorship to which Company is or may arguably be entitled as provided above, and agrees not to (i) disclose such Inventions and Works of Authorship to others, except as required in accordance with Employee's duties as an employee of Company, or (ii) move copies of such written records to, or keep copies of such records at, locations other than the Company's premises, in either case without the express prior written consent of Company.
- d. Continuing Assistance. Employee hereby further agrees that during the term of Employee's employment with Company and at any time thereafter, Employee shall execute and deliver to Company any written documents necessary or appropriate, in Company's sole discretion, to effectuate the assignment to Company of any and all rights in such Inventions and Works of Authorship to which Company is entitled as provided above, and will execute all papers and perform any other lawful acts requested by Company for the preparation, prosecution, procurement, maintenance, enforcement and defense of any trademark, trade secret, copyright, patent and/or other intellectual or industrial property rights in and for such Inventions and Works of Authorship throughout the world, and will execute all documents and perform any other lawful act necessary to vest title in Company to such Inventions and Works of Authorship and all trademarks, trade secrets, copyrights, patents and other intellectual and industrial property rights relating thereto. To the extent that, for any reason or no reason, including without limitation Employee's physical or mental incapacity, Employee fails to comply with Employee's obligations described above in this Section 1(d), Employee hereby appoints Company, acting by and through Company's designated officers or representatives, as Employee's agent and attorney in fact, and grants to Company a power-of-attorney with full power of substitution, to execute, deliver and file any documents, and perform any acts, Employee would otherwise lawfully be obligated to execute, deliver, file or perform pursuant to this Section 1(d), with the same legal force and effect as if executed, delivered, filed or performed by Employee. Such power of attorney shall be deemed coupled with an interest for the purposes of effecting the foregoing assignments, prosecutions, procurement, maintenance, enforcement and defense.
- e. <u>Pre-existing Rights</u>. In the event that Employee uses or incorporates into any Inventions, Works of Authorship or other work product of Employee created in the course of Employee's employment with Company any (i) ideas, inventions, discoveries, computer hardware and software, algorithms, improvements and devices relating thereto conceived, developed or made by Employee prior to, or outside the scope of, Employee's employment with Company, or (ii) any graphic designs, narrative works, business plans, financial

analyses, lab notes, designs, specifications, programming, source code, object code, flow charts, data, compilations of data and other information compiled, authored, developed or otherwise created by Employee prior to, or outside the scope of, Employee's employment with Company (collectively, the "Pre-existing Inventions and Works"), then Employee shall be deemed to have granted to Company a perpetual, irrevocable, fully-paid, royalty-free, non-exclusive, transferable, assignable, sublicenseable right, under any and all trademarks, trade secrets, copyrights, patents and other intellectual and industrial property rights of whatever nature throughout the world covering or embodied by the Pre-existing Inventions and Works, to make, have made, use, copy, distribute, display, perform, modify, translate, offer to sell, sell, and otherwise transfer or dispose of the Pre-existing Inventions and Works and any products and services embodying or covered by such trademarks, trade secrets, copyrights, patents and other intellectual and industrial property rights. Employee covenants and agrees that it shall notify Company in writing of all Pre-existing Inventions and Works prior to incorporating the same into any Inventions, Works of Authorship or other work product created in the course of Employee's employment with the Company.

f. Moral Rights Waiver. Employee hereby releases, waives and to the extent allowed by law assigns in favor of Company any and all "moral" rights related to the Works of Authorship, including without limitation: the right of paternity (i.e., acknowledgment of authorship), disclosure, withdrawal and integrity (i.e., right to prevent changes to the work).

2. <u>Confidentiality and Non-Disparagement.</u>

- For so long as Employee is employed by Company and thereafter a. Non-Disclosure. for so long as the Confidential Information (as hereinafter defined) has not otherwise been publicly disclosed by Company, Employee hereby expressly agrees that (i) Employee shall not divulge, directly or indirectly, other than in the regular and proper course of business of Company, any Confidential Information, and (ii) Employee shall not use, directly or indirectly, any Confidential Information for the benefit of anyone other than Company; provided, however, that Employee shall have no obligation, express or implied, to refrain from using or disclosing to others any Confidential Information which is or hereafter shall become generally available to the public other than through Employee's breach of this Agreement. In the event Employee is or may foreseeably be compelled to disclose any Confidential Information pursuant to any law, regulation, order of a court of competent jurisdiction or other legal process, Employee shall give Company prior written notice, reasonable in light of the circumstances, of such compelled disclosure. In the event Employee makes or learns of any unauthorized disclosure of Confidential Information, Employee shall promptly notify Company of such unauthorized disclosure and in good faith cooperate with Company's efforts to ameliorate the effects of, or otherwise enforce its rights in connection with, such unauthorized disclosure.
- b. <u>Confidential Information</u>. For the purposes of this Agreement, the term "**Confidential Information**" shall mean any information not generally known in the relevant trade or industry (i.e. any trade or industry in which Company engages or is then engaged) or otherwise not generally available to the public, which was obtained from or disclosed to Employee by Company, its affiliates or their predecessors or which was learned,

discovered, developed, conceived, originated or prepared by Employee in the course of Employee's employment with Company, including, without limitation, Inventions, Works of Authorship, and any other product, service, customer, marketing, operating and financial information, computer software, source code, object codes, and flow charts proprietary to Company.

- c. <u>Possession</u>. All Confidential Information in Employee's possession at any time shall be the exclusive property of Company, shall not be copied, summarized, extracted from, or removed from the premises of Company, except in pursuit of the business of Company and at the direction of Company, and shall be delivered to Company, without Employee retaining any copies, upon the termination of Employee's employment with Company or at any earlier time as requested by Company.
- Employee warrants that the performance of its duties and d. Prior Obligations. obligations under this Agreement will not constitute a breach or default under any other agreement to which Employee is bound, that it is not bound by any pre-existing (i.e., before the date of this Agreement first set forth above) obligation of confidentiality or duty to assign rights in any Inventions or Works of Authorship (whether written, oral or otherwise) to any other person or entity that may foreseeably interfere with or conflict with Employee's job duties or compliance with this Agreement, nor is there any pending or threatened proceeding, investigation or action alleging (or to Employee's knowledge any basis for such a proceeding, investigation or action) that Employee is bound by any such obligation or duty. Employee covenants and agrees that, for so long as Employee is employed by Company, Employee shall (i) comply with any pre-existing confidentiality obligations, (ii) in the event Employee's job duties with Company may in the future potentially conflict with such obligations, inform Company of such conflict and refrain from making any disclosures or otherwise taking any action that violate Employee's preexisting confidentiality obligations, (iii) at all times in good faith work with Company to determine a course of action which will allow Employee to perform Employee' job duties without violating any pre-existing confidentiality obligations, and (iv) not hereafter enter into any agreement with any other person or entity that imposes upon Employee obligations of confidentiality that conflict with, or may foreseeably conflict with, the proper performance of Employee's job duties with Company. Employee hereby consents to Company notifying any subsequent employers of Employee that Employee is bound by obligations of confidentiality as set forth in this Agreement.
- e. <u>Defend Trade Secrets Act Notice</u>. Company hereby notifies Employee, and Employee hereby acknowledges, that nothing in this Agreement is intended to interfere with or restrain the immunity provided under the Defend Trade Secrets Act (18 U.S.C. Section 1833(b)). Specifically, an individual shall not be held criminally or civilly liable under any Federal or State trade secret law for the disclosure of a trade secret that is made in a complaint or other document filed in a lawsuit or other proceeding if such filing is made under seal. An individual who files a lawsuit for retaliation by an employer for reporting a suspected violation of law may disclose the trade secret to the attorney of the individual, and may use the trade secret information in the court proceeding, if the individual files any

document containing the trade secret under seal and does not disclose the trade secret except pursuant to court order.

f. Non-disparagement. Employee covenants and agrees that, for so long as Employee is employed by Company and for one year thereafter, Employee shall not make, publish or otherwise disseminate disparaging or derogatory statements or allegations, whether written, oral or otherwise, regarding Company or its affiliates, or any officers, directors, employees, products, services, policies or operations of Company or its affiliates. The provisions of this paragraph 2(f) in no way restrict or prohibit Employee from providing truthful testimony when such testimony is lawfully compelled pursuant to any court order or other legal process.

3. No Further Consideration.

Employee acknowledges and agrees that, except as compensated in accordance with Employee's status as an employee of Company, Employee shall not be entitled to any further or additional compensation in consideration of (i) the assignment by Employee of Employee's rights in any Inventions and/or Works of Authorship as provided for above in paragraphs 1 (a) and (b), (ii) the licenses (if any) granted to Company as provided for above in paragraph 1(e), (iii) complying with the confidentiality and non-disparagement obligations set forth above in paragraph 2, and (iv) the provision of any services as provided for above in paragraph 1(d); provided, however, that Employee shall be reimbursed for actual out-of-pocket expenses incurred in rendering to Company the services specified in paragraph 1(d).

- 4. <u>Term.</u> This Agreement shall commence and be effective as of the date of this Agreement first set forth above and shall terminate on the date Employee ceases to be an employee of Company. Notwithstanding the above, Employee's obligations arising pursuant to paragraphs 1(c) (d) and (e) and 2 (a), (c), (e) and (f) above, and paragraphs 5 and 6 (c), (d), (e) and (f) below, shall survive termination of this Agreement. In addition, those rights, duties and obligations of Company and Employee set forth in this Agreement that, by their nature, are intended to survive the termination of this Agreement shall survive the termination of this Agreement.
- 5. Originality. Employee hereby warrants and represents to Company that, to the best of Employee's knowledge after reasonable inquiry and except as expressly disclosed to Company in writing by Employee, all Inventions, Works of Authorship and Pre-existing Inventions and Works shall be original and shall not have been conceived, reduced to practice, authored or otherwise developed by misappropriating the trade secrets or other proprietary technology or intellectual property of any third parties. The foregoing representation and warranty shall be deemed first made on the date hereof and shall run continuously thereafter and apply to all Inventions, Works of Authorship and Pre-existing Inventions and Works governed by this Agreement.

6. Miscellaneous.

a. <u>Assignability</u>. Employee shall not assign this Agreement without the express prior written consent of Company, which consent may be withheld for any reason or no reason at Company's sole discretion. Any assignment in violation of this provision shall be void *ab*

initio and of no force or effect. Except as otherwise provided herein, this Agreement shall be binding upon the respective heirs, successors, and permitted assigns of the parties hereto. Company shall be free to assign its rights hereunder to any other party without the consent of Employee.

- b. <u>Severability</u>. Any provision of this Agreement which is determined to be illegal, prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such illegality, prohibition or unenforceability without invalidating the remaining provisions hereof which shall be severable and enforceable according to their terms and any such prohibition or unenforceability in any jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction.
- c. Specific Performance and Equitable Relief. Employee acknowledges and agrees that Company would suffer irreparable harm if Company's rights in the Inventions and Works of Authorship were not protected and/or the Confidential Information was disclosed to third parties without Company's consent. Accordingly, Employee hereby consents to and agrees that, in the event of a breach or threatened breach of this Agreement by Employee, Company shall have the right to exercise any and all rights by appropriate action either by law or in equity, including specific performance of Employee's obligations arising pursuant to paragraphs 1(c) and (d) above, and that Company shall be entitled to equitable relief, including injunctive relief, in connection with any breach or threatened breach, by Employee, of Employee's obligations arising pursuant to paragraphs 2(a), (c), (d), (e) and (f) above. Employee further agrees that Employee shall not request that Company post, nor shall Company be obligated to post, a bond in connection with any equitable relief authorized pursuant to this paragraph 6(c) and in fact requested by Company.
- d. <u>No Expectation of Privacy</u>. Employee acknowledges and agrees that Employee has no expectation of privacy in respect of any information (e.g., emails and voicemails), whether personal or otherwise, stored on Company's networking, telecommunications, computer and other equipment, which information may be monitored by Company at any time without notice.
- e. Governing Law. This Agreement shall be governed by, and construed and enforced in accordance with, the laws of the State of California and, to the extent applicable, the laws and regulations of the United States of America governing intellectual property matters, without giving effect to the provisions, policies or principles under California law respecting conflict or choice of law. EMPLOYEE ACKNOWLEDGES AND AGREES THAT, PRIOR TO EXECUTING THIS AGREEMENT, EMPLOYEE WAS AFFORDED AN OPPORTUNITY TO OBTAIN THE ADVICE OF INDEPENDENT LEGAL COUNSEL, AND HAS READ AND UNDERSTOOD ALL OF THIS AGREEMENT. ACCORDINGLY, THIS AGREEMENT SHALL NOT BE CONSTRUED AGAINST ANY PARTY BY REASON OF THE DRAFTING OR PREPARATION OF THIS AGREEMENT.
- f. <u>Choice of Forum</u>. Each of the parties hereby consents to the jurisdiction of any state or federal court located within the County of Los Angeles, State of California, and irrevocably

agrees that all actions or proceedings relating to this Agreement must be litigated in such courts, and each of the parties waives any objections which it may have based on improper venue or forum non conveniens to the conduct of any proceeding in such court.

- g. <u>Entire Agreement</u>. This Agreement, together with the written employment agreement (if any) to which the Company and Employee are a party, contains the entire understanding of the parties with respect to its subject matter, and supersedes all prior written or oral agreements and understandings between the parties with respect to its subject matter. This Agreement may not be changed orally, and any written change or amendment must be signed and accepted by Company.
- h. <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, and by the different parties hereto in separate counterparts, each of which shall be deemed to be an original but all of which taken together shall constitute one and the same agreement (and all signatures need not appear on any one counterpart), and this Agreement shall become effective when one or more counterparts has been signed by each of the parties hereto and delivered to each of the other parties hereto.

[Remainder of page intentionally left blank; signature page follows]

IN WITNESS WHEREOF, the parties have executed this Agreement and caused it to be binding and effective upon each of them in accordance with its terms.

By: Jacoby & Co. Inc. Its: Managing Member	Employee	
its. Managing Member		
By:		
Name: Aaron LoCascio	Jason Brown	
Title: Co-President		
By:		
Name: Adam Schoenfeld		
Title: Co-President		

DISCLOSURE SCHEDULES

In connection with that certain Contribution Agreement (the "Agreement") entered into as of January 4, 2019 by and among Pollen Gear LLC, a Delaware limited liability company (the "Company"), Pollen Gear Holdings LLC, a California limited liability company (the "Seller") and Greenlane Holdings LLC, a Delaware limited liability company (the "Purchaser"), the Company and the Seller hereby deliver these Disclosure Schedules to the Purchaser pursuant to Sections 3 and 4 of the Agreement (the "Disclosure Schedules"). The section numbers in these Disclosure Schedules correspond to the section numbers of the representations and warranties in the Agreement; provided, however that any information disclosed herein under any section number shall be deemed to be disclosed and incorporated into any other section number under the Agreement where such disclosure would be appropriate and is reasonably apparent.

Disclosure of any information or document herein is not a statement or admission that it is material or required to be disclosed herein. These Disclosure Schedules and the information and disclosures contained in these Disclosure Schedules are intended only to qualify and limit the representations and warranties of the Company and the Seller contained in the Agreement and shall not be deemed to expand in any way the scope or effect of any such representations, warranties or covenants, regardless of whether portions of such representations and warranties are duplicated in these Disclosure Schedules. No disclosure in these Disclosure Schedules relating to any fact or circumstance or to any possible breach or violation of any agreement, law or regulation shall be construed as an admission or indication with respect to any such fact or circumstance or that any such breach or violation exists or has actually occurred. Capitalized terms used but not defined herein shall have the same meanings given them in the Agreement.

Schedule 3.05

Litigation

- 1. Notice of Seizure and Information to Claimants Non-CAFRA Form to Rocketmang LLC date-stamped March 12, 2018, Case Number: 2018-2704-000389-01. Rocketmang LLC has contested the seizure and has an agreement in principle to settle the claim, pursuant to which the importation of the products that were the subject of the Notice of Seizure will be permitted which is being finalized as of the date hereof.
- 2. The Company received from eBottles.com ("eBottles"), a company that manufactures and distributes bottles and jars for various uses, a cease and desist letter, dated January 31, 2018, alleging that the Company infringed on a design patent and federal trade dress rights owned by eBottles that eBottles uses for one of eBottles's SKUs (i.e., for its "concentrate jar"). The Company corresponded with eBottles, and has taken the position that eBottles's design patent is invalid and unenforceable. The matter has been inactive.

Seller Capitalization

The Seller has a 100% owned Subsidiary, the Company, which in turn has a 100% owned Subsidiary, Rocketmang LLC.

Organization; Power

The Company is qualified to do business in California and New York.

Company Capitalization

Schedule 4.02(b)

- 1. The Company has issued Convertible Notes in the aggregate amount of \$1,500,500 which shall be converted into equity of the Seller prior to the Closing. The names of the holders of the Convertible Notes and amounts thereunder are set forth below.
 - a. Convertible Note No. 1 dated February 4, 2016, by and between Pollen Gear LLC and Deron J. Haley
 - i. Amount: \$50,000
 - ii. Maturity Date: February 4, 2018
 - iii. Interest: 5%, commencing on February 4, 2016. Interest shall be payable on the earlier of the Maturity Date or any conversion of the Note
 - b. Convertible Note No. 2, dated February 17, 2016, by and between Pollen Gear LLC and Thomas E. Katsis
 - i. Amount: \$100,000
 - ii. Maturity Date: February 17, 2018
 - iii. Interest: 5%, commencing on February 17, 2016
 - c. Convertible Note No. 3 dated February 17, 2016, by and between Pollen Gear LLC and Robert Goldstein
 - i. Amount: \$50,000
 - ii. Maturity Date: February 17, 2018
 - iii. Interest: 5%, commencing on February 17, 2016
 - d. Convertible Note No. 3A, dated February 24, 2016, by and between Pollen Gear LLC and John Seitz
 - i. Amount: \$50,000
 - ii. Maturity Date: February 24, 2018
 - iii. Interest: 5%, commencing on February 24, 2016
 - e. Convertible Note No. 4, dated February 25, 2016, by and between Pollen Gear LLC and Jonathann Kuo.
 - i. Amount \$50,000
 - ii. Maturity Date: February 25, 2018
 - iii. Interest: 5%, commencing on February 25, 2016
 - f. Convertible Note No. 5, dated March 2, 2016, by and between Pollen Gear LLC and George Mueller.
 - i. Amount \$75,000
 - ii. Maturity Date: March 2, 2018
 - iii. Interest: 5%, commencing on March 2, 2016
 - g. Convertible Note No. 6, dated June 22, 2016, by and between Pollen Gear LLC and Crimson Fund I, LLC
 - i. Amount: \$350,000
 - ii. Maturity Date: June 22, 2018
 - iii. Interest: 5%, commencing on June 22, 2016
 - h. Convertible Note, dated October 25, 2016, by and between Pollen Gear LLC and David Cooper.
 - i. Amount: \$25,000
 - ii. Maturity Date: October 25, 2018
 - iii. Interest: 5%, commencing October 25, 2016

- i. Convertible Note No. 8, dated September 22, 2016, by and between Pollen Gear LLC and John Seitz.
 - i. Amount: \$40,000
 - ii. Maturity Date: September 22, 2018
 - iii. Interest: 5%, commencing September 22, 2016
- Convertible Note dated October 25, 2016, by and between Pollen Gear LLC and Crimson Fund I, LLC.
 - i. Amount: \$37,500
 - ii. Maturity Date: October 25, 2018
 - iii. Interest: 5%, commencing October 25, 2016
- k. Convertible Note dated November 2, 2016, by and between Pollen Gear LLC and Mark Pieri.
 - i. Amount: \$15,000
 - ii. Maturity Date: November 2, 2018
 - iii. Interest: 5%, commencing November 2, 2016
- Convertible Note dated November 4, 2016, by and between Pollen Gear LLC and Calum Wallace
 - i. Amount: \$10,000
 - ii. Maturity Date: November 4, 2018
 - iii. Interest: 5%, commencing November 4, 2016
- m. Convertible Note dated November 9, 2016, by and between Pollen Gear LLC and Harrison Future Funds LLC.
 - i. Amount: \$50,000
 - ii. Maturity Date: November 9, 2018
 - iii. Interest: 5% commencing November 9, 2016
- n. Convertible Note dated November 9, 2016 by and between Pollen Gear LLC and Charles Ng.
 - i. Amount: \$50,000
 - ii. Maturity Date: November 9, 2018
 - iii. Interest: 5% commencing November 9, 2016
- o. Convertible Note dated November 11, 2016, by and between Pollen Gear LLC and Darren Plested
 - i. Amount: \$50,000
 - ii. Maturity Date: November 11, 2018
 - iii. Interest: 5% commencing November 11, 2016
- p. Convertible Note dated November 14, 2016, by and between Pollen Gear LLC and UNIS Venture Fund LLC
 - i. Amount: \$88,000
 - ii. Maturity Date: November 14, 2018
 - iii. Interest: 5%, commencing November 14, 2016
- q. Convertible Note dated November 16, 2016, by and between Pollen Gear LLC and Swamy Holdings LLC
 - i. Amount: \$25,000
 - ii. Maturity Date: November 16, 2018
 - iii. Interest: 5%, commencing November 16, 2016
- r. Convertible Note dated November 29, 2016, by and between Pollen Gear LLC and Chi Ming Tse.
 - i. Amount: \$100,000
 - ii. Maturity Date: November 29, 2018
 - iii. Interest: 5%, commencing November 29, 2018

- s. Convertible Note dated February 27, 2017, by and between Pollen Gear LLC and Thomas DesChamps.
 - i. Amount: \$100,000
 - ii. Maturity Date: February 27, 2019
 - iii. Interest: 5%, commencing February 27, 2017
- t. Convertible Note dated March 20, 2017, by and between Pollen Gear LLC and 30 N Capital, LLC.
 - i. Amount: \$25,000
 - ii. Maturity Date: March 20, 2019
 - iii. Interest: 5%, commencing March 20, 2017
- u. Convertible Note dated June 22, 2017, by and between Pollen Gear LLC and CoopMill LLC
 - i. Amount: \$50,000
 - ii. Maturity Date: June 22, 2019
 - iii. Interest: 5%, commencing June 22, 2017
- v. Convertible Note dated September 11, 2017, by and between Pollen Gear LLC and UNIS Venture Fund LLC.
 - i. Amount: \$60,000
 - ii. Maturity Date: September 11, 2019
 - iii. Interest: 5%, commencing June 22, 2017
- w. Convertible Note No. dated November 24, 2017, by and between Pollen Gear LLC and Crimson Fund I, LLC
 - i. Amount: \$25,000
 - ii. Maturity Date: November 24, 2019
 - iii. Interest: 5%, commencing November 24, 2017
- x. Convertible Note dated November 24, 2017, by and between Pollen Gear LLC and Joshua Roberts
 - i. Amount: \$25,000
 - ii. Maturity Date: November 24, 2019
 - iii. Interest: 5%, commencing November 24, 2017
- 2. All of the outstanding Class B Profits Interests of the Company, which will accelerate and vest upon the consummation of the transactions contemplated by the Agreement, will be transferred to Seller prior to the Closing.
- 3. Master Distribution Agreement, dated December 22, 22, 2016, between Pollen Gear LLC, and Warehouse Goods LLC dba Greenlane, in which in Section 1.8, the Company grants Greenlane a right to receive part of the proceeds attributable to a Liquidity Event (as defined therein).

Schedule 4.02(c)

Rocketmang LLC, a Delaware limited liability company. 100% of the membership interests of Rocketmang LLC are owned by the Company.

No Conflict or Violation

None

Consents and Approvals

- 1. Consent of Landlord for assignment of Standard Industrial/Commercial Multi-Tenant Lease, dated May 3, 2018.
- 2. Canon Equipment Lease Agreement, dated July 23, 2018 (Purchaser has waived the requirement to obtain this consent in connection with the Closing).

Financial Statements; No Undisclosed Liabilities

Schedule 4.05(a)

See attached

Schedule 4.05(b)

None

Schedule 4.06(b)

Tax Matters

Internal Revenue Service California Franchise Tax Board New York Department of Taxation and Finance

Absence of Certain Changes

Schedule 4.07(e)

- 1. Ed Kilduff assigned 865,376 Class A Units of the Company to Edward H. Kilduff, Trustee of the RS 2018 Trust dated September 4, 2018.
- 2. Ed Kilduff engaged in a transaction with prior partner to buy-out his equity. See Redemption and Equity Transfer Agreement, dated as of July 16, 2018, among Edward Kilduff, Passa Chattra, and the Company, on the one hand, and Dean Chapman, Freshlink Product Development, LLC, and Pollen Design Inc., on the other hand. This equity buy-out is disclosed on the Capitalization Table in Exhibit B.

Schedule 4.07(f)

- 1. The Company hired two new employees: Kenny Chow, with a salary of \$50,000, and Andrew Reeves, with a salary of \$85,000.
- 2. Shane Blomberg received an additional equity grant, as disclosed in the Company Capitalization Table in Exhibit B.
- 3. Consultant to the Company Andrew Triola received a \$1250/month increase in compensation.
- 4. Consultant to the Company David Chen received a \$2,000 per month increase in compensation.

Schedule 4.07(g)

- 1. From January 1, 2018 to September 30, 2018, Ed Kilduff, who is a partner and not an employee, was incorrectly listed in the Company's payroll. This error is being corrected by the Company.
- 2. Jason Brown and Shane Blomberg are listed as employees of the Company even though they own equity interests in the Company.

Schedule 4.07(j)

The Company's January 21, 2016 Operating Agreement was amended and restated with the Amended and Restated Operating Agreement of Pollen Gear, LLC dated as of July 16, 2018.

Leased Real Property

Schedule 4.09(a)

AIRCR Standard Industrial/Commercial Multi-Tenant Lease – Net, dated May 3, 218 between David Shaw ("*Lessor*") and the Company for the premises located at 601 Cypress Avenue, Unit 405, Hermosa Beach, CA 90254.

Schedule 4.09(b)

None

Assets

N/A

Accounts Receivable

None

Intellectual Property

Schedule 4.12(a)

- 1. <u>Utility and Design Patents and Patent Applications</u>
 - A. PCT/US18/61386, filed 11-15-2018
 - B. PCT/US17/55458, filed 10-06-2017
 - C. 62/715,955, filed 08-08-2018
 - D. 62/711,172, filed 07-27-2018
 - E. 62/586,773, filed 11-15-2017
 - F. 62/490,345, filed 04-26-2017
 - G. 62/439,357, filed 12-27-2016
 - H. 62/422,256, filed 11-15-2016
 - I. 62/419,209, filed 11-08-2016
 - J. 62/405,436, filed 10-07-2016
 - K. 29/670,346, filed 11-15-2018
 - L. 29/670,338, filed 11-15-2018
 - M. 29/658,099, filed 07-27-2018
 - N. 29/658,083, filed 07-27-2018
 - O. 29/658,079, filed 07-27-2018
 - P. 29/626,205, filed 11-15-2017
 - Q. 29/626,177, filed 11-15-2017
 - R. 29/601,826, filed 04-26-2017
 - S. 29/601,823, filed 04-26-2017
 - T. 29/589,025, filed 12-27-2016
 - U. 29/589,022, filed 12-27-2016
 - V. 29/589,016, filed 12-27-2016
 - W. 29/584,476, filed 11-15-2016 and now granted as D807,197
 - X. 29/583,722, filed 11-08-2016
 - Y. 29/580,322, filed 10-07-2016
 - Z. 29/517,993, filed 2-19-2015 and now granted as D787,895.



S4S Pollen Gear Patent Family Diagra

- 2. Foreign Utility and Design Patents and Patent Applications
 - A. AU 201712014
 - B. AU 201712728
 - C. AU 201714307
 - D. AU 201714308
 - E. AU 201716469
 - F. CA 162028
 - G. CA 166225
 - H. CA 174082

- I. CA 174702
- J. CA 174792
- K. CA 177837
- L. CA 179431
- M. CN 201530102693
- N. CN 201530102777
- O. CN 201710930652
- P. CN 201721295200
- Q. CN 201730112145, registered as CN 304481096 S
- R. CN 201730112267, registered as CN 304339937 S
- S. CN 201730112289, registered as CN 304339492 S
- T. CN 201730164473, registered as CN 304424591 S
- U. CN 201730164689, registered as CN 304533541 S
- V. CN 201730516422
- W. CN 201830498071
- X. EUIPO 002676890-0001 to 0005
- Y. EUIPO 003844448-0001 to 0024
- Z. EUIPO 003933894-0001 to 0032
- AA. EUIPO 004419091-0001 to 0019

3. <u>Trademark Registrations and Applications</u>

	Trademark	Country	Application No.	Application Date	Registration No.	Registration Date	Int. Classes	Trademark Status
A.	POLLEN GEAR	Canada	1835611	5/3/17			6, 10, 16, 20, 21, 34, 35, 39	Pending
В.	POLLEN GEAR	Canada	1893606	4/13/18			6, 16, 18, 20, 21	Pending
C.	POLLEN GEAR	China	23997047	11/8/16			6	Abandoned
D.	POLLEN GEAR	China	23997272	11/8/16	23997272	6/14/18	35	Registered
E.	POLLEN GEAR	China	23997273	11/8/16	23997273	6/14/18	34	Registered
F.	POLLEN GEAR	China	23997274	11/8/16	23997274	6/14/18	21	Registered
G.	POLLEN GEAR	China	23997275	11/8/16	23997275	6/14/18	20	Registered

	Trademark	Country	Application No.	Application Date	Registration No.	Registration Date	Int. Classes	Trademark Status
H.	POLLEN GEAR	China	23997276	11/8/16	23997276	6/14/18	16	Registered
I.	POLLEN GEAR	China	23997277	11/8/16	23997277	6/14/18	10	Registered
J.	POLLEN GEAR	China	23997271	11/8/16	23997271	6/14/18	39	Registered
K.	POLLEN GEAR	EUTM	126680852	5/4/17	16680852	8/31/17	6, 10, 16, 20, 21, 34, 35, 39	Registered
L.	POLLEN GEAR	USA	87230681	11/8/16			6, 10, 16, 20, 21, 34, 35, 39	Pending
M.	POP BOX	USA	87230695	11/8/16	5493002	6/12/18	34	Registered
N.	SNAPTECH	USA	87249464	11/28/16	5493032	6/12/18	16, 34, 35	Registered

- 1. In connection with the proposed transaction, IP counsel reviewed the above list of trademarks that was provided by Seller.
- The U.S. mark: ownership confirmed. No liens/encumbrances. The mark is scheduled to be published for opposition on December 11, 2018. If no oppositions or extensions of time are filed before January 11, 2019, the mark will register: http://tsdr.uspto.gov/#caseNumber=87230681&caseSearchType=US_APPLICATION&caseType=DEFAULT&searchType=statusSearch.
- 3. Canadian and EU marks ownership confirmed. No liens/ encumbrances or oppositions have been filed against the marks.
- 4. The Chinese marks: According to Chinese IP counsel, "It does not appear from the records that there have been no liens or encumbrances or licenses applied for recordal or recorded against the application and registrations." Additionally, Chinese IP counsel confirmed Pollen Gear's ownership of the Chinese marks.

Schedule 4.12(b)

See Schedule 3.05(b).

Schedule 4.12(c)

1. Cease and desist letter dated November 1, 2018 sent on behalf of the Company to Packed PKG.

18

PAT

- 2. Cease and desist letter dated November 1, 2018 sent on behalf of the Company to PNP Resources, LLC dba Collective Supply.
- 3. Cease and desist letter dated November 1, 2018 sent on behalf of the Company to Compliant Packaging, LLC.
- 4. Cease and desist letter dated November 2, 2018 sent on behalf of the Company to MSN Packaging, Inc.
- 5. Cease and desist letter dated November 9, 2018 sent on behalf of the Company to Meyking Enterprises, Inc.

Schedule 4.12(d)

1. The Company has not had its consultants sign standalone non-disclosure agreements, but the Company's signed agreements with its consultants contain work-made-for-hire provisions and confidential information/non-disclosure provisions. (See Section 7 of the consulting agreements.)

Schedule 4.12(e)

None.

Schedule 4.12(f)

1. All Company employees, consultants, lawyers, accountants, Conrad Cordeiro – the Company's IT service provider, and the Purchaser.

Schedule 4.12(g)

1. Pursuant to the Redemption and Equity Transfer Agreement, dated as of July 16, 2018, among Edward Kilduff, Passa Chattra, and the Company, on the one hand, and Dean Chapman, Freshlink Product Development, LLC, and Pollen Design Inc., on the other hand, the Company has the exclusive right to use the trademark "POLLEN" throughout the world, but not "Pollen Design" as identified in. U.S. Reg. Nos. 5390178 or 2786203. (See Section 7 of the contract.)

Schedule 4.12(h)

None.

Schedule 4.12(i)

None.

Employee Benefit Plans

Schedule 4.13(a)
None.
Schedule 4.13(g)
None.
Schedule 4.13(h)
See Schedule 4.19(a) with respect to the Brown Agreement and the Blomberg Agreement
Schedule 4.13(m)

See Schedule 4.19(a) with respect to the Brown Agreement and the Blomberg Agreement.

Personnel; Labor Relations

Schedule 4.14(a)

Employee	Position	Hire Date	Status	Туре	Base Salary	Visa/Green Card Status
Ed Kilduff	CEO	6/1/2018	Exempt	Salaried	\$250,000	U.S. Citizen
Shane Blomberg	Design Director	6/1/2018	Exempt	Salaried	\$100,000	U.S. Citizen
Jason Brown	COO	6/1/2018	Exempt	Salaried	\$180,000	U.S. Citizen
Kenny Chow	Designer	9/1/2018	Exempt	Salaried	\$50,000	F-1 Visa
Andrew Reeves	Senior Designer	10/18/2018	Exempt	Salaried	\$85,000	U.S. Citizen

The Company has no agreed upon bonus obligations. All employees are based in the Company's offices in Hermosa Beach.

Schedule 4.14(c)

None.

License and Permits

1. The Company is qualified to do business in California and New York.

22

Insurance

1. Package (GL - Property and Umbrella) policy

Spectrum Policy from The Hartford

Insurer: Sentinel Insurance Company, Limited

Policy No.: 12 SBA AA9097 SB

Insured: Company and Rocketmag LLC

Policy Period: 7/17/18 to 7/17/19

2. Cargo policy - Ocean Cargo Insurance

Chubb Group of Insurance Companies

Policy No.: 0790-48-98TPA

Insured: Company

Policy Period: 7/17/18 to 7/17/19

Payment Card Standards

None.

Contracts and Commitments

Schedule 4.19(a)

- 1. Unit Restriction Agreement between the Company and Shane Blomberg dated as of June 29, 2016.
- 2. Employment Offer Letter between the Company and Jason Brown dated as of May 28, 2018.
- 3. Employment Offer Letter between the Company and Shane Blomberg dated as of June 1, 2018.
- 4. Employment Offer Letter between the Company and Kenny Chow dated as of September 1, 2018.
- 5. Employment Offer Letter between the Company and Andrew Reeves dated as of October 18, 2018.
- 6. Profits Interest Grant Agreement between the Company and Crimson Fund I, LLC dated as of July 20, 2016, as amended.¹
- 7. Profits Interest Grant Agreement between the Company and Jason Brown dated as of May 28, 2018 (the "**Brown Agreement**"). The Brown Agreement provides for an acceleration in full of all unvested units in a change of control transaction.
- 8. Profits Interest Grant Agreement between the Company and Shane Blomberg dated as of August 15, 2018 (the "**Blomberg Agreement**"). The Blomberg Agreement provides for an acceleration in full of all unvested units in a change of control transaction.
- 9. Profits Interest Grant Agreement between the Company and VS Holdings LLC dated as of July 12, 2018.

Schedule 4.19(b)

See Schedule 4.02(b) item 2.

Schedule 4.19(c)

All Company employees, and Andrew Triola, consultant for the Company, have a Company American Express card.

¹ Rob Scott and Mark Pieri do not have profit interests. They have Class B units granted upon formation of the Company.

Schedule 4.19(d)

Master Product Development Agreement, dated January 27, 2016, by and between Jane West LLC and Pollen Design Inc., as assigned to the Company

Schedule 4.19(e)

None.

Schedule 4.19(f)

None.

Schedule 4.19(g)

None.

Schedule 4.19(h)

- 1. See Schedule 4.02(b) item 2.
- 2. See <u>Schedule 4.12(g)</u>.

Schedule 4.19(g)

None.

Schedule 4.19(j)

See Schedule 4.02(b) item 2.

Schedule 4.19(k)

- 1. The Company regularly places procurement purchase orders for goods/inventory which exceed \$25,000. The Company has no fixed suppliers.
 - 2. See <u>Schedule 4.09(a)</u> (Company office lease).

<u>Schedule 4.19(1)</u>

See Schedule 4.02(b) item 2.

Schedule 4.19(m)

1. Confidentiality and Non-Disclosure Agreement, dated March 16, 2016, by among the Company, Pollen Design Inc., Freshlink Product Development LLC, and Asuwant Plastic Packaging Co. Ltd.

- 2. Confidentiality and Non-Disclosure Agreement, dated May 31, 2017, by and among the Company, Pollen Design Inc., Freshlink Product development, LLC, and Shandong Pharmaceutical Glass Co. Ltd.
- 3. Confidentiality and Non-Disclosure Agreement, dated October 1, 2017, by and among the Company, Pollen Design Inc., Freshlink Product Development, LLC, Rocketmang LLC, and Nantong Shunyu Packing Material Co. Ltd.
- 4. Confidentiality and Non-Disclosure Agreement, dated June 9, 2017, by and among the Company, Pollen Design Inc., Freshlink Product Development, LLC, Rocketmang LLC, and Sanlong Glass Product Co. Ltd.
- 5. Confidentiality and Non-Disclosure Agreement, dated May 21, 2016, by and among the Company, Pollen Design Inc., Freshlink Product Development LLC, and Shenzhen Sunny Glassware Co. Ltd.
- 6. Confidentiality and Non-Disclosure Agreement, dated August 28, 2018, by and among the Company, Pollen Design Inc., Freshlink Product Development LLC, and Shenzhen Sunny Glassware Co. Ltd.

See Schedule 4.02(b) item 2.

Pursuant to Section 12 of the Company's office lease (see <u>Schedule 4.09(a)</u>), a "change of control" of the Company (the definition of which this Agreement appears to meet) requires Lessor's consent or Lessor has the option to terminate the lease.

Customers and Suppliers

Schedule 4.20(a)

Warehouse Goods, LLC, dba Greenlane LLC.

Schedule 4.20(b)

- 1. Asuwant Plastic Packaging Co., Ltd
- 2. Haimen Design Glass Products Co., Ltd
- 3. Dongguam Xiaomi Packaging Co., Ltd
- 4. Front Point Services Limited OR Genesis Production Limited
- 5. Haimen San Long Glass Product Co., Ltd
- 6. Nantong Shunyu Packing Material Co Ltd
- 7. Nice-Can Manufacturing Co., Ltd
- 8. Shandong Pharmaceutical Glass Co., Ltd
- 9. Shenzhen Sunny Glassware Co., Ltd
- 10. Yuhuan BoYu Plastic Packaging Factory

Schedule 4.20(c)

None.

28

Compliance with Law

The Company operates in the cannabis accessories industry. Given the rapid change with respect to laws on the federal, state and local levels, there can be no assurance that the Company's operations in this industry comply with all laws with respect to cannabis accessories in particular.

Litigation

Schedule 4.22(a)
See Schedule 3.05
Schedule 4.22(b)

None.

Schedule 4.22(c)

None.

The Company knows existing facts and plans to potentially initiate legal actions against a number of companies which it believes are infringing on its design patents. See Schedule 4.12(c).

Affiliate Transactions

See Schedule 4.14.

Inventory

- 1. The Company is considering replacing 150,000 concentrate jars due to the white color paint being chipped off.
- 2. Company is proactively making Medmen replacement tubes out of glass, because the ink on the plastic tubes, as currently constructed, is rubbing off from the surface of the tubes.
- 3. One of the Company's customers, Cannedescent, has complained about Company-produced boxes being too stiff to open. The Company is considering whether to replace the boxes.

Product Matters

None.

Bank Accounts

- 1. The Company: Citibank checking account number 4998124629.
- 2. Rocketmang LLC: Citibank checking account number 6780852614

Jason Brown, Ed Kilduff, and Andrew Triola have access to the above-referenced checking accounts and are authorized to make withdrawals therefrom or dispositions thereof.

Privacy

None.

RECORDED: 02/16/2022

35